

## NOTICE OF 2024 EXTRAORDINARY GENERAL MEETING AND RELATED DOCUMENTS

Cyprium Metals Limited (ASX: CYM) (**Cyprium** or **the Company**) attaches the following documents in relation to its 2024 Extraordinary General Meeting:

- 1. Letter to Shareholders in relation to the Notice of 2024 Extraordinary General Meeting;
- 2. Notice of 2024 Extraordinary General Meeting; and
- 3. Proxy Form.

The Company advises that the Letter to Shareholders, Notice and Proxy Form has been dispatched to Shareholders.

#### This announcement has been authorised for release by the Board.

Further information:

Matt Fifield

Executive Chair matt.fifield@cypriummetals.com

T +61 8 6374 1550 E communications@cypriummetals.com

#### About Cyprium Metals Limited

Cyprium Metals Limited (ASX: CYM, OTCQB:CYPMF) is an Australian copper company.

Its flagship property is the Nifty Copper Mine in Western Australia, which previously produced significant copper from both oxide and sulphide resources. Cyprium is focused on redeveloping Nifty, which has the advantage of significant invested capital, data from a long operating history, large-scale resources, current operational approvals, and recent investment in the property.

The Company's other assets include significant copper-focused properties in the Paterson and Murchison Provinces, including multiple defined copper resources.

Visit <u>www.cypriummetals.com</u> for further information.



## CYPRIUM METALS LIMITED – EXTRAORDINARY GENERAL MEETING LETTER TO SHAREHOLDERS AND PROXY FORM

#### Dear Shareholder

Cyprium Metals Limited (ASX: CYM) (**Cyprium** or **the Company**) advises that an Extraordinary General Meeting of Shareholders will be held on Friday, 31 January 2025 at 10am (AWST) at Vibe Hotel, 9 Alvan St, Subiaco WA 6008 (**Meeting**).

In accordance with Part 1.2AA of the *Corporations Act 2001*, the Company will only be dispatching physical copies of the Notice of Meeting (**Notice**) to Shareholders who have elected to receive the Notice in physical form. The Notice is being made available to Shareholders electronically and can be viewed and downloaded online at the following link: <u>https://investorhub.cypriummetals.com/</u> Alternatively, the Notice will also be available on the Company's ASX market announcements page (ASX:CYM).

This Notice is given based on circumstances as at the date of this letter. Should circumstances change, the Company will make an announcement on the ASX market announcements platform and on the Company's website at <u>https://investorhub.cypriummetals.com/</u>. Shareholders are urged to monitor the ASX announcements platform and the Company's website.

#### Your vote is important

The business of the Meeting affects your shareholding and your vote is important. To vote in person, please pre-register in advance for the physical meeting by contacting the Company Secretary at <u>david.hwang@cypriummetals.com</u>.

All resolutions will be decided on a poll. The poll will be conducted based on votes submitted by proxy and at the Meeting. Shareholders are strongly encouraged to complete and submit their vote by proxy by using one of the following methods:

| Online   | Lodge the Proxy Form online at <u>https://investor.automic.com.au/#/loginsah</u> by following the instructions: Login to the Automic website using the holding details as shown on the Proxy Form. Click on 'View Meetings' – 'Vote'. To use the online lodgement facility, Shareholders will need their holder number (Securityholder Reference Number (SRN) or Holder Identification Number (HIN)) as shown on the front of the Proxy Form. |  |
|----------|---|--|
| By post  | Completing the enclosed Proxy Form and posting it to:<br>Automic, GPO Box 5193, Sydney NSW 2001   |  |
| By hand  | Completing the enclosed Proxy Form and delivering it by hand to:<br>Automic, Level 5, 126 Phillip Street, Sydney NSW 2000   |  |
| By email | Completing the enclosed Proxy Form and emailing it to: <u>meetings@automicgroup.com.au</u>  |  |

ABN 48 002 678 640 T +61 8 6374 1550 ASX: CYM OTCQB: CYPMF Level 1, 437 Roberts Road Subiaco, WA, 6008 www.cypriummetals.com

Cyprium Metals (ASX:CYM)



Your Proxy instruction must be received not later than 48 hours before the commencement of the Meeting. **Proxy Forms received later than this time will be invalid.** 

The Chair intends to vote all open proxies in favour of all resolutions, where permitted. On behalf of the Board, thank you for your continued support as a shareholder. We look forward to welcoming you to our EGM.

Yours faithfully,

Matthew (Matt) Fifield Executive Chairman

#### **About Cyprium Metals Limited**

Cyprium Metals Limited (ASX: CYM, OTCQB:CYPMF) is an Australian copper company.

Its flagship property is the Nifty Copper Mine in Western Australia, which previously produced significant copper from both oxide and sulphide resources. Cyprium is focused on redeveloping Nifty, which has the advantage of significant invested capital, data from a long operating history, large-scale resources, current operational approvals, and recent investment in the property.

The Company's other assets include significant copper-focused properties in the Paterson and Murchison Provinces, including multiple defined copper resources.

Visit <u>www.cypriummetals.com</u> for further information.

ABN 48 002 678 640 T +61 8 6374 1550 ASX: CYM OTCQB: CYPMF Level 1, 437 Roberts Road Subiaco, WA, 6008 www.cypriummetals.com

## CYPRIUM METALS LIMITED ACN 002 678 640 NOTICE OF EXTRAORDINARY GENERAL MEETING

Notice is given that the Meeting will be held at:

TIME: 10 am (WST)

DATE: Friday, 31 January 2025

PLACE: Vibe Hotel 9 Alvan St, Subiaco WA 6008

The business of the Meeting affects your shareholding and your vote is important.

This Notice of Meeting should be read in its entirety. If Shareholders are in doubt as to how they should vote, they should seek advice from their professional advisers prior to voting.

The Directors have determined pursuant to Regulation 7.11.37 of the Corporations Regulations 2001 (Cth) that the persons eligible to vote at the Meeting are those who are registered Shareholders at 4pm WST on Wednesday, 29 January 2025.

#### **BUSINESS OF THE MEETING**

#### AGENDA

#### 1. **RESOLUTION 1 – RATIFICATION OF PRIOR ISSUE OF TRANCHE 1 PLACEMENT SHARES**

To consider and, if thought fit, to pass, with or without amendment, the following resolution as an **ordinary resolution**:

"That, for the purposes of Listing Rule 7.4 and for all other purposes, Shareholders ratify the issue of 185,714,285 Placement Shares (issued on 20 December 2024) on the terms and conditions set out in the Explanatory Statement."

A voting exclusion statement applies to this Resolution, Please see below.

#### 2. **RESOLUTION 2 – RATIFICATION OF PRIOR ISSUE OF TRANCHE 1 PLACEMENT OPTIONS**

To consider and, if thought fit, to pass, with or without amendment, the following resolution as an **ordinary resolution**:

"That, for the purposes of Listing Rule 7.4 and for all other purposes, Shareholders ratify the issue of 92,857,143 Placement Options (issued on 20 December 2024) on the terms and conditions set out in the Explanatory Statement."

A voting exclusion statement applies to this Resolution, Please see below.

## 3. RESOLUTION 3 – APPROVAL FOR THE ISSUE OF TRANCHE 2 CORNERSTONE SECURITIES UNDER LISTING RULE 7.1

To consider and, if thought fit, to pass, with or without amendment, the following resolution as an **ordinary resolution**:

"That, for the purposes of Listing Rule 7.1 and for all other purposes, approval is given for the Company to issue up to 190,346,000 Placement Shares, 95,173,000 Placement Options and 20,000,000 Cornerstone Options to Flat Footed (or its nominees) on the terms and conditions set out in the Explanatory Statement."

A voting exclusion statement applies to this Resolution, Please see below.

#### 4. RESOLUTION 4 – APPROVAL FOR THE ISSUE OF TRANCHE 2 PLACEMENT SECURITIES TO MR J MATTHEW FIFIELD (EXECUTIVE CHAIR) UNDER LISTING RULE 10.11

To consider and, if thought fit, to pass, with or without amendment, the following resolution as an **ordinary resolution**:

"That, for the purposes of Listing Rule 10.11 and for all other purposes, approval is given for the Company to issue up to 35,714,285 Placement Shares and 17,857,142 Placement Options to Mr J Matthew Fifield (or his nominees) on the terms and conditions set out in the Explanatory Statement"

A voting exclusion statement applies to this Resolution. Please see below.

#### 5. RESOLUTION 5 – APPROVAL FOR THE ISSUE OF TRANCHE 2 PLACEMENT SECURITIES TO MR SCOTT PERRY (NON-EXECUTIVE DIRECTOR) UNDER LISTING RULE 10.11

To consider and, if thought fit, to pass, with or without amendment, the following resolution as an **ordinary resolution**:

"That, for the purposes of Listing Rule 10.11 and for all other purposes, approval is given for the Company to issue up to 53,571,428 Placement Shares and 26,785,714 Placement Options to Mr Scott Perry (or his nominees) on the terms and conditions set out in the Explanatory Statement"

A voting exclusion statement applies to this Resolution. Please see below.

#### 6. RESOLUTION 6 – APPROVAL FOR THE ISSUE OF TRANCHE 2 PLACEMENT SECURITIES TO MR ROSS BHAPPU (NON-EXECUTIVE DIRECTOR) UNDER LISTING RULE 10.11

To consider and, if thought fit, to pass, with or without amendment, the following resolution as an **ordinary resolution**:

"That, for the purposes of Listing Rule 10.11 and for all other purposes, approval is given for the Company to issue up to 17,857,142 Placement Shares and 8,928,571 Placement Options to Mr Ross Bhappu (or his nominee) on the terms and conditions set out in the Explanatory Statement"

A voting exclusion statement applies to this Resolution. Please see below.

#### 7. RESOLUTION 7 – APPROVAL FOR THE ISSUE OF PERFORMANCE RIGHTS TO MR J MATTHEW FIFIELD (EXECUTIVE CHAIR) UNDER LISTING RULE 10.14

To consider and, if thought fit, to pass, with or without amendment, the following resolution as an **ordinary resolution**:

"That, for the purposes of Listing Rule 10.14 and for all other purposes, Shareholders approve the issue of 30,000,000 Performance Rights to Mr J Matthew Fifield (or his nominee) under the Employee Securities Incentive Plan on the terms and conditions set out in the Explanatory Statement."

A voting prohibition statement and voting exclusion statement applies to this Resolution. Please see below.

#### 8. RESOLUTION 8 – APPROVAL OF POTENTIAL TERMINATION BENEFITS TO MR J MATTHEW FIFIELD (EXECUTIVE CHAIR) UNDER EMPLOYEE SECURITIES INCENTIVE PLAN

To consider and, if thought fit, to pass the following resolution as an **ordinary** resolution:

"That, subject to the passing of Resolution 7, pursuant to and in accordance with Part 2D.2 of the Corporations Act (including sections 200B and 200E of the Corporations Act) and for all other purposes, Shareholders approve the giving of Potential Termination Benefits detailed in the Explanatory Statement to Mr J Matthew Fifield (or his nominee) in connection with him ceasing to hold that managerial or executive office."

A voting prohibition statement and voting exclusion statement applies to this Resolution. Please see below.

#### 9. RESOLUTION 9 – APPROVAL OF EXECUTIVE CHAIR SALARY SACRIFICE OFFER (FY25) TO MR J MATTHEW FIFIELD UNDER EMPLOYEE SECURITIES INCENTIVE PLAN

To consider and, if thought fit, to pass the following resolution as an **ordinary** resolution:

"That, for the purposes of Listing Rule 10.14 and for all other purposes, Shareholders approve the issue of up 50% of Mr J Matthew Fifield's salary (including Director's fees) (for the period from 1 January 2025 to 30 June 2025) as Purchased Rights under the Employee Securities Incentive Plan on the terms and conditions set out in the Explanatory Statement."

A voting prohibition statement and voting exclusion statement applies to this Resolution. Please see below.

#### 10. RESOLUTION 10 – APPROVAL OF NED SALARY SACRIFICE OFFER (FY25) TO MR SCOTT PERRY UNDER EMPLOYEE SECURITIES INCENTIVE PLAN

To consider and, if thought fit, to pass the following resolution as an **ordinary resolution**:

"That, for the purposes of Listing Rule 10.14 and for all other purposes, Shareholders approve the issue of up 100% of Mr Scott Perry's NED fees (for the period from 1 January 2025 to 30 June 2025) as Purchased Rights under the Employee Securities Incentive Plan on the terms and conditions set out in the Explanatory Statement."

A voting prohibition statement and voting exclusion statement applies to this Resolution. Please see below.

#### 11. RESOLUTION 11 – APPROVAL OF NED SALARY SACRIFICE OFFER (FY25) TO MR ROSS BHAPPU UNDER EMPLOYEE SECURITIES INCENTIVE PLAN

To consider and, if thought fit, to pass the following resolution as an **ordinary** resolution:

"That, for the purposes of Listing Rule 10.14 and for all other purposes, Shareholders approve the issue of up 100% of Mr Ross Bhappu's NED fees (for the period from 1 January 2025 to 30 June 2025) as Purchased Rights under the Employee Securities Incentive Plan on the terms and conditions set out in the Explanatory Statement."

A voting prohibition statement and voting exclusion statement applies to this Resolution. Please see below.

#### 12. RESOLUTION 12 – APPROVAL OF NED SALARY SACRIFICE OFFER (FY25) TO MR GARY COMB UNDER EMPLOYEE SECURITIES INCENTIVE PLAN

To consider and, if thought fit, to pass the following resolution as an **ordinary resolution**:

"That, for the purposes of Listing Rule 10.14 and for all other purposes, Shareholders approve the issue of up 100% of Mr Gary Comb's NED fees (for the period from 1 January 2025 to 30 June 2025) as Purchased Rights under the Employee Securities Incentive Plan on the terms and conditions set out in the Explanatory Statement." A voting prohibition statement and voting exclusion statement applies to this Resolution. Please see below.

#### 13. **RESOLUTION 13 – AMENDMENT TO CONSTITUTION**

To consider and, if thought fit, to pass the following resolution as a **special resolution**:

"That, for the purposes of section 136(2) of the Corporations Act and for all other purposes, the constitution of the Company be amended with a constitution in the form of the document tabled at this Meeting and signed by the Chair for the purposes of identification, effective immediately."

#### Dated: 31 December 2024

By order of the Board

David Hwang Company Secretary

## **Voting Exclusion Statements**

In accordance with Listing Rule 14.11, the Company will disregard any votes cast in favour of the Resolution set out below by or on behalf of the following persons:

| Resolution 1 –<br>Ratification of prior<br>issue of Tranche 1<br>Placement Shares   | A person who participated in the issue or is a<br>counterparty to the agreement being approved (namely<br>the Placement Participants) or an associate of that person or<br>those persons.   |  |
|---|---|--|
| Resolution 2 –<br>Ratification of prior<br>issues of Tranche 1<br>Placement Options   | A person who participated in the issue or is a counterparty to the agreement being approved (namely the Placement Participants) or an associate of that person or those persons.  |  |
| Resolution 3 –<br>Approval for the issue<br>of Tranche 2<br>Cornerstone Securities<br>under Listing Rule 7.1  | <ul> <li>A person who is expected to participate in, or who will obtain a material benefit as a result of, the proposed issue (except a benefit solely by reason of being a holder of ordinary securities in the Company) (namely, Flat Footed) or an associate of that person (or those persons).</li> <li>However, this does not apply to a vote cast in favour of the Resolution by: <ul> <li>(a) a person as a proxy or attorney for a person who is entitled to vote on the Resolution, in accordance with the directions given to the proxy or attorney to vote on the Resolution in that way; or</li> <li>(b) the Chair as proxy or attorney for a person who is entitled to vote on the Resolution, in accordance with a direction given to the Chair to vote on the Resolution as the Chair decides; or</li> <li>(c) a holder acting solely in a nominee, trustee, custodial or other fiduciary capacity on behalf of a beneficiary provided the following conditions are met:</li> <li>(i) the beneficiary provides written confirmation to the holder that the beneficiary is not excluded from voting, and is not an associate of a person excluded from voting, on the Resolution; and</li> <li>(ii) the holder votes on the Resolution in accordance with directions given by the beneficiary to the holder to vote in that way.</li> </ul> </li> </ul> |  |
| Resolution 4 –<br>Approval for the issue<br>of Tranche 2<br>Placement Securities<br>to Mr J Matthew Fifield<br>(Executive Chair)<br>under Listing Rule<br>10.11 | J Matthew Fifield (or his nominees) and any other person who<br>will obtain a material benefit as a result of the issue of the<br>securities (except a benefit solely by reason of being a holder<br>of ordinary securities in the Company) or an associate of that<br>person or those persons.   |  |

| Resolution 5 –<br>Approval for the issue<br>of Tranche 2<br>Placement Securities<br>to Mr Scott Perry (Non-<br>Executive Director)<br>under Listing Rule<br>10.11 | Scott Perry (or his nominees) and any other person who will<br>obtain a material benefit as a result of the issue of the<br>securities (except a benefit solely by reason of being a holder<br>of ordinary securities in the Company) or an associate of that<br>person or those persons. |
|---|---|
| Resolution 6 –<br>Approval for the issue<br>of Tranche 2<br>Placement Securities<br>to Mr Ross Bhappu<br>(Non-Executive<br>Director) under Listing<br>Rule 10.11  | Ross Bhappu (or his nominees) and any other person who will<br>obtain a material benefit as a result of the issue of the<br>securities (except a benefit solely by reason of being a holder<br>of ordinary securities in the Company) or an associate of that<br>person or those persons. |
| Resolutions 7 –<br>Approval for the issue<br>of Performance Rights<br>to Mr J Matthew Fifield<br>(Executive Chair)<br>under Listing Rule<br>10.14                 | Any person referred to in Listing Rule 10.14.1, 10.14.2 or 10.14.3<br>who is eligible to participate in the employee incentive<br>scheme or an associate of that person or those persons.   |
| Resolutions 9, 10, 11<br>and 12 – Approval of<br>Salary Sacrifice Offers<br>(FY25) to Directors<br>under Employee<br>Incentive Securities<br>Plan                 | Any person referred to in Listing Rule 10.14.1, 10.14.2 or 10.14.3<br>who is eligible to participate in the employee incentive<br>scheme or an associate of that person or those persons.   |

However, this does not apply to a vote cast in favour of the Resolution by:

- (a) a person as a proxy or attorney for a person who is entitled to vote on the Resolution, in accordance with the directions given to the proxy or attorney to vote on the Resolution in that way; or
- (b) the Chair as proxy or attorney for a person who is entitled to vote on the Resolution, in accordance with a direction given to the Chair to vote on the Resolution as the Chair decides; or
- (c) a holder acting solely in a nominee, trustee, custodial or other fiduciary capacity on behalf of a beneficiary provided the following conditions are met:
  - (i) the beneficiary provides written confirmation to the holder that the beneficiary is not excluded from voting, and is not an associate of a person excluded from voting, on the resolution; and
  - (ii) the holder votes on the resolution in accordance with directions given by the beneficiary to the holder to vote in that way.

## Voting Prohibition Statements

| Resolutions 7, 9, 10, 11<br>and 12   | In accordance with section 250BD and section 200E(2A) of<br>the Corporations Act, a person appointed as a proxy must<br>not vote, on the basis of that appointment, on this Resolution<br>if:   |  |  |
|--|---|--|--|
| Resolution 7 –<br>Approval for the issue<br>of Performance Rights<br>to Mr J Matthew Fifield<br>(Executive Chair)<br>under Listing Rule                    | <ul> <li>(a) the proxy is either:</li> <li>(i) a member of the Key Management<br/>Personnel; or</li> <li>(ii) a Closely Related Party of such a member;<br/>and</li> </ul>  |  |  |
| 10.14  | (b) the appointment does not specify the way the proxy is to vote on this Resolution.   |  |  |
| Resolutions 9, 10, 11<br>and 12 – Approval of<br>Salary Sacrifice Offers<br>(FY25) to Directors<br>under Employee<br>Incentive Securities<br>Plan          | <ul> <li>However, the above prohibition does not apply if:</li> <li>(a) the proxy is the Chair; and</li> <li>(b) the appointment expressly authorises the Chair to exercise the proxy even though this Resolution is connected directly or indirectly with remuneration of a member of the Key Management Personnel.</li> </ul>   |  |  |
| Resolution 8 –<br>Approval of potential<br>termination benefits to<br>Mr J Matthew Fifield on<br>retirement under<br>Employee Securities<br>Incentive Plan | <ul> <li>A vote on this Resolution must not be cast (in any capacity) by or on behalf of J Matthew Fifield (or his nominee) or any of his, or their, associates. However, subject to the voting exclusion below and the further voting prohibition below, this does not prevent the casting of a vote if:</li> <li>(a) it is cast by a person as a proxy appointed by writing that specifies how the proxy is to vote on this Resolution; and</li> <li>(b) it is not cast on behalf of J Matthew Fifield or his nominee(s) or any of his associates.</li> </ul> |  |  |
|  | <ul> <li>Additionally, in accordance with section 250BD of the Corporations Act, as this resolution is in connection with the remuneration of a member of the Company's Key Management Personnel, a person appointed as a proxy must not vote, on the basis of that appointment, on this Resolution if:</li> <li>(a) the proxy is either:</li> <li>(b) the appointment does not specify the way the proxy is to vote on this Resolution.</li> </ul>   |  |  |
|  | <ul> <li>However, the above prohibition does not apply if:</li> <li>(a) the proxy is the Chair; and</li> <li>(b) the appointment expressly authorises the Chair to exercise the proxy even if the Resolution is connected directly or indirectly with remuneration of a member of the Key Management Personnel.</li> </ul>  |  |  |

#### Voting by proxy

To vote by proxy, please complete the enclosed Proxy Form and return by the time and in accordance with the instructions set out on the Proxy Form.

In accordance with section 249L of the Corporations Act, Shareholders are advised that:

- each Shareholder has a right to appoint a proxy;
- the proxy need not be a Shareholder of the Company; and
- a Shareholder who is entitled to cast two or more votes may appoint two proxies and may specify the proportion or number of votes each proxy is appointed to exercise. If the member appoints two proxies and the appointment does not specify the proportion or number of the member's votes, then in accordance with section 249X(3) of the Corporations Act, each proxy may exercise one-half of the votes.

Shareholders and their proxies should be aware that:

- if proxy holders vote, they must cast all directed proxies as directed; and
- any directed proxies which are not voted will automatically default to the Chair, who must vote the proxies as directed.

#### Voting in person

To vote in person, attend the Meeting at the time, date and place set out above.

Should you wish to discuss the matters in this Notice of Meeting please do not hesitate to contact the Company Secretary on +61 8 6374 1550.

### EXPLANATORY STATEMENT

This Explanatory Statement has been prepared to provide information which the Directors believe to be material to Shareholders in deciding whether or not to pass the Resolutions.

#### 1. **RESOLUTIONS 1 AND 2 – RATIFICATION OF PRIOR ISSUE OF PLACEMENT SECURITIES**

#### 1.1 Overview of the Placement

On 13 December 2024, the Company announced that it had successfully raised approximately A\$13.5 million (before costs) via a two-tranche placement of fully paid ordinary shares in the Company at an issue price of A\$0.028 (**Placement Shares**) to existing institutional and sophisticated investors (**Placement**). Under the Placement, participants were also offered free-attaching options, with an exercise price of A\$0.042 and expiry date of 31 December 207 (**Placement Options**), at a ratio of 1 Placement Option for every 2 Placement Shares.

The Company raised approximately A\$5.2 in the first tranche of the Placement which completed on Friday, 20 December 2024 (**Tranche 1**) via the issue of 33,118,053 Placement Shares in reliance on Listing Rule 7.1, 152,596,232 Placement Shares in reliance on Listing Rule 7.1A and 92,857,143 Placement Options in reliance on Listing Rule 7.1.

The proposed issue of Placement Securities in the second tranche of the Placement is subject to obtaining Shareholder approval, which is being sought under Resolutions 3, 4, 5 and 6. Subject to the approval of Shareholders, the Company will issue Placement Shares and Placement Options to Flat Footed and certain Directors to raise approximately A\$8.3 million under Tranche 2. In consideration for Flat Footed agreeing to cornerstone the Placement, the Company has also agreed to issue 20,000,000 Cornerstone Options (on identical terms to the Placement Options), subject to Shareholder approval.

#### 1.2 Listing Rules 7.1 and 7.1A

Shareholder approval is being sought under Resolution 1 to approve and ratify the prior issue and allotment of:

- (a) 152,596,232 Placement Shares under ASX Listing Rule 7.1A; and
- (b) 33,118,053 Placement Shares and 92,857,143 Placement Options under Listing Rule 7.1.

Broadly speaking, and subject to a number of exceptions, Listing Rule 7.1 limits the amount of equity securities that a listed company can issue without the approval of its shareholders over any 12 month period to 15% of the fully paid ordinary securities it had on issue at the start of that period.

At the Company's annual general meeting held on 28 November 2024, the Company sought and obtained the approval of Shareholders under Listing Rule 7.1A to increase this 15% limit by an extra 10% to 25%.

The issue of Placement Shares and Placement Options in Tranche 1 did not fit within any of the exceptions (to ASX Listing Rules 7.1 and 7.1A) and, as it has not been approved by the Company's Shareholders, it effectively uses up part of the expanded 25% limit in Listing Rule 7.1 and 7.1A, reducing the Company's capacity to issue further equity securities without Shareholder approval under Listing Rule

7.1 and 7.1A for the 12 month period following the Tranche 1 issue date (noting that the extra 10% under Listing Rule 7.1A will expire unless reapproved by the Company's Shareholders on an annual basis).

Listing Rule 7.4 allows the shareholders of a listed company to approve an issue of equity securities after it has been made or agreed to be made. If they do, the issue is taken to have been approved under Listing Rule 7.1 and so does not reduce the Company's capacity to issue further equity securities without Shareholder approval under Listing Rule 7.1.

A note to Listing Rule 7.4 also provides that an issue made in accordance with Listing Rule 7.1A can be approved subsequently under Listing Rule 7.4 and, if it is, the issue will then be excluded from variable "E" in Listing Rule 7.1A.2 (which means that the Company's capacity to issue further equity securities without Shareholder approval under Listing Rule 7.1A is not reduced).

The Company wishes to retain as much flexibility as possible to issue additional equity securities into the future without having to obtain Shareholder approval for such issues under Listing Rule 7.1. To this end, this Resolution seeks Shareholder approval to subsequently approve the issue of Placement Shares for the purposes of Listing Rule 7.4.

If Resolution 1 is passed, the issue of Placement Shares and Placement Options in Tranche 1 will be excluded in calculating the Company's 25% capacity to issue equity securities under ASX Listing Rules 7.1 (15%) and 7.1A (10%) without Shareholder approval over the 12 month period following the Issue Date.

If Resolution 1 is not passed, the issue of Placement Shares and Placement Options in Tranche 1 will be included in calculating the Company's 25% capacity to issue equity securities under ASX Listing Rules 7.1 (15%) and 7.1A (10%) without Shareholder approval over the 12 month period following the Issue Date.

#### 1.3 Information required by ASX Listing Rule 7.5

The following information is provided to Shareholders for the purposes of Listing Rule 7.5.

- (a) The Placement Shares and Placement Options were issued under Tranche 1 to institutional and other sophisticated investors. In accordance with ASX Guidance Note 21, the Company confirms that the following Tranche 1 investors were substantial shareholders of the Company who were issued more than 1% of the Company's issued capital:
  - (i) Paradice Investment Management Pty Ltd; and
  - (ii) CI Global Asset Management.

Other than above, the Company confirms that no other Tranche 1 investors were:

- (i) related parties of the Company, members of the Company's Key Management Personnel, substantial holders of the Company, advisers of the Company or an associate of any of these parties; and
- (ii) issued more than 1% of the issued capital of the Company.
- (b) The Company issued:

- (i) 152,596,232 Placement Shares under ASX Listing Rule 7.1A; and
- (ii) 33,118,053 Placement Shares and 92,857,143 Placement Options under ASX Listing Rule 7.1.
- (c) The Placement Shares were fully paid on issue and ranked equally in all aspects with all existing fully paid ordinary shares previously issued by the Company. The terms of the Placement Options are set out in Schedule 1.
- (d) The Placement Shares and Placement Options under Tranche 1 were issued on 20 December 2024.
- (e) Each of the Placement Shares were issued at an issue price of A\$0.028 per Placement Share, with one free-attaching Placement Option for every 2 Placement Shares issued.
- (f) Funds raised from the issue of the Placement Shares have been and will be used by the Company as follows:
  - (i) Nifty site costs;
  - (ii) permit support and DFS preparation and costs;

tenement maintenance and geology work;

- (iii) financing costs associated with the MetalsX convertible notes and Glencore facility; and
- (iv) working capital and costs of the Placement.

#### 1.4 Board recommendation

The Board recommends that Shareholders vote in favour of Resolutions 1 and 2.

# 2. RESOLUTION 3 – APPROVAL FOR THE ISSUE OF CORNERSTONE SECURITIES UNDER LISTING RULE 7.1

#### 2.1 General

As announced by the Company on 13 December 2024, Flat Footed (or its nominees) agreed to cornerstone the Placement by subscribing for 190,346,000 Placement Shares at an issue price of A\$0.028 per Share (to raise circa. A\$5.3 million) and free attaching 148,744,427 Placement Options under Tranche 2 of the Placement.

In consideration for Flat Footed agreeing to cornerstone the Placement, subject to Shareholder approval being obtained, the Company agreed to issue 20,000,000 Cornerstone Options (on identical terms to the Placement Options) to Flat Footed (or its nominees).

Resolution 3 seeks Shareholder approval to issue and allot the following

Securities to Flat Footed (or its nominees):

- (a) 190,346,000 Placement Shares;
- (b) 95,173,000 Placement Options; and
- (c) 20,000,000 Cornerstone Options,

(collectively the Cornerstone Securities).

As summarised in Section 1.2 above, Listing Rule 7.1 limits the amount of equity securities that a listed company can issue without the approval of its shareholders over any 12 month period to 15% of the fully paid ordinary shares it had on issue at the start of that period.

The proposed issue of the Cornerstone Securities does not fall within any of the exceptions set out in Listing Rule 7.2 and exceeds the 15% limit in Listing Rule 7.1. It therefore requires the approval of Shareholders under Listing Rule 7.1.

If the exercise of any Placement Option or Cornerstone Option by Flat Footed would result in it acquiring a relevant interest in 20% or more of the Shares, the Company must first convene a general meeting to seek Shareholder approval in accordance with section 611 of the Corporations Act (unless another exemption exists) before those Placement Options or Cornerstone Options can be exercised, so as to avoid the 20% relevant interest threshold being exceeded.

#### 2.2 Technical information required by Listing Rule 14.1A

If Resolution 3 is passed, the Company will be able to proceed with the issue of the Cornerstone Securities, which forms part of Tranche 2 of the Placement. In addition, the issue of the Cornerstone Securities will be excluded from the calculation of the number of equity securities that the Company can issue without Shareholder approval under Listing Rule 7.1.

If Resolution 3 is not passed, the Company will not be able to proceed with the issue of the Cornerstone Securities, which forms part of Tranche 2 of the Placement. If Tranche 2 of the Placement is not completed, this may have significant adverse commercial and financial consequences on the Company.

Resolution 3 seeks Shareholder approval for the purposes of Listing Rule 7.1 for the issue of the Cornerstone Securities.

#### 2.3 Technical information required by Listing Rule 7.1

Pursuant to and in accordance with Listing Rule 7.3, the following information is provided in relation to Resolution 3.

- (a) The Cornerstone Securities will be issued to Flat Footed, a substantial holder in the Company.
- (b) The number of Placement Shares to be issued to Flat Footed in Tranche 2 is more than 1% of the issued Share capital of the Company, however Flat Footed will not exceed 19.99% of the issued Share capital of the Company.
- (c) The maximum number of Cornerstone Securities to be issued is as follows:
  - (i) 190,346,000 Placement Shares;
  - (ii) 95,173,000 Placement Options; and
  - (iii) 20,000,000 Cornerstone Options.
- (d) The Placement Shares will be fully paid on issue and rank equally in all aspects with all existing fully paid ordinary shares previously issued by the Company. The terms of the Placement Options and Cornerstone Options are identical and set out in Schedule 1.
- (e) The Cornerstone Securities will be issued no later than 3 months after the date of the Meeting (or such later date to the extent permitted by any ASX waiver or modification of the Listing Rules) and it is intended that the issue of the Cornerstone Securities will occur on the same date;

- (f) The issue price of the Placement Shares will be A\$0.028 per Placement Share, with one free-attaching Placement Option for every 2 Placement Shares issued. There is no issue price for the Cornerstone Options, which is proposed to be issue as consideration for Flat Footed in consideration for being the cornerstone investor to the Placement.
- (g) The purpose of the issue of the Tranche 2 Cornerstone Securities is to raise capital and complete Tranche 2 of the Placement, which the Company intends to apply in the manner set out in Section 1.3(f).
- (h) Other than as disclosed in this Resolution, there are no other material terms of the agreement between the Company and Flat Footed with respect to the issue of the Tranche 2 Cornerstone Securities.
- (i) The Tranche 2 Cornerstone Securities are not being issued under, or to fund, a reverse takeover.

#### 2.4 Board recommendation

The Board recommends that Shareholders vote in favour of Resolution 3.

# 3. RESOLUTIONS 4 TO 6 – APPROVAL FOR THE ISSUE OF TRANCHE 2 PLACEMENT SECURITIES TO DIRECTORS (OR THEIR NOMINEES) UNDER LISTING RULE 10.11

#### 3.1 General

As announced by the Company on 13 December 2024, a majority of the Directors (or their nominees), subject to Shareholder approval being obtained, agreed to participate in the Placement under Tranche 2 in the aggregate amount of A\$3.0 million.

Resolutions 4 to 6 seek Shareholder approval for the Director participation in the Placement as follows:

- (a) Resolution 4: Mr J Matthew Fifield, Executive Chairman (or his nominees) for A\$1 million, being 35,714,285 Placement Shares and 17,857,142 Placement Options;
- (b) Resolution 5: Mr Scott Perry, Non-Executive Director (or his nominees) for A\$1.5 million, being 53,571,428 Placement Shares and 26,785,714 Placement Options; and
- (c) Resolution 6: Mr Ross Bhappu, Non-Executive Director (or his nominees) for A\$500K, being 17,857,142 Placement Shares and 8,928,571 Placement Options.

#### 3.2 Chapter 2E of the Corporations Act

For a public company, or an entity that the public company controls, to give a financial benefit to a related party of the public company, the public company or entity must:

- (a) obtain the approval of the public company's members in the manner set out in sections 217 to 227 of the Corporations Act; and
- (b) give the benefit within 15 months following such approval,

unless the giving of the financial benefit falls within an exception set out in sections 210 to 216 of the Corporations Act.

The proposed issue of Placement Shares and Placement Options (which are types of an equity security, for the purposes of the Chapter 2E of the Corporations Act) constitutes the giving of a financial benefit.

A "related party" for the purposes of the Corporations Act and the Listing Rules is widely defined and includes a director of a public company, a spouse of a director of a public company or an entity controlled by a director of a public company. The definition of "related party" also includes a person whom there is reasonable grounds to believe will become a "related party" of a public company.

As the Placement Shares and Placement Options are on the same terms as offered to non-related parties under the Placement, the Company relies on the "arm's length terms" exception as set out in section 210 of the Corporations Act for the purposes of these Resolutions and therefore does not seek Shareholder approval for the purposes of Chapter 2E of the Corporations Act.

#### 3.3 Listing Rule 10.11

Listing Rule 10.11 provides that unless one of the exceptions in Listing Rule 10.12 applies, a listed company must not issue or agree to issue equity securities to:

- 10.11.1 a related party;
- 10.11.2 a person who is, or was at any time in the 6 months before the issue or agreement, a substantial (30%+) holder in the company;
- 10.11.3 a person who is, or was at any time in the 6 months before the issue or agreement, a substantial (10%+) holder in the company and who has nominated a director to the board of the company pursuant to a relevant agreement which gives them a right or expectation to do so;
- 10.11.4 an associate of a person referred to in Listing Rules 10.11.1 to 10.11.3; or
- 10.11.5 a person whose relationship with the company or a person referred to in Listing Rules 10.11.1 to 10.11.4 is such that, in ASX's opinion, the issue or agreement should be approved by its shareholders,

unless it obtains the approval of its shareholders.

The Participation falls within Listing Rule 10.11.1 and does not fall within any of the exceptions in Listing Rule 10.12. It therefore requires the approval of Shareholders under Listing Rule 10.11.

Resolutions 4 to 6 seek the required Shareholder approval for the Participation under and for the purposes of Listing Rule 10.11.

#### 3.4 Technical information required by Listing Rule 14.1A

If Resolutions 4 to 6 are passed, the Company will be able to proceed with the issue of the Placement Shares and Placement Options, which forms part of Tranche 2 of the Placement, within one month after the date of the Meeting.

As approval pursuant to Listing Rule 7.1 is not required as approval is being sought under Listing Rule 10.11, the issue of the Placement Shares and Placement Options under Resolutions 4 to 6 will not use up any of the Company's 15% annual placement capacity. If Resolutions 4 to 6 are not passed, the Company will not be able to proceed with the issue of the Placement Shares and Placement Options, which forms part of Tranche 2 of the Placement. If Tranche 2 of the Placement is not completed, this may have significant adverse commercial and financial consequences on the Company.

#### 3.5 Technical information required by Listing Rule 10.13

Pursuant to and in accordance with Listing Rule 10.13, the following information is provided in relation to Resolutions 4 to 6:

- (a) The recipients are as follows:
  - (i) Resolution 4: Mr J Matthew Fifield, Executive Chairman (or his nominees);
  - (ii) Resolution 5: Mr Scott Perry, Non-Executive Director (or his nominees); and
  - (iii) Resolution 6: Mr Ross Bhappu, Non-Executive Director (or his nominees).
- (b) Each of Messrs Fifield, Perry and Bhappu satisfy Listing Rule 10.11.1 as they are current Directors of the Company, therefore, related parties.
- (c) The number of Securities to be issued are as follows:
  - (i) Resolution 4: 35,714,285 Placement Shares with free-attaching 17,857,142 Placement Options to Mr Fifield (or his nominees);
  - (ii) Resolution 5: 53,571,428 Placement Shares with free-attaching 26,785,714 Placement Options to Mr Perry (or his nominees); and
  - (iii) Resolution 6: 17,857,142 Placement Shares with free-attaching 8,928,571 Placement Options to Mr Bhappu (or his nominees).
- (d) The Placement Shares will be fully paid ordinary shares in the capital of the Company and be issued on the same terms and conditions as the Company's existing fully paid ordinary shares.
- (e) The terms and conditions of the Placement Options are set out in Schedule 1.
- (f) The Placement Shares and Placement Options will be issued no later than 1 month after the date of the Meeting and it is intended that issue of the Placement Shares and Placement Options will occur on the same date.
- (g) The Placement Shares will be issued at an issue price of A\$0.028 per Share. There will be no issue price for the Placement Options, which are free-attaching options.
- (h) The purpose of the issue of Placement Shares and Placement Options is to raise capital and complete Tranche 2of the Placement. The funds raised will be put towards the activities set out in Section 1.3(f).
- (i) The proposed issue of Placement Shares and Placement Options to Messrs Fifield, Perry and Bhappu (or their nominees) is not intended to remunerate or incentivise the Directors. The purpose was to raise capital on the same terms as non-related parties under the Placement.
- (j) Other than as disclosed in these Resolutions, there are no other material terms of the agreement between the Company and investors (including the Directors) with respect to the issue of the Placement Shares and Placement Options.

#### 4. RESOLUTION 7 – APPROVAL FOR THE ISSUE OF PERFORMANCE RIGHTS TO MR J MATTHEW FIFIELD (EXECUTIVE CHAIR) UNDER LISTING RULE 10.14

#### 4.1 General

As part of the FY25 long term incentives for Mr J Matthew Fifield, Executive Chairman, the Company seeks Shareholder approval, for the purposes of Listing Rule 10.14 to issue 30,000,000 performance rights under the Company's Employee Securities Incentive Plan (Incentive Plan), on the terms and conditions set out below.

Under the Incentive Plan, the Board has discretion to grant performance rights to any employee it declares to be an eligible employee, upon the terms set out in the Incentive Plan (and upon such terms and conditions as the Board determines).

The proposed grant of performance rights to Mr Fifield (or his nominees) seeks to further align his interests with those of Shareholders by linking Mr Fifield's rewards to long term performance for Shareholders by imposing performance-related conditions.

Summary of the material terms of the performance rights are as follows:

- Vesting Period: From 1 January 2025 to 31 December 2028
- Vesting Conditions:
  - Tranche 1: 7,350,000 rights vesting on 60-day VWAP of \$0.05 during Vesting Period
  - Tranche 2: 7,350,000 rights vesting on 30-day VWAP of \$0.07 during Vesting Period
  - Tranche 3: 6,300,000 rights vesting on 30-day VWAP of \$0.11 during Vesting Period
  - Tranche 4: 4,500,000 rights vesting on delivery of feasibility study on concentrate project
  - Tranche 5: 4,500,000 rights vesting on board Final Investment Decision of either cathode or concentrate project
- Minimum Service Condition: 2 years
- Acceleration: All rights whether vested or not accelerate and fully vest on:
  - o 60-day VWAP of \$0.13
  - Change of control
- Expiry: 4 years from the date of issue

#### 4.2 Chapter 2E of the Corporations Act

For a public company, or an entity that the public company controls, to give a financial benefit to a related party of the public company, the public company or entity must:

(a) obtain the approval of the public company's members in the manner set out in sections 217 to 227 of the Corporations Act; and

(b) give the benefit within 15 months following such approval,

unless the giving of the financial benefit falls within an exception set out in sections 210 to 216 of the Corporations Act.

The proposed issue of performance rights (which is a type of an equity security, for the purposes of the Chapter 2E of the Corporations Act) constitutes the giving of a financial benefit.

A "related party" for the purposes of the Corporations Act and the Listing Rules is widely defined and includes a director of a public company, a spouse of a director of a public company or an entity controlled by a director of a public company. The definition of "related party" also includes a person whom there is reasonable grounds to believe will become a "related party" of a public company.

An exception to the requirement to obtain shareholder approval in accordance with Chapter 2E applies where the financial benefit constitutes part of the related party's "reasonable remuneration".

The Board (other than Mr Fifield) considers that the grant of performance rights to Mr Fifield, and any issue of Shares upon the vesting and exercise of the performance rights], constitutes part of the "reasonable remuneration" payable to Mr Fifield and, accordingly, approval under Chapter 2E of the Corporations Act is not being sought. In reaching this conclusion, the Board has had regard to a variety of factors including market practice and the remuneration offered to persons in comparable positions at comparable companies.

#### 4.3 Listing Rule 10.14

Listing Rule 10.14 provides that a listed company must not permit any of the following persons to acquire equity securities under an employee incentive scheme:

- 10.14.1 a director of the company;
- 10.14.2 an associate of a director of the company; or
- 10.14.3 a person whose relationship with the company or a person referred to in Listing Rule 10.14.1 or 10.14.2 is such that, in ASX's opinion, the acquisition should be approved by its shareholders,

Unless it obtains the approval of its shareholders.

The proposed issue of performance rights to Mr Fifield, Executive Chairman, falls within Listing Rule 10.14.1 above and therefore requires the approval of the Company's Shareholders under Listing Rule 10.14.

Resolution 7 seeks the required Shareholder approval to issue the performance rights under and for the purposes of Listing Rule 10.14.

#### 4.4 Technical information required by Listing Rule 14.1A

If Resolution 7 is passed, the Company will be able to proceed with the issue of performance rights to Mr Fifield as part of this FY25 long term incentives.

If Resolution 7 is not passed, the Company will not be able to proceed with the issue of performance rights to Mr Fifield as part of his FY25 long term incentives, and therefore, the Board may be required to propose alternative remuneration

structures to continue to incentivise and remunerate Mr Fifield, which may not be as cash efficient for the Company.

#### 4.5 Technical information required by Listing Rule 10.15

Pursuant to and in accordance with Listing Rule 10.15, the following information is provided in relation to Resolution 7.

- (a) The recipient is Mr J Matthew Fifield, Executive Chairman (or his nominees).
- (b) Mr Fifield satisfies Listing Rule 10.14.1 as he is a current Director of the Company.
- (c) The Company proposes to issue 30,000,000 performance rights under the Company's Incentive Plan to Mr Fifield.
- (d) Mr Fifield's current total remuneration package is \$480,000 (excluding super) per annum.
- (e) No other incentive securities have been previously granted to Mr Fifield under the Incentive Plan.
- (f) A summary of the material terms of the performance rights are set out in Section 4.1. The Board considered that a performance right (as part of the FY25 long term incentives) was an appropriate security as the performance hurdles assist in further aligning the interests of Mr Fifield with Shareholders of the Company. Based on a Black Scholes model, value of the performance rights is \$500,460.

The breakdown between the tranches is as follows: Tranche 1: \$139,194, Tranche 2: \$130,273, Tranche 3: \$98,411, Tranche 4: \$66,291 and Tranche 5: \$66,291.

- (g) If Shareholder approval is obtained, it is anticipated that the performance rights will be issued shortly after the Meeting and in any event, no later than 3 years after the date of the Meeting.
- (h) The performance rights be issued for nil consideration and no cash amount will be payable to the Company upon vesting and/or exercise.
- (i) A summary of the material terms of the Employee Securities Incentive Plan is set out in Schedule 2.
- (j) No loans will be made by the Company to Mr Fifield in relation to the issue of performance rights.
- (k) Details of any equity securities issued under the Incentive Plan will be published in the annual report of the Company relating to the period in which they were issued, along with a statement that approval for the issue was obtained under Listing Rule 10.14.
- (I) Any additional persons covered by Listing Rule 10.14 who become entitled to participate in an issue of Equity Securities under the Incentive Plan and who were not named in this Notice will not participate until approval is obtained under Listing Rule 10.14.

#### 4.6 Board recommendation

The Board (other than Mr Fifield) recommends that Shareholders vote in favour of Resolution 7.

#### 5. RESOLUTION 8 – APPROVAL OF POTENTIAL TERMINATION BENEFITS TO MR J MATTHEW FIFIELD (EXECUTIVE CHAIR) UNDER EMPLOYEE SECURITIES INCENTIVE PLAN

#### 5.1 General

Resolution 8 seeks Shareholder approval in accordance with Part 2D.2 of the Corporations Act (including sections 200B and 200E of the Corporations Act), Listing Rule 10.19 for the Company to give certain termination benefits to Mr J Matthew Fifield in connection with him ceasing to be an officer of, or ceasing to hold a managerial or executive office in, the Company or a related body corporate.

The benefits for which approval is being sought under Resolution 8 include the benefits that may result from automatic/accelerated vesting of performance rights or from the Board exercising discretions conferred under the Incentive Plan in relation to the performance rights (together, the **Potential Termination Benefits**).

#### 5.2 Section 200B of the Corporations Act

The Corporations Act restricts the benefits which can be given to individuals who hold a managerial or executive office (as defined in the Corporations Act) in connection with the retirement from their position in the Company or its related bodies corporate.

In accordance with section 200B of the Corporations Act, to give a benefit in connection with a relevant person's retirement from an office, the Company must, subject to various exceptions, obtain the approval of Shareholders in the manner set out in section 200E of the Corporations Act.

As stated above in Section 5.1, approval for the Potential Termination Benefits is being sought under Resolution 8, which includes the benefits that may result from automatic/accelerated vesting of performance rights or from the Board exercising discretions conferred under the Incentive Plan in relation to the Performance Rights.

#### 5.3 Specific information required by section 200E of the Corporations Act

The following additional information is provided for the purposes of obtaining Shareholder approval in respect of the Potential Termination Benefits payable to Mr J Matthew Fifield for the purposes of section 200E of the Corporations Act.

The amount or value of the benefit which may arise in connection with Mr J Matthew Fifield's retirement from a managerial or executive office cannot presently be ascertained. However, matters, events and circumstances that will, or are likely to affect the calculation of that amount or value include:

- (a) the number of performance rights held prior to ceasing employment or engagement with the Company;
- (b) the outstanding conditions (if any) of vesting of the performance rights and the number that the Board determines to (or which automatically) vest, lapse or leave on foot;
- (c) the applicable performance measures and the achievement of such measures;
- (d) the portion of the relevant performance rights that have expired at the time Mr J Matthew Fifield ceases employment or engagement;

- (e) the circumstances of, or reasons for, ceasing employment with the Company;
- (f) the length of service with the Company and performance over that period of time;
- (g) any other factors that the Board determines to be relevant when exercising its discretion to provide potential termination benefits to Mr J Matthew Fifield;
- (h) the market price of the Company's Shares on ASX at the relevant time when the amount or value of the Performance Rights is determined;
- (i) any changes in law; and
- (j) the risk-free rate of return in Australia and the estimated volatility of the Company's Shares on ASX at the relevant time.

The Company will likely calculate the value of the benefit at the relevant time based on the above factors and using the fair market value to value the performance rights.

As set out in the material terms in Section 4.1 of this Notice, vesting of any performance rights is also subject to a Minimum Service Condition, being that Mr J Matthew Fifield must be engaged by the Company for a period of at least 2 years from the commencement of the Vesting Period (1 January 2025). In addition, all rights whether vested or not accelerate and fully vest certain events, including change of control (which may result in Mr Fifield ceasing to be an officer of, or ceasing to hold a managerial or executive office in, the Company or a related body corporate). As provided for in the terms and conditions of the Incentive Plan, the Board has discretion to waive any Vesting Conditions applicable to the performance rights. In circumstances where Mr Fifield ceases to be an eligible participant under the Incentive Plan prior to the Minimum Service Condition being met and is considered by the Board (acting in good faith) to be a good leaver, the Minimum Service Condition may be waived by the Board (acting reasonably) and by reference to the length of service of Mr Fifield. In this instance, the Board may determine that a percentage of the performance rights remain on issue, subject to the terms and conditions of the performance rights (including any applicable and remaining Vesting Conditions). In these circumstances, any performance rights that the Board determines appropriate to remain on issue will, upon satisfaction of any Vesting Conditions met, vest and be able to be converted into Shares.

#### 5.4 Listing Rule 10.19

Listing Rule 10.19 provides that without approval of shareholders, an entity must ensure that no officer of the entity or any of its child entities will be, or may be, entitled to termination benefits if the value of those benefits and the termination benefits that may become payable to all officers together exceed 5% of the equity interests of the entity as set out in the latest accounts given to ASX under the Listing Rules. For the purpose of the Listing Rules, termination benefits include payments, property and advantages that are receivable on termination of employment, engagement or office, except those from any superannuation or provident fund and those required by law to be made. At this stage, it is uncertain if the value of the Potential Termination Benefits the subject of Resolution 8 may exceed this 5% threshold. If and when Listing Rule 10.19 is triggered, the Company will seek separate Shareholder approval at the time.

## 6. RESOLUTIONS 9 TO 12 – APPROVAL OF SALARY SACRIFICE OFFERS (FY25) TO DIRECTORS UNDER EMPLOYEE SECURITIES INCENTIVE PLAN

#### 6.1 General

As part of the Company's Incentive Plan, the Company wishes to provide an opportunity for the each of the Company's Directors to forgo some or all of their salary and/or Directors' fees to acquire Shares in the Company.

Importantly, this will not increase the face value of their salary and/or Directors' fees, but will assist the Company in preserving cash whilst also providing potential upside for the Directors (and further alignment with Shareholders).

#### 6.2 Chapter 2E of the Corporations Act

For a public company, or an entity that the public company controls, to give a financial benefit to a related party of the public company, the public company or entity must:

- (a) obtain the approval of the public company's members in the manner set out in sections 217 to 227 of the Corporations Act; and
- (b) give the benefit within 15 months following such approval,

unless the giving of the financial benefit falls within an exception set out in sections 210 to 216 of the Corporations Act.

The proposed issue of performance rights (which is a type of an equity security, for the purposes of the Chapter 2E of the Corporations Act) constitutes the giving of a financial benefit.

A "related party" for the purposes of the Corporations Act and the Listing Rules is widely defined and includes a director of a public company, a spouse of a director of a public company or an entity controlled by a director of a public company. The definition of "related party" also includes a person whom there is reasonable grounds to believe will become a "related party" of a public company.

An exception to the requirement to obtain shareholder approval in accordance with Chapter 2E applies where the financial benefit constitutes part of the related party's "reasonable remuneration".

The Board (other with each impacted Director excluded) considers that the grant of the salary sacrifice offer (FY25) and ability to subscribe for Purchased Rights, constitutes part of the "reasonable remuneration" payable to each of the Directors and, accordingly, approval under Chapter 2E of the Corporations Act is not being sought. In reaching this conclusion, the Board notes that each Director is not increasing their respective salary and/or Director's fees (which are at market rates previously set), but is being provided an opportunity to convert part or all of their cash remuneration to Purchased Rights (which can convert to Shares).

#### 6.3 Listing Rule 10.14

Listing Rule 10.14 provides that a listed company must not permit any of the following persons to acquire equity securities under an employee incentive scheme:

- 10.14.1 a director of the company;
- 10.14.2 an associate of a director of the company; or
- 10.14.3 a person whose relationship with the company or a person referred to in Listing Rule 10.14.1 or 10.14.2 is such that, in ASX's opinion, the acquisition should be approved by its shareholders,

Unless it obtains the approval of its shareholders.

The proposed issue of Purchased Rights to each of the Directors falls within Listing Rule 10.14.1 above and therefore requires the approval of the Company's Shareholders under Listing Rule 10.14.

Resolutions 9 to 12 seeks the required Shareholder approval to issue the Purchased Rights under and for the purposes of Listing Rule 10.14.

#### 6.4 Technical information required by Listing Rule 14.1A

If Resolutions 9 to 12 are passed, the Company will be able to complete the grant of the salary sacrifice offer (FY25) and ability for each of the Directors to subscribe for Purchased Rights, and accordingly, forgo some or all of their salary and/or Directors' fees to acquire Shares in the Company. This will assist the Company in preserving cash whilst also providing potential upside for the Directors (and further alignment with Shareholders).

If Resolutions 9 to 12 are not passed, the Company will not be able to complete the grant of the salary sacrifice offer (FY25) and ability for each of the Directors to subscribe for Purchased Rights. Accordingly, the Company will have to continue to pay cash remuneration to each of the Directors.

#### 6.5 Technical information required by Listing Rule 10.15

Pursuant to and in accordance with Listing Rule 10.15, the following information is provided in relation to Resolution 7.

- (a) The recipients are as follows:
  - (i) Resolution 9: Mr J Matthew Fifield, Executive Chairman;
  - (ii) Resolution 10: Mr Scott Perry, Non-Executive Director;
  - (iii) Resolution 11: Mr Ross Bhappu, Non-Executive Director; and
  - (iv) Resolution 12: Mr Gary Comb, Non-Executive Director.
- (b) Each of Messrs Fifield, Perry, Bhappu and Comb satisfy Listing Rule 10.14.1 as they are current Directors of the Company.
- (c) The maximum number of Purchased Rights each of the Directors will be eligible to receive will be determined by dividing the portion of their respective Sacrifice Amount contributed in a Contribution Period by the Allocation Price for the Contribution Period, which is calculated as the 90day volume weighted average price (VWAP) of a Share for the period ending on the last trading day of the Contribution Period.

NEDs must salary sacrifice at least 50% and up to 100% of their Director fees. Current NED fees is \$60,000 (excluding super) per annum. Therefore, for each of Messrs Perry, Bhappu and Comb, they must salary sacrifice at least \$30,000 and up to \$60,000 per annum.

The Executive Chair can salary sacrifice up to 50% of his salary (which includes Director fees). Current Executive Chair salary is \$480,000 (excluding super) per annum. Therefore, Mr Fifield can salary sacrifice up to \$240,000 per annum.

The Purchased Rights will be fully vested when granted to each recipient but will be subject to a 12-month Exercise Restriction.

The actual number of Purchased Rights that could be issued cannot be determined at this stage, however, the following table sets out the possible dilutionary impact on existing Shareholders of the Company (based on the issued share capital of the Company as at 20 December 2024) if the grant of all of the Purchased Rights the subject of Resolutions 9 to 12 are approved by Shareholders and they are all subsequently exercised by, and the underlying Shares are all issued to, the holders.

The figures below are indicative only as the average monthly VWAP prior to the Grant Date will not be known until the Grant Date.

|                         | Potential number of Purchased Rights that could be<br>issued (which then convert to Shares) – Minimum<br>salary sacrifice |           |           |         |
|-------------------------|---|-----------|-----------|---------|
| Example 90-<br>day VWAP | \$0.020   | \$0.024   | \$0.028   | \$0.032 |
| M Fifield               | Nil   | Nil       | Nil       | Nil     |
| S Perry                 | 1,500,000   | 1,250,000 | 1,071,429 | 937,500 |
|                         | (0.09%)   | (0.07%)   | (0.06%)   | (0.05%) |
| R Bhappu                | 1,500,000   | 1,250,000 | 1,071,429 | 937,500 |
|                         | (0.09%)   | (0.07%)   | (0.06%)   | (0.05%) |
| G Comb                  | 1,500,000   | 1,250,000 | 1,071,429 | 937,500 |
|                         | (0.09%)   | (0.07%)   | (0.06%)   | (0.05%) |

|                         | Potential number of Purchased Rights that could be<br>issued (which then convert to Shares) – Maximum<br>salary sacrifice |            |           |           |
|-------------------------|---|------------|-----------|-----------|
| Example 90-<br>day VWAP | \$0.020   | \$0.024    | \$0.028   | \$0.032   |
| M Fifield               | 12,000,000  | 10,000,000 | 8,571,429 | 7,500,000 |
|                         | (0.70%)   | (0.58%)    | (0.50%)   | (0.44%)   |
| S Perry                 | 3,000,000   | 2,500,000  | 2,142,857 | 1,875,000 |
|                         | (0.09%)   | (0.15%)    | (0.13%)   | (0.11%)   |
| R Bhappu                | 3,000,000   | 2,500,000  | 2,142,857 | 1,875,000 |
|                         | (0.09%)   | (0.15%)    | (0.13%)   | (0.11%)   |
| G Comb                  | 3,000,000   | 2,500,000  | 2,142,857 | 1,875,000 |
|                         | (0.09%)   | (0.15%)    | (0.13%)   | (0.11%)   |

- (d) Each of the current NED fees (Messrs Perry, Bhappu and Comb) is \$60,000 (excluding super) per annum. Mr Fifield's current total remuneration package is \$480,000 (excluding super) per annum.
- (e) No other incentive securities have been previously granted to any of the Directors under the current Incentive Plan.
- (f) A summary of the material terms of the Purchased Rights is set out in Schedule 3.

The Board considered that Purchased Rights an appropriate security as part of a salary sacrifice offer which could assist the Company in preserving cash whilst also providing potential upside for the Directors (and further alignment with Shareholders).

The value of the Purchased Right will be calculated in accordance with the 90-day VWAP, as it will be a direct conversion of the cash remuneration.

- (g) If Shareholder approval is obtained, it is anticipated that the Purchased Rights will be issued to each recipient on or about the day immediately following the end of each Contribution Period, subject to complying with the Company's Securities Trading Policy, in any event, no later than 3 years after the date of the Meeting.
- (h) The Purchased Rights be issued for a deemed issue price which is calculated in accordance with the 90-day VWAP.
- (i) A summary of the material terms of the Incentive Plan is set out in Schedule 2.
- (j) No loans will be made by the Company to any of the Directors in relation to the issue of Purchased Rights.
- (k) Details of any equity securities issued under the Incentive Plan will be published in the annual report of the Company relating to the period in which they were issued, along with a statement that approval for the issue was obtained under Listing Rule 10.14.
- (I) Any additional persons covered by Listing Rule 10.14 who become entitled to participate in an issue of Equity Securities under the Incentive Plan and who were not named in this Notice will not participate until approval is obtained under Listing Rule 10.14.

#### 7. **RESOLUTION 13 – AMENDMENT TO CONSTITUTION**

#### 7.1 Background

A company may modify or repeal its constitution or a provision of its constitution by special resolution of Shareholders.

Resolution 13 is a special resolution which will enable the Company to amend its existing Constitution (**Amended Constitution**) to permit the use of technology in hybrid meetings, ensure consistency between the ASX Listing Rules such as ASX Listing Rule 14.4, and promote enhanced governance in the administration of meetings.

#### The Company will <u>not</u> be seeking approval to hold virtual only meetings. Amendments are being sought to allow for hybrid and in-person meetings only.

A copy of the Amended Constitution is attached as Schedule 4, and separately, is available for review by Shareholders at the office of the Company. A copy of the Amended Constitution can also be sent to Shareholders upon request to the Company Secretary. Shareholders are invited to contact the Company if they have any queries or concerns.

#### 7.2 Board recommendation

The Board recommends that Shareholders vote in favour of Resolution 13.

### GLOSSARY

**\$** means Australian dollars.

7.1A Mandate has the meaning given in Section 3.1.

ASIC means the Australian Securities & Investments Commission.

**ASX** means ASX Limited (ACN 008 624 691) or the financial market operated by ASX Limited, as the context requires.

**Board** means the current board of directors of the Company.

**Business Day** means Monday to Friday inclusive, except New Year's Day, Good Friday, Easter Monday, Christmas Day, Boxing Day, and any other day that ASX declares is not a business day.

Chair means the chair of the Meeting.

Closely Related Party of a member of the Key Management Personnel means:

- (a) a spouse or child of the member;
- (b) a child of the member's spouse;
- (c) a dependent of the member or the member's spouse;
- (d) anyone else who is one of the member's family and may be expected to influence the member, or be influenced by the member, in the member's dealing with the entity;
- (e) a company the member controls; or
- (f) a person prescribed by the Corporations Regulations 2001 (Cth) for the purposes of the definition of 'closely related party' in the Corporations Act.

**Company** means Cyprium Metals Limited (ACN 002 678 640).

Constitution means the Company's constitution.

**Cornerstone Option** means an Option with identical terms to a Placement Option offered to Flat Footed in consideration for being the cornerstone investor to the Placement.

**Cornerstone Securities** means 190,346,000 Placement Shares, 95,173,000 Placement Options and 20,000,000 Cornerstone Options proposed to be issued to Flat Footed, subject to Shareholder approval.

Corporations Act means the Corporations Act 2001 (Cth).

Directors means the current directors of the Company.

**Employee Securities Incentive Plan** means the employee incentive of the Company readopted by Shareholders at the annual general meeting of the Company held on 28 May 2024.

**Equity Securities** includes a Share, a right to a Share or Option, an Option, a convertible security and any security that ASX decides to classify as an Equity Security.

**Explanatory Statement** means the explanatory statement accompanying the Notice.

Extraordinary General Meeting or Meeting means the meeting convened by the Notice.

Flat Footed means Flat Footed LLC.

**Key Management Personnel** has the same meaning as in the accounting standards issued by the Australian Accounting Standards Board and means those persons having authority and responsibility for planning, directing and controlling the activities of the Company, or if the Company is part of a consolidated entity, of the consolidated entity, directly or indirectly, including any director (whether executive or otherwise) of the Company, or if the Company is part of a consolidated entity, of an entity within the consolidated group.

Listing Rules means the Listing Rules of ASX.

**Notice** or **Notice of Meeting** means this notice of meeting including the Explanatory Statement and the Proxy Form.

**Option** means an option to acquire a Share.

**Placement** means the two-tranche placement to raise up to A\$13.5 million as announced by the Company on Friday, 13 December 2024.

**Placement Option** means a free-attaching Option offered under the Placement, the terms of which are in Schedule 1, on the basis of 1 Placement Option for every 2 Placement Shares.

Placement Share means a Share subscribed for by investors under the Placement.

**Performance Right** means a right to acquire a Share.

**Proxy Form** means the proxy form accompanying the Notice.

**Resolutions** means the resolutions set out in the Notice, or any one of them, as the context requires.

Schedule means a schedule to this Notice.

Section means a section of the Explanatory Statement.

Share means a fully paid ordinary share in the capital of the Company.

Shareholder means a registered holder of a Share.

WST means Western Standard Time as observed in Perth, Western Australia.

### SCHEDULE 1 – TERMS OF PLACEMENT AND CORNERSTONE OPTIONS

#### a) Entitlement

Each Option entitles the holder to subscribe for one Share upon exercise of the Option.

#### b) Exercise Price

Subject to paragraph (j), the amount payable upon exercise of each Option will be A\$0.042 (**Exercise Price**).

#### c) Expiry Date

Each Option will expire at 5:00 pm (WST) on 31 December 2027 (**Expiry Date**). An Option not exercised before the Expiry Date will automatically lapse on the Expiry Date.

#### d) Exercise Period

The Options are exercisable at any time on or prior to the Expiry Date (**Exercise Period**).

#### e) Notice of Exercise

The Options may be exercised during the Exercise Period by notice in writing to the Company in the manner specified on the Option certificate (**Notice of Exercise**) and payment of the Exercise Price for each Option being exercised in Australian currency by electronic funds transfer or other means of payment acceptable to the Company.

#### f) Exercise Date

A Notice of Exercise is only effective on and from the later of the date of receipt of the Notice of Exercise and the date of receipt of the payment of the Exercise Price for each Option being exercised in cleared funds (**Exercise Date**).

#### g) Required Approvals on Exercise

Notwithstanding any other paragraph in this Schedule, the Options may not be converted into Shares in accordance with this Schedule if such exercise would cause the holder or the Company to breach any applicable law, including under the Corporations Act and/or the ASX Listing Rules, and the Company and the holder shall use best endeavours to obtain any required approvals (including Shareholder approvals) to allow exercise of the Options.

#### h) Timing of issue of Shares on exercise

Within 5 Business Days after the Exercise Date, the Company will:

- i. issue the number of Shares required under these terms and conditions in respect of the number of Options specified in the Notice of Exercise and for which cleared funds have been received by the Company;
- ii. give ASX a notice that complies with section 708A(5)(e) of the Corporations Act, or, if the Company is unable to issue such a notice, lodge with ASIC a prospectus prepared in accordance with the Corporations Act and do all such things necessary to satisfy section 708A(11) of the Corporations Act to ensure that an offer for sale of the Shares does not require disclosure to investors; and
- iii. apply for official quotation on ASX of Shares issued pursuant to the exercise of the Options.

#### i) Shares issued on exercise

Shares issued on exercise of the Options rank equally with the then issued shares of the Company.

#### j) Reconstruction of capital

If at any time the issued capital of the Company is reconstructed, all rights of an Optionholder are to be changed in a manner consistent with the Corporations Act and the ASX Listing Rules at the time of the reconstruction, subject to the provisions with respect to rounding of entitlements as sanctioned by a meeting of Shareholders approving a reconstruction of capital, in all other respects the terms of exercise of the Options will remain unchanged.

#### k) Participation in new issues

There are no participation rights or entitlements inherent in the Options and holders will not be entitled to participate in new issues of capital offered to Shareholders during the currency of the Options without exercising the Options. The Company will, where required pursuant to the ASX Listing Rules, provide Optionholders with notice prior to the books record date (to determine entitlements to any new issue of Securities made to Shareholders generally) to exercise the Options, in accordance with the requirements of the ASX Listing Rules.

#### I) Change in exercise price

Subject to paragraph (j), an Option does not confer the right to a change in Exercise Price or a change in the number of underlying securities over which the Option can be exercised.

#### m) Transferability

The Options are transferable subject to any restriction or escrow arrangements imposed by ASX or under applicable Australian securities laws.

#### n) Quotation

Notwithstanding paragraph (h)(iii), the Options are unquoted and application will not be made to the ASX for official quotation of the Options.

## SCHEDULE 2 – KEY TERMS OF EMPLOYEE SECURITIES INCENTIVE PLAN

A summary of the terms of the Company's Employee Securities Incentive Plan (Incentive Plan) is set out below.

| Eligible Participant                       | Eligible Participant means a person who is a full-time or part-<br>time employee, officer, or contractor of the Company, or an<br>Related Body Corporate, or such other person who has been<br>determined by the Board to be eligible to participate in the<br>Plan from time to time. The Company will seek Shareholder<br>approval for Director and related party participation in<br>accordance with Listing Rule 10.14.   |
|--|---|
| Purpose                                    | The purpose of the Plan is to:<br>(a) assist in the reward, retention and motivation of Eligible<br>Participants;<br>(b) link the reward of Eligible Participants to Shareholder value<br>creation; and<br>(c) align the interests of Eligible Participants with shareholders<br>of the Group (being the Company and each of its<br>Associated Bodies Corporate), by providing an opportunity to<br>Eligible Participants to receive an equity interest in the<br>Company in the form of Securities.                            |
| Plan administration                        | The Plan will be administered by the Board. The Board may<br>exercise any power or discretion conferred on it by the Plan<br>rules in its sole and absolute discretion. The Board may<br>delegate its powers and discretion.  |
| Eligibility, invitation<br>and application | The Board may from time to time determine that an Eligible<br>Participant may participate in the Plan and make an<br>invitation to that Eligible Participant to apply for Securities on<br>such terms and conditions as the Board decides.<br>On receipt of an Invitation, an Eligible Participant may apply<br>for the Securities the subject of the invitation by sending a<br>completed application form to the Company. The Board may<br>accept an application from an Eligible Participant in whole or<br>in part.         |
|  | If an Eligible Participant is permitted in the invitation, the<br>Eligible Participant may, by notice in writing to the Board,<br>nominate a party in whose favour the Eligible Participant<br>wishes to renounce the invitation.   |
| Grant of Securities                        | The Company will, to the extent that it has accepted a duly<br>completed application, grant the Participant the relevant<br>number of Securities, subject to the terms and conditions set<br>out in the invitation, the Plan rules and any ancillary<br>documentation required.   |
| Terms of Convertible<br>Securities         | Securities Each 'Convertible Security' represents a right to<br>acquire one or more Shares (for example, under an option or<br>performance right), subject to the terms and conditions of the<br>Plan. Prior to a Convertible Security being exercised a<br>Participant does not have any interest (legal, equitable or<br>otherwise) in any Share the subject of the Convertible Security<br>by virtue of holding the Convertible Security. A Participant<br>may not sell, assign, transfer, grant a security interest over or |

|  | otherwise deal with a Convertible Security that has been<br>granted to them unless otherwise determined by the Board. A<br>Participant must not enter into any arrangement for the<br>purpose of hedging their economic exposure to a<br>Convertible Security that has been granted to them.   |
|--|--|
| Vesting of Convertible<br>Securities                           | Any vesting conditions applicable to the grant of Convertible<br>Securities will be described in the invitation. If all the vesting<br>conditions are satisfied and/or otherwise waived by the<br>Board, a vesting notice will be sent to the Participant by the<br>Company informing them that the relevant Convertible<br>Securities have vested. Unless and until the vesting notice is<br>issued by the Company, the Convertible Securities will not be<br>considered to have vested. For the avoidance of doubt, if<br>the vesting conditions relevant to a Convertible Security are<br>not satisfied and/or otherwise waived by the Board, that<br>Convertible Security will lapse.  |
| Exercise of Convertible<br>Securities and cashless<br>exercise | To exercise a Convertible Security, the Participant must<br>deliver a signed notice of exercise and, subject to a cashless<br>exercise of Convertible Securities (see below), pay the<br>exercise price (if any) to or as directed by the Company, at<br>any time following vesting of the Convertible Security (if<br>subject to vesting conditions) and prior to the expiry date as<br>set out in the invitation or vesting notice.  |
|  | An invitation may specify that at the time of exercise of the<br>Convertible Securities, the Participant may elect not to be<br>required to provide payment of the exercise price for the<br>number of Convertible Securities specified in a notice of<br>exercise, but that on exercise of those Convertible Securities<br>the Company will transfer or issue to the Participant that<br>number of Shares equal in value to the positive difference<br>between the Market Value of the Shares at the time of<br>exercise and the exercise price that would otherwise be<br>payable to exercise those Convertible Securities.<br>Market Value means, at any given date, the volume<br>weighted average price per Share traded on the ASX over<br>the 5 trading days immediately preceding that given date,<br>unless otherwise specified in an invitation. |
|  | A Convertible Security may not be exercised unless and until<br>that Convertible Security has vested in accordance with the<br>Plan rules, or such earlier date as set out in the Plan rules.  |
| Delivery of Shares on<br>exercise of<br>Convertible Securities | As soon as practicable after the valid exercise of a<br>Convertible Security by a Participant, the Company will issue<br>or cause to be transferred to that Participant the number of<br>Shares to which the Participant is entitled under the Plan rules<br>and issue a substitute certificate for any remaining<br>unexercised Convertible Securities held by that Participant.  |
| Forfeiture of<br>Convertible Securities                        | Where a Participant who holds Convertible Securities ceases<br>to be an Eligible Participant or becomes insolvent, all<br>unvested Convertible Securities will automatically be forfeited<br>by the Participant, unless the Board otherwise determines in its<br>discretion to permit some or all of the Convertible Securities to<br>vest. Where the Board determines that a Participant has<br>acted fraudulently or dishonestly; committed an act which   |

|   | has brought the Company, the Group or any entity within the<br>Group into disrepute, or wilfully breached his or her duties to<br>the Group or where a Participant is convicted of an offence<br>in connection with the affairs of the Group; or has a judgment<br>entered against him or her in any civil proceedings in respect<br>of the contravention by the Participant of his or her duties at<br>law, in equity or under statute, in his or her capacity as an<br>employee, consultant or officer of the Group, the Board may<br>in its discretion deem all unvested Convertible Securities held<br>by that Participant to have been forfeited.   |
|---|--|
|   | Unless the Board otherwise determines, or as otherwise set out<br>in the Plan rules:<br>(a) any Convertible Securities which have not yet vested will<br>be forfeited immediately on the date that the Board<br>determines (acting reasonably and in good faith) that any<br>applicable vesting conditions have not been met or cannot<br>be met by the relevant date; and<br>(b) any Convertible Securities which have not yet vested will<br>be automatically forfeited on the expiry date specified in the<br>invitation or vesting notice.   |
| Change of control                       | If a change of control event occurs in relation to the<br>Company, or the Board determines that such an event is<br>likely to occur, the Board may in its discretion determine the<br>manner in which any or all of the Participant's Convertible<br>Securities will be dealt with, including, without limitation, in a<br>manner that allows the Participant to participate in and/or<br>benefit from any transaction arising from or in connection<br>with the change of control event.  |
| Rights attaching to<br>Plan Shares      | All Shares issued or transferred under the Plan or issued or<br>transferred to a Participant upon the valid exercise of a<br>Convertible Security, (Plan Shares) will rank pari passu in all<br>respects with the Shares of the same class. A Participant will<br>be entitled to any dividends declared and distributed by the<br>Company on the Plan Shares and may participate in any<br>dividend reinvestment plan operated by the Company in<br>respect of Plan Shares. A Participant may exercise any voting<br>rights attaching to Plan Shares.  |
| Disposal restrictions on<br>Plan Shares | If the invitation provides that any Plan Shares are subject to<br>any restrictions as to the disposal or other dealing by a<br>Participant for a period, the Board may implement any<br>procedure it deems appropriate to ensure the compliance<br>by the Participant with this restriction.   |
| Adjustment of<br>Convertible Securities | For so long as a Plan Share is subject to any disposal<br>restrictions under the Plan, the Participant will not:<br>(a) transfer, encumber or otherwise dispose of, or have a<br>security interest granted over that Plan Share; or<br>(b) take any action or permit another person to take any<br>action to remove or circumvent the disposal restrictions<br>without the express written consent of the Company.<br>If there is a reorganisation of the issued share capital of the<br>Company (including any subdivision, consolidation,<br>reduction, return or cancellation of such issued capital of the<br>Company), the rights of each Participant holding Convertible<br>Securities will be changed to the extent necessary to comply |

|                                | with the Listing Rules applicable to a reorganisation of capital at the time of the reorganisation.   |
|--------------------------------|---|
|                                | If Shares are issued by the Company by way of bonus issue<br>(other than an issue in lieu of dividends or by way of dividend<br>reinvestment), the holder of Convertible Securities is entitled,<br>upon exercise of the Convertible Securities, to receive an<br>issue of as many additional Shares as would have been issued<br>to the holder if the holder held Shares equal in number to the<br>Shares in respect of which the Convertible Securities are<br>exercised.   |
|                                | Unless otherwise determined by the Board, a holder of<br>Convertible Securities does not have the right to participate<br>in a pro rata issue of Shares made by the Company or sell<br>renounceable rights.   |
| Participation in new<br>issues | There are no participation rights or entitlements inherent in<br>the Convertible Securities and holders are not entitled to<br>participate in any new issue of Shares of the Company during<br>the currency of the Convertible Securities without exercising<br>the Convertible Securities.   |
| Compliance with applicable law | No Security may be offered, granted, vested or exercised if to do so would contravene any applicable law.   |
| Amendment of Plan              | Subject to the following paragraph, the Board may at any<br>time amend any provisions of the Plan rules, including<br>(without limitation) the terms and conditions upon which any<br>Securities have been granted under the Plan and determine<br>that any amendments to the Plan rules be given retrospective<br>effect, immediate effect or future effect.   |
|                                | No amendment to any provision of the Plan rules may be<br>made if the amendment materially reduces the rights of any<br>Participant as they existed before the date of the<br>amendment, other than an amendment introduced primarily<br>for the purpose of complying with legislation or to correct<br>manifest error or mistake, amongst other things, or is agreed<br>to in writing by all Participants.   |
| Plan duration                  | The Plan continues in operation until the Board decides to<br>end it. The Board may from time to time suspend the<br>operation of the Plan for a fixed period or indefinitely and<br>may end any suspension. If the Plan is terminated or<br>suspended for any reason, that termination or suspension<br>must not prejudice the accrued rights of the Participants.<br>If a Participant and the Company (acting by the Board)<br>agree in writing that some or all of the Securities granted to<br>that Participant are to be cancelled on a specified date or<br>on the occurrence of a particular event, then those Securities<br>may be cancelled in the manner agreed between the<br>Company and the Participant. |

Income Tax Assessment Act The Plan is a plan to which Subdivision 83A-C of the Income Tax Assessment Act 1997 (Cth) applies (subject to the conditions in that Act).

### **SCHEDULE 3 – MATERIAL TERMS OF PURCHASED RIGHTS**

#### a) Entitlement

Each Purchased Rights entitles the holder to subscribe for one Share upon conversion of the Purchased Right.

#### b) Vesting

The Purchased Rights will be fully vested on issue, however will be subject to a 12month Exercise Restriction from the date of grant.

#### c) Conversion Price

There is no amount payable on conversion of each Purchased Right after the lifting of the Exercise Restriction.

#### d) Expiry Date

5 years from the date Purchased Rights are granted.

#### e) Exercise Period

Following the lifting of the Exercise Restriction, the Purchased Rights are exercisable at any time on or prior to the Expiry Date (**Exercise Period**).

#### f) Notice of Exercise

The Purchased Rights may be exercised during the Exercise Period by notice in writing to the Company in the manner specified in the invitation letter (**Notice of Exercise**).

#### g) Exercise Date

A Notice of Exercise is only effective on and from the later of the date of receipt of the Notice of Exercise (**Exercise Date**). If Purchased Rights are not exercised by the Expiry Date, they will be automatically exercised prior to the Expiry Date.

#### h) Required Approvals on Exercise

Notwithstanding any other paragraph in this Schedule, the Purchased Rights may not be converted into Shares in accordance with this Schedule if such exercise would cause the holder or the Company to breach any applicable law, including under the Corporations Act and/or the ASX Listing Rules, and the Company and the holder shall use best endeavours to obtain any required approvals (including Shareholder approvals) to allow exercise of the Purchased Rights.

#### i) Timing of issue/allocation of Shares on exercise

As soon as practicable after the Exercise Date, the Company will:

- Issue, allocate or cause to be transferred the number of Shares required under these terms and conditions in respect of the number of Purchased Rights specified in the Notice of Exercise and for which cleared funds have been received by the Company;
- v. give ASX a notice that complies with section 708A(5)(e) of the Corporations Act, or, if the Company is unable to issue such a notice, lodge with ASIC a prospectus prepared in accordance with the Corporations Act and do all such things necessary to satisfy section 708A(11) of the Corporations Act to ensure that an offer for sale of the Shares does not require disclosure to investors; and
- vi. apply for official quotation on ASX of Shares issued pursuant to the exercise of the Purchased Rights.

#### j) Shares issued/allocated on exercise

Shares issued on exercise of the Purchased Rights rank equally with the then issued shares of the Company.

#### k) Reconstruction of capital

If at any time the issued capital of the Company is reconstructed, all rights of a participant holding Purchased Rights will be changed in a manner consistent with the Corporations Act and the ASX Listing Rules at the time of the reconstruction, subject to the provisions with respect to rounding of entitlements as sanctioned by a meeting of Shareholders approving a reconstruction of capital, in all other respects the terms of exercise of the Purchased Rights will remain unchanged.

#### I) Participation in new issues

There are no participation rights or entitlements inherent in the Purchased Rights and holders will not be entitled to participate in new issues of capital offered to Shareholders during the currency of the Purchased Rights without exercising the Purchased Rights. The Company will, where required pursuant to the ASX Listing Rules, provide Rightsholders with notice prior to the books record date (to determine entitlements to any new issue of Securities made to Shareholders generally) to exercise the Purchased Rights, in accordance with the requirements of the ASX Listing Rules.

#### m) Change in exercise price

Subject to paragraph (k), a Purchased Right does not confer the right to a change in Conversion Price or a change in the number of underlying securities over which the Purchased Right can be exercised.

#### n) Transferability

The Purchased Rights are not transferable and may not be dealt with (except with Board approval or by force of law upon death or bankruptcy).

#### o) Quotation

The Purchased Rights are unquoted and application will not be made to the ASX for official quotation of the Purchased Rights.

# **SCHEDULE 4 – AMENDED CONSTITUTION**

**CORPORATIONS ACT 2001** 

# CONSTITUTION

of

# CYPRIUM METALS LIMITED ACN 002 678 640

Adopted by Special Resolution of the members

# CONTENTS

| 1. | INTERPRE  | TATION  | . 1   |
|----|---|---|---|
|    | 1.1<br>1.2<br>1.3<br>1.4<br>1.5<br>1.6<br>1.7<br>1.8  | Definitions<br>Interpretation<br>Corporations Act Definitions<br>Status of Constitution<br>General Authorisation<br>Displacement of Replaceable Rules<br>Enforceability<br>Jurisdiction   | . 4<br>. 4<br>. 5<br>. 5  |
| 2. | SHARE C   | APITAL AND VARIATION OF RIGHTS  | . 5   |
|    | 2.1<br>2.2<br>2.3<br>2.4<br>2.5<br>2.6<br>2.7<br>2.8<br>2.9<br>2.10<br>2.11<br>2.12<br>2.13<br>2.14<br>2.15                               | Rights Attaching to Shares<br>Issue of Shares.<br>Share Options<br>Variation of class rights.<br>Effect of share issue on class rights<br>Preference Shares.<br>Recognition of Trusts.<br>Unregistered Interests<br>Share Certificates and Share Option Certificates<br>Section 1071H of the Corporations Act.<br>Commissions<br>Restricted Securities.<br>Non-Issue or Cancellation of Certificate.<br>No Prohibition on Foreign Ownership<br>Payment of Interest out of Capital | . 5<br>. 6<br>. 6<br>. 6<br>. 6<br>. 6<br>. 7<br>7<br>. 8<br>8                        |
| 3. |   | M SHAREHOLDING  |   |
|    | 3.1<br>3.2<br>3.3<br>3.4<br>3.5<br>3.6<br>3.7<br>3.8<br>3.7<br>3.8<br>3.9<br>3.10<br>3.11<br>3.12<br>3.13<br>3.14<br>3.15<br>3.16<br>3.17 | Effect of this Clause<br>Definitions<br>Minimum Shareholding<br>Sale of Listed Securities of Minority Member<br>Acceptance of Offer<br>Appointment of Attorney<br>Transfer<br>Proceeds of Sale<br>Receipt of Proceeds<br>Registration of Purchaser<br>Remedies Limited<br>Cost of Sale of Listed Securities<br>Exemption from clause 3<br>Notice to Exempt<br>Takeover Offer or Announcement<br>Use by Company of Clause 3<br>Notice to New Minority Members                      | . 8<br>. 8<br>. 9<br>. 9<br>. 9<br>10<br>10<br>10<br>10<br>10<br>11<br>11<br>11<br>11 |
| 4. | <b>UNCERTI</b><br>4.1<br>4.2  | FICATED HOLDINGS AND ELECTRONIC TRANSFERS<br>Electronic or Computerised Holding<br>Statement of Holdings  | 12  |
|    | 4.2<br>4.3<br>4.4   | Share Certificates<br>Listing Rules   | 12  |
| 5. | LIEN  |   | 13  |
|    | 5.1   | Lien for Members Debts  | 13  |

|    | 5.2   | Generally   |      |
|----|-------|---|------|
|    | 5.3   | Exemption   | . 14 |
|    | 5.4   | Dividends   |      |
|    | 5.5   | Sale of Shares  | . 14 |
|    | 5.6   | Restrictions on Sale  | . 14 |
|    | 5.7   | Person Authorised to Sign Transfers                                   | . 14 |
|    | 5.8   | Proceeds of Sale  | . 15 |
|    | 5.9   | Protection of Lien under ASX Settlement Operating Rules               | . 15 |
|    | 5.10  | Further Powers re Forfeited Shares and Liens                          | . 15 |
| 6. | CALLS | ON SHARES   | . 15 |
|    | 6.1   | Calls   | . 15 |
|    | 6.2   | Payment of Calls  |      |
|    | 6.3   | Quoted Shares   |      |
|    | 6.4   | Unquoted Shares   |      |
|    | 6.5   | Joint Liability   |      |
|    | 6.6   | Deemed Calls  |      |
|    | 6.7   | Differentiation between Shareholders                                  |      |
|    | 6.8   | Payments in Advance of Calls  |      |
|    | 6.9   | Outstanding Moneys  |      |
|    | 6.10  | Revocation/Postponement or Extension                                  |      |
|    | 6.11  | Compliance with Listing Rules and Corporations Act                    |      |
|    | 6.12  | Waive   |      |
| _  |       |   |      |
| 7. |       | TURE OF SHARES  |      |
|    | 7.1   | Failure to Pay Call   |      |
|    | 7.2   | Forfeiture  |      |
|    | 7.3   | Sale of Forfeited Shares  |      |
|    | 7.4   | Continuing Liability  |      |
|    | 7.5   | Officer's Statement Prima Facie Evidence                              |      |
|    | 7.6   | Procedures  |      |
|    | 7.7   | Listing Rules and ASX Settlement Operating Rules                      |      |
|    | 7.8   | Waive   | . 18 |
| 8. | TRANS | FER OF SHARES   | . 19 |
|    | 8.1   | Form of Transfer  | . 19 |
|    | 8.2   | CHESS Transfers   | . 19 |
|    | 8.3   | Participation in CHESS  | . 19 |
|    | 8.4   | Registration Procedure  | . 19 |
|    | 8.5   | Power to Refuse to Register   | . 20 |
|    | 8.6   | Closure of Register   | . 20 |
|    | 8.7   | Retention of Transfers by Company                                     | . 20 |
|    | 8.8   | Power to suspend registration of transfers                            | . 20 |
|    | 8.9   | Powers of Attorney  | . 20 |
|    | 8.10  | Other Securities  | . 20 |
|    | 8.11  | Branch Register   | . 21 |
|    | 8.12  | Compliance with ASX Settlement Operating Rules                        | . 21 |
|    | 8.13  | Issuer Sponsored Subregister  |      |
|    | 8.14  | Transferor Holds Shares until Registration of Transfer                |      |
|    | 8.15  | Waive   |      |
| 9. | TRANS | MISSION OF SHARES   | . 21 |
|    | 9.1   | Death of Shareholder Leaving a Will                                   | . 21 |
|    | 9.2   | Death or Bankruptcy of Shareholder or the Shareholder becomes of unso |      |
|    |       | mind  | . 21 |
|    | 9.3   | Registration by Transmission or to Beneficiary                        |      |
|    | 9.4   | Limitations to Apply  |      |
|    | 9.5   | Death of a Joint Holder   | . 22 |

|     | 9.6    | Joint Personal Representatives   |      |
|-----|--------|--|------|
|     | 9.7    | ASX Settlement Transfer  |      |
|     | 9.8    | Joint Holders  | . 22 |
| 10. | CHAN   | GES TO CAPITAL STRUCTURE   | 23   |
| 10. |        |  |      |
|     | 10.1   | Alterations to Capital   |      |
|     | 10.2   | Reduction of Capital   |      |
|     | 10.3   | Buy-Backs  |      |
|     | 10.4   | Fractions  | . 24 |
| 11. | WRITTE | N RESOLUTIONS  | . 24 |
| 12. | GENER  | AL MEETINGS  | . 24 |
|     | 12.1   | Convening of General Meetings of Shareholders by Directors' Resolution       | . 24 |
|     | 12.2   | Change of place or postponement of a General Meeting of Shareholders         | . 24 |
|     | 12.3   | Convening of General Meetings of Shareholders by a Director or requisition . |      |
|     | 12.4   | Cancellation of a General Meeting of Shareholders                            |      |
|     | 12.5   | Notice   |      |
|     | 12.6   | Irregularities in giving notice  |      |
|     | 12.7   | Business at General Meeting  |      |
|     | 12.8   | Notice to Home Branch  |      |
|     | 12.9   | Annual General Meeting   |      |
|     |        |  |      |
| 13. | PROCE  | EDINGS AT GENERAL MEETINGS   | . 26 |
|     | 13.1   | Quorum   | . 26 |
|     | 13.2   | Persons Entitled to Attend a General Meeting                                 | . 26 |
|     | 13.3   | Refusal of Admission to Meetings   |      |
|     | 13.4   | Insufficient room  |      |
|     | 13.5   | Chairman   |      |
|     | 13.6   | Vacating Chair   |      |
|     | 13.7   | Disputes Concerning Procedure  |      |
|     | 13.8   | General Conduct  |      |
|     | 13.9   | Adjournment  |      |
|     | 13.10  | Notice of Resumption of Adjourned Meeting                                    |      |
|     | 13.11  | How resolutions are decided  |      |
|     | 13.12  | Casting Vote   |      |
|     |        | •  |      |
|     | 13.13  | Voting Rights  |      |
|     | 13.14  | Voting - Show of Hands   | . 29 |
|     | 13.15  | Results of Voting  |      |
|     | 13.16  | Poll   |      |
|     | 13.17  | Manner of Taking Poll  |      |
|     | 13.18  | Meeting May Continue   |      |
|     | 13.19  | Voting by Joint Holders  |      |
|     | 13.20  | Shareholder under Disability   |      |
|     | 13.21  | Payment of Calls   | . 30 |
|     | 13.22  | Objection to Voting  | . 30 |
|     | 13.23  | Restrictions on voting   | . 30 |
|     | 13.24  | Proxies  | . 30 |
|     | 13.25  | Electronic Appointment of Proxy  | . 31 |
|     | 13.26  | Name of proxy  |      |
|     | 13.27  | Incomplete proxy appointment   |      |
|     | 13.28  | No right to speak or vote if appointing Shareholder present                  |      |
|     | 13.29  | Rights where 2 proxies or attorneys are appointed                            |      |
|     | 13.30  | More than 2 proxies or attorneys appointed                                   |      |
|     | 13.31  | Proxy Votes  |      |
|     | 13.31  | Representatives of Corporate Shareholders                                    |      |
|     |        |  |      |
|     | 13.33  | More than one Representative present   |      |
|     | 13.34  | Rights of Representatives, proxies and attorneys                             | . 33 |

|      | 13.35  | Board may determine Direct Voting to apply   |  |
|------|--|--|--|
|      | 13.36  | Direct Voting instrument – form, signature and deposit   | 34   |
|      | 13.37  | Voting Forms   | 35   |
|      | 13.38  | Direct Votes count on a poll   |  |
|      | 13.39  | Withdrawal of a Direct Vote  |  |
|      | 13.40  | Validity of Direct Vote  | 36   |
| 13A. | USE OF   | TECHNOLOGY AT GENERAL MEETINGS   | 36   |
|      | 13A.1  | Use of technology  | 36   |
|      | 13A.2  | Communication of meeting documents   | 37   |
| 14.  | THE DIR  | ECTORS   | 37   |
|      | 14.1   | Number of Directors  |  |
|      | 14.2   | Rotation of Directors  |  |
|      | 14.3   | Election of Directors  |  |
|      | 14.4   | Additional Directors   |  |
|      | 14.5   | Removal of Director  |  |
|      | 14.6   | Vacation of Office   |  |
|      | 14.7   | Remuneration   |  |
|      | 14.8   | Initial Fees to Directors  |  |
|      | 14.9   | Expenses   |  |
|      | 14.10  | No Share Qualification   |  |
| 15.  | POWER  | S AND DUTIES OF DIRECTORS  |  |
|      | 15.1   | Management of the Company  |  |
|      | 15.2   | Borrowings   |  |
|      | 15.3   | Attorneys  |  |
|      | 15.4   | Cheques, etc   |  |
|      | 15.5   | Retirement Benefits for Directors  |  |
|      |  |  | 11   |
|      | 15.6   | Securities to Directors or Shareholders  | 41   |
| 16.  |  | Securities to Directors or Shareholders  |  |
| 16.  | <b>PROCEI</b><br>16.1  | EDINGS OF DIRECTORS<br>Convening a Meeting   | <b>41</b><br>41  |
| 16.  | <b>PROCEI</b><br>16.1<br>16.2  | EDINGS OF DIRECTORS<br>Convening a Meeting<br>Procedure at Meetings  | <b> 41</b><br>41<br>42   |
| 16.  | <b>PROCEI</b><br>16.1<br>16.2<br>16.3  | EDINGS OF DIRECTORS<br>Convening a Meeting<br>Procedure at Meetings<br>Quorum  | <b>41</b><br>41<br>42<br>42  |
| 16.  | <b>PROCEI</b><br>16.1<br>16.2<br>16.3<br>16.4  | EDINGS OF DIRECTORS<br>Convening a Meeting<br>Procedure at Meetings<br>Quorum<br>Secretary May Attend and Be Heard   | <b>41</b><br>41<br>42<br>42<br>42  |
| 16.  | <b>PROCEI</b><br>16.1<br>16.2<br>16.3<br>16.4<br>16.5  | EDINGS OF DIRECTORS<br>Convening a Meeting<br>Procedure at Meetings<br>Quorum<br>Secretary May Attend and Be Heard<br>Majority Decisions   | <b>41</b><br>41<br>42<br>42<br>42<br>42  |
| 16.  | <b>PROCEI</b><br>16.1<br>16.2<br>16.3<br>16.4<br>16.5<br>16.6  | EDINGS OF DIRECTORS<br>Convening a Meeting<br>Procedure at Meetings<br>Quorum<br>Secretary May Attend and Be Heard<br>Majority Decisions<br>Casting Votes  | <b>41</b><br>42<br>42<br>42<br>42<br>42<br>42  |
| 16.  | <b>PROCEI</b><br>16.1<br>16.2<br>16.3<br>16.4<br>16.5<br>16.6<br>16.7  | EDINGS OF DIRECTORS<br>Convening a Meeting<br>Procedure at Meetings<br>Quorum<br>Secretary May Attend and Be Heard<br>Majority Decisions<br>Casting Votes<br>Alternate Directors   | <b>41</b><br>41<br>42<br>42<br>42<br>42<br>42<br>42  |
| 16.  | <b>PROCEI</b><br>16.1<br>16.2<br>16.3<br>16.4<br>16.5<br>16.6<br>16.7<br>16.8  | EDINGS OF DIRECTORS<br>Convening a Meeting<br>Procedure at Meetings<br>Quorum<br>Secretary May Attend and Be Heard<br>Majority Decisions<br>Casting Votes<br>Alternate Directors<br>Continuing Directors May Act   | <b>41</b><br>42<br>42<br>42<br>42<br>42<br>42<br>42<br>43  |
| 16.  | PROCEI<br>16.1<br>16.2<br>16.3<br>16.4<br>16.5<br>16.6<br>16.7<br>16.8<br>16.9   | EDINGS OF DIRECTORS<br>Convening a Meeting<br>Procedure at Meetings<br>Quorum<br>Secretary May Attend and Be Heard<br>Majority Decisions<br>Casting Votes<br>Alternate Directors<br>Continuing Directors May Act<br>Chairman   | <b>41</b><br>42<br>42<br>42<br>42<br>42<br>42<br>43<br>43  |
| 16.  | PROCEI<br>16.1<br>16.2<br>16.3<br>16.4<br>16.5<br>16.6<br>16.7<br>16.8<br>16.9<br>16.10  | EDINGS OF DIRECTORS<br>Convening a Meeting<br>Procedure at Meetings<br>Quorum<br>Secretary May Attend and Be Heard<br>Majority Decisions<br>Casting Votes<br>Alternate Directors<br>Continuing Directors May Act<br>Chairman<br>Committees   | <b>41</b><br>42<br>42<br>42<br>42<br>42<br>42<br>42<br>43<br>43<br>43  |
| 16.  | PROCEI<br>16.1<br>16.2<br>16.3<br>16.4<br>16.5<br>16.6<br>16.7<br>16.8<br>16.9<br>16.10<br>16.11   | EDINGS OF DIRECTORS<br>Convening a Meeting<br>Procedure at Meetings<br>Quorum<br>Secretary May Attend and Be Heard<br>Majority Decisions<br>Casting Votes<br>Alternate Directors<br>Continuing Directors May Act<br>Chairman<br>Committees<br>Written Resolutions  | <b>41</b><br>42<br>42<br>42<br>42<br>42<br>42<br>42<br>43<br>43<br>43<br>44  |
| 16.  | PROCES<br>16.1<br>16.2<br>16.3<br>16.4<br>16.5<br>16.6<br>16.7<br>16.8<br>16.9<br>16.10<br>16.11<br>16.12  | EDINGS OF DIRECTORS<br>Convening a Meeting<br>Procedure at Meetings<br>Quorum<br>Secretary May Attend and Be Heard<br>Majority Decisions<br>Casting Votes<br>Alternate Directors<br>Continuing Directors May Act<br>Chairman<br>Committees<br>Written Resolutions<br>Defective Appointment   | <b>41</b><br>42<br>42<br>42<br>42<br>42<br>42<br>42<br>43<br>43<br>43<br>44  |
| 16.  | PROCES<br>16.1<br>16.2<br>16.3<br>16.4<br>16.5<br>16.6<br>16.7<br>16.8<br>16.9<br>16.10<br>16.11<br>16.12<br>16.13   | EDINGS OF DIRECTORS<br>Convening a Meeting<br>Procedure at Meetings<br>Quorum<br>Secretary May Attend and Be Heard<br>Majority Decisions<br>Casting Votes<br>Alternate Directors<br>Continuing Directors May Act<br>Chairman<br>Committees<br>Written Resolutions<br>Defective Appointment<br>Directors May Hold Other Offices   | <b>41</b><br>42<br>42<br>42<br>42<br>42<br>42<br>42<br>42<br>42<br>43<br>43<br>43<br>44<br>44  |
| 16.  | PROCEI<br>16.1<br>16.2<br>16.3<br>16.4<br>16.5<br>16.6<br>16.7<br>16.8<br>16.7<br>16.8<br>16.9<br>16.10<br>16.11<br>16.12<br>16.13<br>16.14  | EDINGS OF DIRECTORS<br>Convening a Meeting<br>Procedure at Meetings<br>Quorum<br>Secretary May Attend and Be Heard<br>Majority Decisions<br>Casting Votes<br>Alternate Directors May Act<br>Continuing Directors May Act<br>Chairman<br>Committees<br>Written Resolutions<br>Defective Appointment<br>Directors May Hold Other Offices<br>Directors May Hold Shares, etc.  | <b>41</b><br>42<br>42<br>42<br>42<br>42<br>42<br>42<br>42<br>43<br>43<br>43<br>44<br>44<br>44  |
| 16.  | PROCEI<br>16.1<br>16.2<br>16.3<br>16.4<br>16.5<br>16.6<br>16.7<br>16.8<br>16.9<br>16.10<br>16.11<br>16.12<br>16.13<br>16.14<br>16.15   | EDINGS OF DIRECTORS  | <b>41</b><br>41<br>42<br>42<br>42<br>42<br>42<br>42<br>43<br>43<br>43<br>44<br>44<br>44<br>44  |
| 16.  | PROCEI<br>16.1<br>16.2<br>16.3<br>16.4<br>16.5<br>16.6<br>16.7<br>16.8<br>16.9<br>16.10<br>16.11<br>16.12<br>16.13<br>16.14<br>16.15<br>16.16  | EDINGS OF DIRECTORS<br>Convening a Meeting<br>Procedure at Meetings<br>Quorum<br>Secretary May Attend and Be Heard<br>Majority Decisions<br>Casting Votes<br>Alternate Directors<br>Continuing Directors May Act<br>Chairman<br>Committees<br>Written Resolutions<br>Defective Appointment<br>Directors May Hold Other Offices<br>Directors May Hold Shares, etc<br>Directors Not Accountable for Benefits<br>Disclosure of Interests in Related Matters   | <b>41</b><br>41<br>42<br>42<br>42<br>42<br>42<br>42<br>42<br>43<br>43<br>43<br>43<br>44<br>44<br>44<br>44  |
| 16.  | PROCEI<br>16.1<br>16.2<br>16.3<br>16.4<br>16.5<br>16.6<br>16.7<br>16.8<br>16.9<br>16.10<br>16.11<br>16.12<br>16.13<br>16.14<br>16.15<br>16.16<br>16.17   | EDINGS OF DIRECTORS<br>Convening a Meeting<br>Procedure at Meetings<br>Quorum<br>Secretary May Attend and Be Heard<br>Majority Decisions<br>Casting Votes<br>Alternate Directors<br>Continuing Directors May Act<br>Chairman<br>Committees<br>Written Resolutions<br>Defective Appointment<br>Directors May Hold Other Offices<br>Directors May Hold Shares, etc<br>Directors Not Accountable for Benefits<br>Disclosure of Interests in Related Matters<br>Disclosure of Shareholding   | <b>41</b><br>42<br>42<br>42<br>42<br>42<br>42<br>42<br>43<br>43<br>43<br>43<br>44<br>44<br>44<br>44<br>44<br>44<br>45                              |
| 16.  | PROCES   | EDINGS OF DIRECTORS<br>Convening a Meeting<br>Procedure at Meetings<br>Quorum<br>Secretary May Attend and Be Heard<br>Majority Decisions<br>Casting Votes<br>Alternate Directors<br>Continuing Directors May Act<br>Chairman<br>Committees<br>Written Resolutions<br>Defective Appointment<br>Directors May Hold Other Offices<br>Directors May Hold Shares, etc.<br>Directors Not Accountable for Benefits<br>Disclosure of Interests in Related Matters<br>Disclosure of Shareholding<br>Related Body Corporate Contracts                              | <b>41</b><br>41<br>42<br>42<br>42<br>42<br>42<br>42<br>43<br>43<br>43<br>43<br>44<br>44<br>44<br>44<br>45<br>45                                    |
| 16.  | PROCEI<br>16.1<br>16.2<br>16.3<br>16.4<br>16.5<br>16.6<br>16.7<br>16.8<br>16.9<br>16.10<br>16.11<br>16.12<br>16.13<br>16.14<br>16.15<br>16.16<br>16.17   | EDINGS OF DIRECTORS  | <b>41</b><br>42<br>42<br>42<br>42<br>42<br>42<br>42<br>42<br>42<br>42<br>43<br>43<br>43<br>43<br>44<br>44<br>44<br>44<br>44<br>45<br>45<br>45      |
| 16.  | PROCEI<br>16.1<br>16.2<br>16.3<br>16.4<br>16.5<br>16.6<br>16.7<br>16.8<br>16.7<br>16.8<br>16.9<br>16.10<br>16.11<br>16.12<br>16.13<br>16.14<br>16.15<br>16.16<br>16.17<br>16.18<br>16.19<br>16.20            | EDINGS OF DIRECTORS<br>Convening a Meeting<br>Procedure at Meetings<br>Quorum<br>Secretary May Attend and Be Heard<br>Majority Decisions<br>Casting Votes<br>Alternate Directors<br>Continuing Directors May Act<br>Chairman<br>Committees<br>Written Resolutions<br>Defective Appointment<br>Directors May Hold Other Offices<br>Directors May Hold Shares, etc<br>Directors Not Accountable for Benefits<br>Disclosure of Interests in Related Matters<br>Disclosure of Shareholding<br>Related Body Corporate Contracts<br>Voting, Affixation of Seal | <b>41</b><br>42<br>42<br>42<br>42<br>42<br>42<br>43<br>43<br>43<br>43<br>43<br>44<br>44<br>44<br>44<br>45<br>45<br>45                              |
|      | PROCEI<br>16.1<br>16.2<br>16.3<br>16.4<br>16.5<br>16.6<br>16.7<br>16.8<br>16.7<br>16.8<br>16.9<br>16.10<br>16.11<br>16.12<br>16.13<br>16.14<br>16.15<br>16.16<br>16.17<br>16.18<br>16.19<br>16.20            | EDINGS OF DIRECTORS<br>Convening a Meeting   | 41<br>41<br>42<br>42<br>42<br>42<br>42<br>42<br>42<br>43<br>43<br>43<br>43<br>43<br>43<br>44<br>44<br>44<br>45<br>45<br>45<br>45                   |
|      | PROCES<br>16.1<br>16.2<br>16.3<br>16.4<br>16.5<br>16.6<br>16.7<br>16.8<br>16.7<br>16.8<br>16.9<br>16.10<br>16.11<br>16.12<br>16.13<br>16.14<br>16.15<br>16.16<br>16.17<br>16.18<br>16.19<br>16.20<br>MEETING | EDINGS OF DIRECTORS  | 41<br>41<br>42<br>42<br>42<br>42<br>42<br>42<br>42<br>43<br>43<br>43<br>43<br>43<br>43<br>44<br>44<br>44<br>45<br>45<br>45                         |
|      | PROCES<br>16.1<br>16.2<br>16.3<br>16.4<br>16.5<br>16.6<br>16.7<br>16.8<br>16.7<br>16.8<br>16.9<br>16.10<br>16.11<br>16.12<br>16.13<br>16.14<br>16.15<br>16.16<br>16.17<br>16.18<br>16.19<br>16.20<br>MEETING | EDINGS OF DIRECTORS<br>Convening a Meeting   | 41<br>41<br>42<br>42<br>42<br>42<br>42<br>42<br>42<br>42<br>43<br>43<br>43<br>43<br>43<br>43<br>44<br>44<br>44<br>45<br>45<br>45<br>45<br>45<br>46 |

| 18. | MANAG  | GING AND EXECUTIVE DIRECTORS AND SECRETARIES  | 46                               |
|-----|--|---|----------------------------------|
|     | 18.1<br>18.2<br>18.3<br>18.4<br>18.5                                 | Appointment<br>Remuneration<br>Powers<br>Rotation<br>Secretary  | 47<br>47<br>47                   |
| 19. | SEALS  |   | 47                               |
|     | 19.1<br>19.2<br>19.3   | Common Seal<br>Execution of Documents Without a Seal<br>Share Seal  | 47                               |
| 20. | ACCOU  | INTS, AUDIT AND RECORDS   | 48                               |
|     | 20.1<br>20.2<br>20.3   | Accounting records to be kept<br>Audit<br>Inspection  | 48                               |
| 21. | MINUTE   | S   | 49                               |
|     | 21.1<br>21.2<br>21.3   | Minutes to be Kept<br>Signature of Minutes<br>Requirements of the Corporations Act  | 49                               |
| 22. | DIVIDEN  | IDS AND RESERVES  | 49                               |
|     | 22.1<br>22.2<br>22.3<br>22.4<br>22.5<br>22.6<br>22.7<br>22.8<br>22.9 | Dividends<br>Interim Dividend<br>No Interest<br>Reserves<br>Carrying forward profits<br>Alternative Method of Payment of Dividend<br>Shareholders entitled to dividend<br>Payment of Dividends<br>Unclaimed Dividends | 49<br>49<br>50<br>50<br>50<br>50 |
|     | 22.10  | Breach of Restriction Agreement   |                                  |
| 23. | CAPITA   | LISATION OF PROFITS   | 51                               |
|     | 23.1<br>23.2<br>23.3   | Capitalisation<br>Application of Capitalised Amounts<br>Procedures  | 51                               |
| 24. | BONUS  | SHARE PLAN  | 52                               |
|     | 24.1<br>24.2   | Authorisation of Bonus Share Plan<br>Amendment and Revocation   |                                  |
| 25. | DIVIDEN  | ID REINVESTMENT PLAN  |                                  |
|     | 25.1<br>25.2   | Authorisation of Dividend Reinvestment Plan<br>Amendment and Revocation   |                                  |
| 26. | NOTICE   | S   |                                  |
|     | 26.1<br>26.2<br>26.3   | Service by the Company to Shareholders<br>Service of notices by the Company to Directors<br>Service of notices by Directors, Alternate Directors and Shareholders to<br>Company                                       | 53<br>he                         |
|     | 26.4<br>26.5<br>26.6<br>26.7<br>26.8                                 | Deemed receipt of Notice<br>Notice to Joint Holders<br>Notices to Personal Representatives and Others<br>Persons Entitled to Notice<br>Change of Address  | 53<br>54<br>54<br>54             |

|        | 26.9                                 | Incorrect Address  | 55             |
|--------|--------------------------------------|--|----------------|
| 27.    | WINDING                              | G UP   | 55             |
|        | 27.1<br>27.2<br>27.3                 | Distribution in Kind<br>Trust for Shareholders<br>Distribution in Proportion to Shares Held        | 55             |
| 28.    | INDEMN                               | ITIES AND INSURANCE  | 55             |
|        | 28.1<br>28.2<br>28.3<br>28.4<br>28.5 | Liability to Third Parties<br>Defending Proceedings<br>Insurance<br>Disclosure<br>Definition       | 56<br>56<br>57 |
| 29.    | DIRECTO                              | DRS' ACCESS TO INFORMATION   | 57             |
| 30.    | OVERSE                               | AS SHAREHOLDERS  | 57             |
| 31.    |                                      | MANAGEMENT   | 57             |
|        | 31.1<br>31.2<br>31.3<br>31.4         | Local Management<br>Local Boards or Agencies<br>Appointment of Attorneys<br>Authority of Attorneys | 58<br>58       |
| 32.    | DISCOV                               | ERY  | 58             |
| 33.    | COMPLI                               | ANCE (OR INCONSISTENCY) WITH THE LISTING RULES   | 58             |
| 34.    | CONSIST                              | IENCY WITH CHAPTER 2E OF THE CORPORATIONS ACT  | 59             |
|        | 34.1<br>34.2                         | Requirements of Chapter 2E<br>Definitions  |                |
| 35.    | INADVE                               | RTENT OMISSIONS  | 60             |
| 36.    | TRANSIT                              | IONAL  | 60             |
|        | 36.1<br>36.2                         | Provisions Relating to Official Quotation of Securities<br>Severance                               |                |
| SCHEDU | LE 1 – PRE                           | FERENCE SHARES (CLAUSE 2.6)  | 61             |

#### **CORPORATIONS ACT**

#### CONSTITUTION

of

#### CYRPIUM METALS LIMITED ACN 002 678 640

#### 1. INTERPRETATION

#### 1.1 Definitions

In this Constitution:

Alternate Director means a person appointed as an alternate director under clause 16.7.

**ASIC** means Australian Securities and Investments Commission.

**ASX** means ASX Limited (ACN 008 624 691) or the Australian Securities Exchange, as the context requires.

ASX Settlement means ASX Settlement Pty Ltd (ACN 008 504 532).

**ASX Settlement Operating Rules** means the operating rules of ASX Settlement.

ASX Settlement Transfer means a transfer of quoted securities or quoted rights effected in:

- (a) accordance with the ASX Settlement Operating Rules; or
- (b) substantial accordance with the ASX Settlement Operating Rules and determined by ASX Settlement to be an effective transfer.

Auditor means the Company's auditor.

Bonus Share Plan means a plan implemented under clause 24.

**Business Day** means a day other than a Saturday, a Sunday, New Year's Day, Australia Day, Good Friday, Easter Monday, Anzac Day, Christmas Day, Boxing Day and any other day declared and published by ASX to be a day which is not a business day.

**CHESS Approved Securities** means securities of the Company for which CHESS approval has been given in accordance with the ASX Settlement Operating Rules, or such amended definition as may be prescribed by the Listing Rules from time to time.

**CHESS System** means the Clearing House Electronic Subregister System operated by ASX Settlement or such other securities clearing house as is approved pursuant to the Corporations Act and to which the Listing Rules apply.

**Company** means Cyprium Metals Limited (ACN 002 678 640) or as it is from time to time named in accordance with the Corporations Act of this jurisdiction.

Constitution means this Constitution as altered or amended from time to time.

Corporations Act means the Corporations Act 2001 (Cth).

Corporations Regulations means the Corporations Regulations 2001 (Cth).

**Director** means a person appointed to the position of a director of the Company and where appropriate, includes an Alternate Director.

Directors means all or some of the Directors acting as a board.

**Direct Vote** means a notice of a Shareholder's voting intention delivered to the Company by post, fax, electronic or other means approved by the Board and otherwise in accordance with this Constitution and regulations, rules and procedures made by the Board in accordance with clause 13.35.

**Dispose** has the meaning ascribed to it by the Listing Rules.

Dividend Reinvestment Plan means a plan implemented under clause 25.

**Duty** means any transfer, transaction or registration duty or similar charge imposed by any Government Authority and includes any interest, fine, penalty, charge or other amount imposed in respect of any of them.

**Government Authority** means a government or government department, a governmental or semi-governmental or judicial person (whether autonomous or not) charged with the administration of any applicable law.

Holding Lock has the meaning ascribed to it by the Listing Rules.

**Home Branch** means the state branch of ASX designated as such in relation to the Company by ASX.

**Listed Securities** means any Shares, Share Options, stock, debentures, debenture stock or other securities for the time being issued by the Company and officially quoted by ASX on its stock market.

**Listing Rules** means the listing rules of ASX and any other rules of ASX which are applicable while the Company is admitted to the official list of ASX, each as amended or replaced from time to time, except to the extent of any express written waiver by ASX.

#### Loan Securities includes:

- (a) unsecured notes or unsecured deposit notes;
- (c) mortgage debentures or mortgage debenture stock;
- (d) debentures or debenture stock; and
- (e) for the purposes of the Listing Rules, convertible loan securities.

Office means the registered office of the Company.

**Officer** means any Director or Secretary of the Company or such other person within the meaning of that term as defined by the Corporations Act.

Official List means the Official List of the ASX.

**Prescribed Rate** means the interest rate which is 2% above the Reserve Bank of Australia cash rate as published or quoted from time to time, or such other rate as may from time to time be fixed by the Directors, calculated daily.

**Registered Office** means the registered office of the Company in the State.

**Register of Shareholders** means the register of Shareholders kept by the Company in accordance with section 169 of the Corporations Act (including any branch register and any computerised or electronic subregister established and administered under the ASX Settlement Operating Rules).

**Related Body Corporate** means a corporation which by virtue of the provisions of section 50 of the Corporations Act is deemed to be related to the relevant corporation and **related** has a corresponding meaning.

**Representative** means a person authorised to act as a representative of a corporation under clause 13.32.

**Restricted Securities** has the meaning ascribed to it by the Listing Rules.

**Restriction Deed** has the meaning ascribed to it by the Listing Rules.

**Seal** means the common seal of the Company and includes any official seal and, where the context so admits, the Share Seal of the Company.

**Secretary** means any person appointed to perform the duties of a secretary of the Company.

Securities has the meaning ascribed to it by the Listing Rules.

Share means a share in the capital of the Company.

**Shareholder** means a person or company registered in the Register of Shareholders as the holder of one or more Shares and includes any person or company who is a member of the Company in accordance with or for the purposes of the Corporations Act.

**Shareholding Account** means an entry in the Register of Shareholders in respect of a Shareholder for the purpose of providing a separate identification of some or all of the ordinary Shares registered from time to time in the name of that Shareholder and **Securities Account** has an equivalent meaning in relation to Listed Securities of all kinds, including ordinary Shares.

Share Option means an option to require the Company to allot and issue a Share.

Share Seal means the duplicate common seal referred to in clause 19.3.

State means Western Australia.

#### 1.2 Interpretation

- (a) A reference in this Constitution to a partly paid share is a reference to a share on which there is an amount unpaid.
- (b) A reference in this Constitution to an amount unpaid on a share includes a reference to any amount of the issue price which is unpaid.
- (c) Unless the contrary intention appears, in this Constitution:
  - (i) the singular includes the plural and the plural includes the singular;
  - (ii) words that refer to any gender include all genders;
  - (iii) words used to refer to persons generally or to refer to a natural person include a body corporate, body politic, partnership, joint venture, association, board, group or other body (whether or not the body is incorporated);
  - (iv) a reference to a person includes that person's successors and legal personal representatives;
  - (v) a reference to a statute or regulation, or a provision of any of them includes all statutes, regulations or provisions amending, consolidating or replacing them, and a reference to a statute includes all regulations, proclamations, ordinances and by-laws issued under that statute;
  - (vi) a reference to the Listing Rules or the ASX Settlement Operating Rules includes any variation, consolidation or replacement of those rules and is to be taken to be subject to any applicable waiver or exemption; and
  - (vii) a reference to writing includes any method of reproducing words in a visible form.
- (d) In this Constitution, headings and body type are only for convenience and do not affect the meaning of this Constitution.

#### 1.3 Corporations Act Definitions

Any word or expression defined in or for the purposes of the Corporations Act shall, unless otherwise defined in clause 1.1 or the context otherwise requires, have the same meaning when used in this Constitution, and the rules of interpretation specified in or otherwise applicable to the Corporations Act shall, unless the context otherwise requires, apply in the interpretation of this Constitution.

#### 1.4 Status of Constitution

This Constitution is adopted by the Company in substitution for any former memorandum and articles of association or other consistent documents of the Company. To the extent permitted by law, the replaceable rules provided for in the Corporations Act do not apply to the Company.

#### 1.5 General Authorisation

Where the Corporations Act authorises or permits a company to do anything, if so authorised by its constitution, the Company is authorised by this rule to do that thing.

#### 1.6 Displacement of Replaceable Rules

The provisions of the Corporations Act that apply to public companies as replaceable rules are displaced completely by this Constitution in relation to the Company except to the extent they are repeated in this Constitution.

#### 1.7 Enforceability

If any provision of this Constitution is or becomes illegal, invalid or unenforceable in any jurisdiction then that illegality, invalidity or unenforceability does not affect the legality, validity or enforceability in that jurisdiction of any other provision of this Constitution or the legality, validity or enforceability in any other jurisdiction of that provision or any other provision of this Constitution.

#### 1.8 Jurisdiction

The courts having jurisdiction in the state of Western Australia have non-exclusive jurisdiction to settle any dispute arising out of or in connection with this Constitution and each Shareholder irrevocably submits to the jurisdiction of those courts.

#### 2. SHARE CAPITAL AND VARIATION OF RIGHTS

#### 2.1 Rights Attaching to Shares

Subject to this Constitution and to the terms of issue of Shares, all Shares attract the right to receive notice of and to attend and vote at all general meetings of the Company, the right to receive dividends, in a winding up or a reduction of capital, the right to participate equally in the distribution of the assets of the Company (both capital and surplus), subject to any amounts unpaid on the Share and, in the case of a reduction, to the terms of the reduction.

#### 2.2 Issue of Shares

Without prejudice to any special rights previously conferred on the holders of any existing Shares or class of Shares, unissued Shares shall be under the control of the Directors and, subject to the Corporations Act, the Listing Rules and this Constitution, the Directors may at any time issue such number of Shares either as ordinary Shares or Shares of a named class or classes (being either an existing class or a new class) at the issue price that the Directors determine and with such preferred, deferred, or other special rights or such restrictions, whether with regard to dividend, voting, return of capital or otherwise, as the Directors shall, in their absolute discretion, determine.

#### 2.3 Share Options

Subject to the Listing Rules, the Directors may at any time and from time to time issue Share Options on such terms and conditions as the Directors shall, in their absolute discretion, determine.

#### 2.4 Variation of class rights

If at any time the share capital of the Company is divided into different classes of Shares, the rights attached to any class (unless otherwise provided by the terms of issue of the Shares of that class) may be varied, whether or not the Company is being wound up, with the consent in writing of the holders of three quarters of the issued Shares of that class, or if authorised by a special resolution passed at a separate meeting of the holders of the Shares of the class. Any variation of rights under this clause 2.4 shall be subject to Part 2F.2 of Chapter 2F of the Corporations Act. The provisions of this Constitution relating to general meetings shall apply so far as they are capable of application and with necessary alterations to every such separate meeting except that a quorum is constituted by two persons who together hold or represent by proxy not less than one-third of the issued Shares of the class.

#### 2.5 Effect of share issue on class rights

The rights attached to any class of shares are not taken to be varied by the issue or creation of further shares ranking equally with them unless expressly provided by the terms of issue of the shares of that class.

#### 2.6 Preference Shares

Subject to the Listing Rules and the Corporations Act, the Company may issue Preference Shares:

- (a) that are liable to be redeemed whether at the option of the Company or otherwise; and
- (b) including, without limitation preference shares of the kind described in clause 2.6(a) in accordance with the terms of Schedule 1.

#### 2.7 Recognition of Trusts

Except as permitted or required by the Corporations Act, the Company shall not recognise a person as holding a Share or Share Option upon any trust.

#### 2.8 Unregistered Interests

The Company is not bound by or compelled in any way to recognise any equitable, contingent, future or partial right or interest in any Share or Share Option (whether or not it has notice of the interest or right concerned) unless otherwise provided by this Constitution or by law, except an absolute right of ownership in the registered holder of the Share or Share Option.

#### 2.9 Share Certificates and Share Option Certificates

Subject to the ASX Settlement Operating Rules (if applicable), clause 4 and the Listing Rules, a person whose name is entered as a Shareholder in the Register of Shareholders is entitled without payment to receive a Share certificate or notice (as the case may be) in respect of the Share under the Seal in accordance with the Corporations Act but, in respect of a Share or Shares held jointly by several persons, the Company is not bound to issue more than one certificate or notice. Delivery of a certificate or notice for a Share to one of several joint Shareholders is sufficient delivery to all such holders. In addition:

- (a) Share certificates or notices in respect of Shares shall only be issued in accordance with the Listing Rules;
- (b) subject to this Constitution, the Company shall dispatch all appropriate Share certificates within one month from the date of issue of any of its Shares and within one month after the date upon which a transfer of any of its Shares is lodged with the Company;
- (c) where a Share certificate is lost, worn out or destroyed, the Company shall issue a duplicate certificate in accordance with the requirements of section 1070D of the Corporations Act and the Listing Rules; and
- (d) the above provisions of this clause 2.9 shall, with necessary alterations, apply to Share Options.

If securities of the Company are CHESS Approved Securities and held in uncertificated mode, then the preceding provisions of this clause 2.9 do not apply to those CHESS Approved Securities and the Company shall allot such CHESS Approved Securities and enter those CHESS Approved Securities into the Shareholder's uncertificated holding in accordance with the Listing Rules and the ASX Settlement Operating Rules.

#### 2.10 Section 1071H of the Corporations Act

Clause 2.9 shall not apply if and to the extent that, on an application by or on behalf of the Company, the ASIC has made a declaration under section 1071H(5) of the Corporations Act published in the Commonwealth of Australia Gazette that the Company is a person in relation to whom section 1071H of the Corporations Act does not apply.

#### 2.11 Commissions

The Company may, subject to the Listing Rules, exercise the powers of paying commission conferred by section 258C of the Corporations Act. Such commission may be satisfied by the payment of cash or the allotment of fully or partly paid Shares or partly in the one way and partly in the other. The Company may also on any issue of Shares pay such brokerage as may be lawful.

#### 2.12 Restricted Securities

The Company shall comply in all respects with the requirements of the Listing Rules with respect to Restricted Securities. Without limiting the generality of the above:

- (a) a holder of Restricted Securities must not Dispose of, or agree or offer to Dispose of, the Securities during the escrow period applicable to those Securities except as permitted by the Listing Rules or the ASX;
- (b) if the Restricted Securities are in the same class as quoted Securities, the holder will be taken to have agreed in writing that the Restricted Securities are to be kept on the Company's issuer sponsored subregister and are to have a Holding Lock applied for the duration of the escrow period applicable to those Securities;
- (c) the Company will refuse to acknowledge any Disposal (including, without limitation, to register any transfer) of Restricted Securities during the escrow

period applicable to those Securities except as permitted by the Listing Rules or the ASX;

- (d) a holder of Restricted Securities will not be entitled to participate in any return of capital on those Securities during the escrow period applicable to those Securities except as permitted by the Listing Rules or the ASX; and
- (e) if a holder of Restricted Securities breaches a Restriction Deed or a provision of this Constitution restricting a Disposal of those Securities, the holder will not be entitled to any dividend or distribution, or to exercise any voting rights, in respect of those Securities for so long as the breach continues.

#### 2.13 Non-Issue or Cancellation of Certificate

Notwithstanding any other provision of this Constitution, the Company need not issue a certificate, and may cancel any certificate without issuing a certificate in substitution, in respect of any Shares or Share Options of the Company in any circumstances where the non-issue or cancellation of that certificate is permitted by the Corporations Act, the Listing Rules or the ASX Settlement Operating Rules.

#### 2.14 No Prohibition on Foreign Ownership

Nothing in this Constitution shall have the effect of limiting or restricting the ownership of any securities of the Company by foreign persons except where such limits or restrictions are prescribed by Australian law.

#### 2.15 Payment of Interest out of Capital

Where any Shares are issued for the purpose of raising money to defray the expenses of the construction of any works or buildings or the provision of any plant which cannot be made profitable for a lengthened period the Company may pay interest on so much of such share capital as is paid up for the period and may charge this interest to capital as part of the cost of construction of the works, buildings or plant.

#### 3. MINIMUM SHAREHOLDING

#### 3.1 Effect of this Clause

The provisions of this clause have effect notwithstanding any other provision of this Constitution, except clause 33.

#### 3.2 Definitions

In this clause:

**Authorised Price** means the price per share of the Listed Securities equal to the simple average of the last closing price of the Listed Securities quoted on ASX for each of the ten trading days immediately preceding the date of any offer received by the Company pursuant to clause 3.5.

**Date of Adoption** means the date upon which this clause is inserted in this Constitution by special resolution of the members of the Company.

Date of Effect has the meaning given in clause 3.13.

**Minimum Shareholding** means a number of shares equal to a "marketable parcel" of Listed Securities within the meaning of the Listing Rules.

**Minority Member** means a member holding less than the Minimum Shareholding on or at any time after the Date of Adoption.

**New Minority Member** means a member who is the holder or a joint holder of a New Minimum Shareholding.

**New Minimum Shareholding** means a holding of shares in the same class created after the Date of Adoption by the transfer of a parcel of shares the aggregate market price of which, at the time at which a transfer of those Listed Securities was initiated or a paper based transfer of those Listed Securities was lodged with the Company, was less than a marketable parcel.

**Purchaser** means the person or persons (including one or more members) whose offer or offers to purchase Listed Securities is or are accepted by the Company.

#### 3.3 Minimum Shareholding

Subject to clauses 3.13 and 3.14, on and from the Date of Effect, the shareholding of a member which is less than the Minimum Shareholding may be sold by the Company pursuant to the provisions of this clause 3.

#### 3.4 Sale of Listed Securities of Minority Member

Subject to clauses 3.13 and 3.14, on and from the Date of Effect, each Minority Member shall be deemed to have irrevocably appointed the Company as his agent:

- (a) to sell all the Listed Securities held by him at a price not less than the Authorised Price and without any cost being incurred by the Minority Member;
- (b) to deal with the proceeds of the sale of those Listed Securities in accordance with this clause; and
- (c) where the Listed Securities are CHESS Approved Securities held in uncertificated form, to initiate a Holding Adjustment (as defined in the ASX Settlement Operating Rules) to move the securities from the CHESS Holding (as defined in the ASX Settlement Operating Rules) of the Minority Member to an Issuer Sponsored or Certificated Holding (as defined in the ASX Settlement Operating Rules) for the sale of the Listed Securities.

#### 3.5 Acceptance of Offer

Where the Company receives an offer for the purchase of all the Listed Securities of a Minority Member to whom this clause applies at the date of the offer at a price not less than the Authorised Price, the Company may accept the offer on behalf of that Minority Member.

#### 3.6 Appointment of Attorney

The Company shall, by instrument in writing, appoint a person or persons to act as attorney or attorneys of each Minority Member to whom this clause applies, to execute an instrument or instruments of transfer of their Listed Securities to the Purchaser.

#### 3.7 Transfer

Where:

- (a) all the Listed Securities of each Minority Member to whom this clause applies at any time are sold to one Purchaser; or
- (b) all the Listed Securities of two or more Minority Members to whom this clause applies at any time are sold to one Purchaser,

the transfer may be effected by one instrument of transfer.

#### 3.8 Proceeds of Sale

The Company shall receive the aggregate proceeds of the sale of all of the Listed Securities of each Minority Member to whom this clause applies at any time and shall:

- (a) immediately cause the name of the Purchaser to be entered in the Register of Shareholders as the holder of the Listed Securities sold; and
- (b) within fourteen days of receipt of the relevant share certificate or otherwise as soon as is practicable, cause the pro rata proportions of the proceeds attributable to each Minority Member to be sent to each Minority Member by electronic transfer or cheque mailed to his address in the Register of Shareholders (or in the case of joint holders, to the address of the holder whose name is shown first in the Register of Shareholders), this cheque or electronic transfer to be made payable to the Minority Member (or, in the case of joint holders, to them jointly). In the case where a Minority Member's whereabouts are unknown or where a Minority Member fails to return the share certificate or certificates (where required) relating to the Listed Securities sold, the proceeds of sale shall be applied in accordance with the applicable laws dealing with unclaimed moneys.

#### 3.9 Receipt of Proceeds

The receipt by the Company of the proceeds of sale of Listed Securities of a Minority Member shall be a good discharge to the Purchaser of all liability in respect of the purchase of the Listed Securities.

#### 3.10 Registration of Purchaser

Upon entry of the name of the Purchaser in the Register of Shareholders as the holder of the Listed Securities of a Minority Member to whom this clause applies:

- (a) the Purchaser shall not be bound to see to the regularity of the actions and proceedings of the Company pursuant to this Constitution or to the application of the proceeds of sale; and
- (b) the validity of the sale shall not be impeached by any person.

#### 3.11 Remedies Limited

The remedy of any Minority Member to whom this clause applies in respect of the sale

of his or her Listed Securities is expressly limited to a right of action in damages against the Company to the exclusion of any other right, remedy or relief against any other person.

#### 3.12 Cost of Sale of Listed Securities

The Company shall bear all the costs of the sale of the Listed Securities.

#### 3.13 Exemption from clause 3

- (a) The Company must give written notice to a Minority Member and, where the Shares are CHESS Approved Securities, to the Controlling Participant (as defined in the ASX Settlement Operating Rules) for the holding of the Minority Member, advising of the Company's intention to sell his or her shareholding pursuant to this clause 3.
- (b) Unless the Minority Member, within 6 weeks from the date the notice was sent from the Company in accordance with this clause 3, gives written notice to the Company that it desires its shareholding to be exempted from clause 3, then the Company will be free to sell the Shares held by the relevant Minority Member immediately following expiry of the 6 week period in accordance with this clause 3 (**Date of Effect**).
- (c) Where Shares are CHESS Approved Securities, a written notice by the Company in terms of this clause shall comply with the ASX Settlement Operating Rules.

#### 3.14 Notice to Exempt

Where a Minority Member has given written notice to the Company that it desires its shareholding to be exempted from clause 3 it may, at any time prior to the sale of the Listed Securities under clause 3.8, revoke or withdraw that notice. In that event the provisions of clause 3 shall apply to the Minority Member.

#### 3.15 Takeover Offer or Announcement

The Company shall not commence to sell Listed Securities comprising less than a Minimum Shareholding following the announcement of a takeover offer or takeover announcement for the Company. If a takeover bid is announced after a notice is given but before an agreement is entered into for the sale of the Listed Securities, this clause 3 ceases to operate for those Listed Securities. However, despite clause 3.16, a new notice under clause 3.13 may be given after the offer period if the takeover bid closes.

#### 3.16 Use by Company of Clause 3

Subject to clause 3.15, this clause 3 may be invoked only once in any twelve month period after its adoption or re-adoption.

#### 3.17 Notice to New Minority Members

If the Directors determine that a member is a New Minority Member, the Company may give the member notice in writing stating that the member is a New Minority Member, specifying the number of shares making up the New Minimum Shareholding, the market price of those Listed Securities and the date on which the market price was determined and stating that the Company intends to sell the Listed Securities specified in the notice in accordance with the provisions of its Constitution. Unless the Directors determine otherwise, if the Company gives such a notice, all rights of the member to vote and to receive dividends in respect of the shares specified in the notice are suspended until the Listed Securities are sold or that member ceases to be a New Minority Member and any dividends that would, but for this clause 3.17, have been paid to that member must be held by the Company and paid to that member within 30 days after the earlier of:

- (a) the date the Listed Securities specified in the notice are transferred; and
- (b) the date that the Company ceases to be entitled to sell those Listed Securities under the sale notice.

#### 4. UNCERTIFICATED HOLDINGS AND ELECTRONIC TRANSFERS

#### 4.1 Electronic or Computerised Holding

The Directors may do anything they consider necessary or desirable and which is permitted under the Corporations Act and the Listing Rules to facilitate the participation by the Company in the CHESS System and any other computerised or electronic system established or recognised by the Corporations Act or the Listing Rules for the purposes of facilitating dealings in Shares or securities.

#### 4.2 Statement of Holdings

Where the Directors have determined not to issue share certificates or to cancel existing Share certificates, a Shareholder shall have the right to receive such statements of the holdings of the Shareholder as are required to be distributed to a Shareholder under the Corporations Act or the Listing Rules.

#### 4.3 Share Certificates

If the Directors determine to issue a certificate for Shares held by a Shareholder, the provisions in relation to Share certificates contained in clause 2 shall apply.

#### 4.4 Listing Rules

The Company shall comply with the Listing Rules and the ASX Settlement Operating Rules in relation to the CHESS System.

#### 5. LIEN

#### 5.1 Lien for Members Debts

The Company has a first and paramount lien on each Share (except where the Share is a Listed Security and is fully paid up) registered in a Shareholder's name in respect of all money owed to the Company by the Shareholder (including any money payable under clause 5.2 to the extent that the Company has made a payment in respect of a liability or a requirement referred to in that clause) but not any unpaid call once the Share has been forfeited under section 254Q of the Corporations Act. The lien extends to reasonable interest and expenses incurred because the amount is not paid.

#### 5.2 Generally

Whenever any law for the time being of any country, state or place imposes or purports to impose any immediate or future possible liability upon the Company to make any payments or empowers any government or taxing authority or governmental official to require the Company to make any payment in respect of any Shares held either jointly or solely by any Shareholder, or in respect of any transfer of Shares, or of any dividends, bonuses or other moneys due or payable or accruing due or which may become due or payable to such Shareholder by the Company on or in respect of any Shares or for or on account or in respect of any Shareholder, and whether in consequence of:

- (a) the death of such Shareholder;
- (b) the non-payment of any income tax or other tax by such Shareholder;
- (c) the non-payments of any estate, probate, succession or death, duty or of any other Duty by the executor or administrator of such Shareholder or by or out of his estate; or
- (d) any other act or thing,

the Company in every case:

- (e) shall be fully indemnified by such Shareholder or his executor or administrator from all liability;
- (f) shall have a lien upon all dividends, bonuses and other moneys payable in respect of the Shares held either jointly or solely by this Shareholder for all moneys paid by the Company in respect of the Shares or in respect of any dividend, bonus or other money or for an account or in respect of this Shareholder under or in consequence of any law, together with interest at the Prescribed Rate from date of payment to date of repayment, and may deduct or set off against any dividend, bonus or other moneys so paid or payable by the Company together with interest at the Prescribed Rate;
- (g) may recover as a debt due from this Shareholder or his or her executor or administrator, wherever constituted or situate, any moneys paid by the Company under or in consequence of any such law and interest on these moneys at the Prescribed Rate and for the period mentioned above in excess of any dividend, bonus or other money as mentioned above then due or payable by the Company to such Shareholder; and

(h) may, subject to the Listing Rules, if any such money be paid or payable by the Company under any such law, refuse to register a transfer of any Shares by this Shareholder or his executor or administrator until the money and interest mentioned above is set off or deducted or, in case the money and interest exceeds the amount of any dividend, bonus or other money then due or payable by the Company to the Shareholder, until this excess is paid to the Company.

Nothing contained in this clause shall prejudice or affect any right or remedy which any law may confer or purport to confer on the Company, and, as between the Company and every such Shareholder, his or her executor, administrator and estate, wherever constituted or situate, any right or remedy which this law shall confer on the Company shall be enforceable by the Company.

#### 5.3 Exemption

The Directors may at any time exempt a Share wholly or in part from the provisions of this clause 5.

#### 5.4 Dividends

Whenever the Company has a lien on a Share, the lien extends to all dividends payable in respect of the Share.

#### 5.5 Sale of Shares

Subject to clause 5.6, the Company may sell, in such manner as the Directors think fit, any Shares on which the Company has a lien.

#### 5.6 Restrictions on Sale

A Share on which the Company has a lien shall not be sold unless:

- (a) the sum in respect of which the lien exists is presently payable; and
- (b) the Company has, not less than 14 days before the date of the sale, given to the registered holder for the time being of the Share or the person entitled to the Share by reason of the death or bankruptcy of the registered holder a notice in writing setting out, and demanding payment of, that part of the amount in respect of which the lien exists as is presently payable.

#### 5.7 Person Authorised to Sign Transfers

For the purpose of giving effect to a sale of a Share under clause 5.5, the Directors may authorise a person to transfer the Shares sold to the purchaser of the Shares. The Company shall register the purchaser as the holder of the Shares comprised in any such transfer and he or she is not bound to see to the application of the purchase money. The title of the purchaser to the Shares is not affected by any irregularity or invalidity in connection with the sale.

#### 5.8 Proceeds of Sale

The proceeds of a sale under clause 5.5 shall be applied by the Company in payment of that part of the amount in respect of which the lien exists as is presently payable, and the residue (if any) shall (subject to any like lien for sums not presently payable that existed upon the Shares before the sale) be paid to the person entitled to the Shares at the date of the sale.

#### 5.9 Protection of Lien under ASX Settlement Operating Rules

The Company may do all such things as may be necessary or appropriate for it to do under the ASX Settlement Operating Rules to protect any lien, charge or other right to which it may be entitled under any law or this Constitution.

#### 5.10 Further Powers re Forfeited Shares and Liens

Where a transfer following the sale of any Shares after forfeiture or for enforcing a lien, charge or right to which the Company is entitled under any law or under this Constitution is effected by an ASX Settlement Transfer, the Company may do all things necessary or desirable for it to do under the ASX Settlement Operating Rules in relation to that transfer.

#### 6. CALLS ON SHARES

#### 6.1 Calls

- (a) The Directors may by resolution make calls on Shareholders of partly paid Shares to satisfy the whole or part of the debt owing on those Shares provided that the dates for payment of those Shares were not fixed at the time of issue.
- (b) A call shall be deemed to have been made at the time when the resolution of the Directors authorising the call was passed.
- (c) A call may be required or permitted to be paid by instalments.
- (d) Failure to send a notice of a call to any Shareholder or the non-receipt of a notice by any Shareholder does not invalidate the call.

#### 6.2 Payment of Calls

A Shareholder to whom notice of a call is given in accordance with this Constitution must pay to the Company the amount called in accordance with the notice.

#### 6.3 Quoted Shares

(a) The Directors must not make the date for payment of calls, (**Due Date**), for Shareholders who hold quoted partly paid Shares, less than 30 Business Days and no more than 40 Business Days from the date the Company dispatches notices to relevant Shareholders that a call is made.

- (b) If after a call is made, new Shareholders purchase the same class of Share subject to the call, or if the holdings of the original Shareholders on whom the call was made change, Directors must dispatch a notice informing these Shareholders that a call has been made at least 4 days before the Due Date.
- (c) The Company must enter a call payment on the Company register no more than 5 Business Days after the Due Date.

#### 6.4 Unquoted Shares

The Directors must not make the Due Date for Shareholders who hold unquoted partly paid Shares, less than 5 Business Days from the date the Company dispatches notices to relevant Shareholders that a call is made.

#### 6.5 Joint Liability

The joint holders of a Share are jointly and severally liable to pay all calls in respect of the Share.

#### 6.6 Deemed Calls

Any amount that, by the terms of issue of a Share, becomes payable on allotment or at a fixed date, shall for the purposes of this Constitution be deemed to be a call duly made and payable, and, in case of non-payment, all the relevant provisions of this Constitution as to payment of interest and expenses, forfeiture or otherwise apply as if the amount had become payable by virtue of a call duly made and notified.

#### 6.7 Differentiation between Shareholders

The Directors may, on the issue of Shares, differentiate between the holders as to the amount of calls to be paid and the times of payment.

#### 6.8 Payments in Advance of Calls

The Directors may accept from a Shareholder the whole or any part of the amount unpaid on a Share even if no part of that amount has been called up, in which case the Directors shall nominate whether the amount so paid is to be treated as capital or a loan to the Company by the Shareholder, and:

- (a) if the amount paid is nominated to be capital, it shall be deemed as from the date of the nomination to have been applied in paying up (so far as it will extend) the unpaid balance of the total issue price of the Share, but the dividend entitlement attaching to the Share shall remain as it was prior to the payment so made until there is a call in respect of the Share under this clause 6 of an amount equal to or greater than the amount so paid; or
- (b) if the amount paid is nominated to be a loan to the Company, it shall carry interest at a rate, not exceeding the Prescribed Rate, as is agreed between the Directors and the Shareholder, shall not be repayable unless the Directors so determine, shall not confer on the Shareholder any rights attributable to subscribed capital, and shall, unless so repaid, be applied in payment of calls on the Share as and when the calls become due.

#### 6.9 Outstanding Moneys

Any moneys payable in respect of a call made in accordance with this Constitution which remain outstanding shall from and including the day for payment until the date payment is received bear interest at the Prescribed Rate.

#### 6.10 Revocation/Postponement or Extension

The Directors may revoke or postpone a call or extend time for payment in accordance with the Listing Rules and/or the Corporations Act, if revocation or postponement is not prohibited by either.

#### 6.11 Compliance with Listing Rules and Corporations Act

The Company shall comply with the Listing Rules and the Corporations Act in relation to calls. All Listing Rule requirements in relation to calls are not covered in this Constitution.

#### 6.12 Waive

The Directors may, to the extent the law permits, waive or compromise all or part of any payment due to the Company under the terms of issue of a Share under this clause 6.

#### 7. FORFEITURE OF SHARES

#### 7.1 Failure to Pay Call

If a Shareholder fails to pay a call or instalment of a call on the day appointed for payment of the call or instalment, the Directors may, at any time after this day during the time any part of the call or instalment remains unpaid (but subject to this clause 7.1) serve a notice on such Shareholder requiring payment of so much of the call or instalment as is unpaid, together with any interest that has accrued and all costs and expenses incurred by the Company as a result of the non-payment. The notice shall name a further day being not less than 14 days after the date of notice on or before which the payment required by the notice is to be made and shall state that, in the event of non-payment at or before the time appointed, the Shares in respect of which the call was made will be liable to be forfeited.

#### 7.2 Forfeiture

If the requirements of a notice served under clause 7.1 are not complied with, any Share in respect of which a call is unpaid at the expiration of 14 days after the day for its payment may be forfeited by a resolution of the Directors to that effect. Such a forfeiture shall include all dividends and other distributions declared in respect of the forfeited Shares and not actually paid or distributed before the forfeiture.

#### 7.3 Sale of Forfeited Shares

Subject to the Corporations Act and the Listing Rules, a forfeited Share may be sold or otherwise disposed of on the terms and in the manner that the Directors determine and, at any time before a sale or disposition, the forfeiture may be cancelled on the terms the Directors determine.

#### 7.4 Continuing Liability

A person whose Shares have been forfeited ceases to be a Shareholder in respect of the forfeited Shares, but remains liable to pay the Company all money that, at the date of forfeiture, was payable by him to the Company in respect of the Shares (including interest at the Prescribed Rate from the date of forfeiture on the money for the time being unpaid if the Directors decide to enforce payment of the interest), but his or her liability ceases if and when the Company receives payment in full of all the money (including interest) payable in respect of the Shares.

#### 7.5 Officer's Statement Prima Facie Evidence

A statement in writing declaring that the person making the statement is a Director or a Secretary of the Company, and that a Share in the Company has been duly forfeited on a date stated in the statement, is prima facie evidence of the facts stated in the statement as against all persons claiming to be entitled to the Share.

#### 7.6 Procedures

The Company may receive the consideration (if any) given for a forfeited Share on any sale or disposition of the Share, execute a transfer of the Share in favour of the person to whom the Share is sold or disposed of and take all other steps necessary or desirable to transfer or dispose of those shares to the relevant transferee. Upon the execution of the transfer, the transferee shall be registered as the holder of the Share and is not bound to see to the application of any money paid as consideration. The title of the transferee to the Share is not affected by any irregularity or invalidity in connection with the forfeiture, sale or disposal of the Share.

#### 7.7 Listing Rules and ASX Settlement Operating Rules

The Company shall comply with the Listing Rules with respect to forfeited Shares and may do all such things as may be necessary or appropriate for it to do under the ASX Settlement Operating Rules to protect any lien, charge or other right to which it may be entitled under any law or this Constitution.

#### 7.8 Waive

The Directors may:

- (a) exempt a Share from all or part of this clause 7;
- (b) waive or compromise all or part of any payment due to the Company under this clause 7; and
- (c) before a forfeited Share has been sold, reissued and otherwise disposed of, cancel the forfeiture on the conditions they decide.

#### 8. TRANSFER OF SHARES

#### 8.1 Form of Transfer

Subject to this Constitution, Shareholders may transfer any Share held by them by:

- (a) an ASX Settlement Transfer or any other method of transferring or dealing in Shares introduced by ASX or operated in accordance with the ASX Settlement Operating Rules or Listing Rules and in any such case recognised under the Corporations Act; or
- (b) an instrument in writing in any usual or common form or in any other form that the Directors approve.

#### 8.2 CHESS Transfers

- (a) The Company must comply with all obligations imposed on the Company under the Corporations Act, the Listing Rules and the ASX Settlement Operating Rules in respect of an ASX Settlement Transfer or any other transfer of Shares.
- (b) Notwithstanding any other provision in this Constitution, the Company must not prevent, delay or interfere with the registration of an ASX Settlement Transfer or any other transfer of Shares.

#### 8.3 Participation in CHESS

The Directors may do anything they consider necessary or desirable and which is permitted under the Corporations Act, the Listing Rules and the ASX Settlement Operating Rules to facilitate participation by the Company in any system established or recognised by the Corporations Act and the Listing Rules or the ASX Settlement Operating Rules in respect of transfers of or dealings in marketable securities.

#### 8.4 Registration Procedure

Where an instrument of transfer referred to in clause 8.1(b) is to be used by a Shareholder to transfer Shares, the following provisions apply:

- (a) the instrument of transfer must be executed by or on behalf of both the transferor and the transferee unless it is a sufficient transfer of marketable securities within the meaning of the Corporations Act and any Duty duly paid if required by law;
- (b) the instrument of transfer shall be left at the Registered Office for registration accompanied by the certificate for the Shares to be transferred (if any) and such other evidence as the Directors may require to prove the title of the transferor and his right to transfer the shares;
- (c) subject to clause 33, a reasonable fee may be charged on the registration of a transfer of Shares or other securities; and
- (d) on registration of a transfer of Shares, the Company must cancel the old certificate (if any).

#### 8.5 Power to Refuse to Register

The Directors may refuse to register any transfer of Shares (other than an ASX Settlement Transfer) where:

- (a) the Listing Rules permit the Company to do so;
- (b) the Listing Rules require the Company to do so; or
- (c) the transfer is a transfer of Restricted Securities which is or might be in breach of the Listing Rules or any escrow agreement entered into by the Company in relation to such Restricted Securities pursuant to the Listing Rules.

Where the Directors refuse to register a transfer in accordance with this clause, they shall send notice of the refusal and the precise reasons for the refusal to the transferee and the lodging broker (if any) in accordance with the Listing Rules.

#### 8.6 Closure of Register

Subject to the Listing Rules and the ASX Settlement Operating Rules, the Register of Shareholders may be closed during such time as the Directors may determine, not exceeding 30 days in each calendar year or any one period of more than 5 consecutive Business Days.

#### 8.7 Retention of Transfers by Company

All instruments of transfer which are registered will be retained by the Company, but any instrument of transfer which the Directors decline or refuse to register (except in the case of fraud) shall on demand be returned to the transferee.

#### 8.8 Power to suspend registration of transfers

The Directors may suspend the registration of transfers at any times, and for any periods, permitted by the ASX Settlement Operating Rules that they decide.

#### 8.9 Powers of Attorney

Any power of attorney granted by a Shareholder empowering the recipient to transfer Shares which may be lodged, produced or exhibited to the Company or any Officer of the Company will be taken and deemed to continue and remain in full force and effect, as between the Company and the grantor of that power, and the power of attorney may be acted on, until express notice in writing that it has been revoked or notice of the death of the grantor has been given and lodged at the Office or at the place where the Register of Shareholders is kept.

#### 8.10 Other Securities

The provisions of this clause 8 shall apply, with necessary alterations, to any other Listed Securities for the time being issued by the Company.

#### 8.11 Branch Register

The Company may cause a Register of Shareholders to be kept in any place (including without limitation, a branch register) and the Directors may from time to time make such provisions as they (subject to the Corporations Act, the Listing Rules and the ASX Settlement Operating Rules) may think fit with respect to the keeping of any such Register.

#### 8.12 Compliance with ASX Settlement Operating Rules

The Company shall comply with the ASX Settlement Operating Rules and the Listing Rules in relation to all matters covered by those rules.

#### 8.13 Issuer Sponsored Subregister

The Company may establish and maintain an issuer sponsored subregister in compliance with any relevant provisions of the Corporations Act, the Listing Rules or the ASX Settlement Operating Rules.

#### 8.14 Transferor Holds Shares until Registration of Transfer

A transferor of Shares remains the registered holder of the Shares transferred until an ASX Settlement Transfer has taken effect in accordance with the ASX Settlement Operating Rules or the transfer is registered in the name of the transferee and is entered in the Register of Shareholders in respect of them, whichever is the earlier.

#### 8.15 Waive

The Directors may, to the extent the law permits, waive any of the requirements of this clause 8 and prescribe alternative requirements instead.

#### 9. TRANSMISSION OF SHARES

#### 9.1 Death of Shareholder Leaving a Will

On the death of a Shareholder who leaves a will appointing an executor, the executor shall be entitled as from the date of death, and on behalf of the deceased Shareholder's estate, to the same dividends and other advantages and to the same rights whether in relation to meetings of the Company, or voting or otherwise, as the Shareholder would have been entitled to if he or she had not died, whether or not probate of the will has been granted. Nevertheless, if probate of the will is granted to a person or persons other than the executor first referred to in this clause 9, his or her executor's rights shall cease, and these rights shall only be exercisable by the person or persons to whom probate is granted as provided in clauses 9.2 and 9.3. The estate of a deceased Shareholder will not be released from any liability to the Company in respect of the Shares.

#### 9.2 Death or Bankruptcy of Shareholder or the Shareholder becomes of unsound mind

Subject to clause 9.1, where the registered holder of a Share dies, becomes bankrupt, or the Shareholder becomes of unsound mind, his or her personal representative or the trustee of his or her estate, as the case may be, shall be entitled upon the production of such information as is properly required by the Directors, to the same dividends and other advantages, and to the same rights (whether in relation to

meetings of the Company, or to voting or otherwise), as the registered holder would have been entitled to if he or she had not died or become bankrupt.

#### 9.3 Registration by Transmission or to Beneficiary

A person becoming entitled to a Share in consequence of the death or, subject to the *Bankruptcy Act* 1966, the bankruptcy of a Shareholder or the Shareholder becoming of an unsound mind may, upon information being produced that is properly required by the Directors, elect by written notice to the Company either to be registered himself or herself as holder of the Share or to have some other person nominated by the person registered as the transferee of the Share. If this person elects to have another person registered, he or she shall execute a transfer of the Share to that other person.

#### 9.4 Limitations to Apply

All the limitations, restrictions and provisions of this Constitution relating to the right to transfer Shares and the registration of a transfer of Shares are applicable to any notice or transfer as if the death, bankruptcy of the Shareholder or on the Shareholder becoming of unsound mind had not occurred and the notice or transfer were a transfer signed by that Shareholder.

#### 9.5 Death of a Joint Holder

In the case of the death of a Shareholder who was a joint holder, the survivor or survivors shall be the only persons recognised by the Company as having any title to the deceased's interest in the Shares, but this clause 9.5 does not release the estate of a deceased joint holder from any liability in respect of a Share that had been jointly held by this person with one or more other persons.

#### 9.6 Joint Personal Representatives

Where two or more persons are jointly entitled to any Share in consequence of the death of the registered holder, they shall, for the purpose of this Constitution, be deemed to be joint holders of the Share.

#### 9.7 ASX Settlement Transfer

In the case of an ASX Settlement Transfer the provisions of this clause 9 are subject to any obligation imposed on the Company or the person entitled to the relevant Shares on the death or bankruptcy of a member by the Listing Rules, the ASX Settlement Operating Rules or any law.

#### 9.8 Joint Holders

If more than three persons are registered as holders of Shares in the Company in the Register of Shareholders (or a request is made to register more than three persons), then only the first three persons will be regarded as holders of Shares in the Company and all other names will be disregarded by the Company for all purposes.

#### 10. CHANGES TO CAPITAL STRUCTURE

#### 10.1 Alterations to Capital

Subject to the Corporations Act and the Listing Rules, the Company may, by ordinary resolution:

- (a) issue new Shares of such amount specified in the resolution;
- (b) consolidate and divide all or any of its Shares into Shares of larger amount than its existing Shares;
- (c) subject to the Listing Rules, sub-divide all or any of its Shares into Shares of smaller amount, but so that in the sub-division the proportion between the amount paid and the amount (if any) unpaid on each such Share of a smaller amount remains the same; and
- (d) cancel Shares that, at the date of the passing of the resolution, have not been taken or agreed to be taken by any person or have been forfeited and, subject to the Corporations Act, reduce the amount of its share capital by the amount of the Shares so cancelled,

and the Directors may take such action as the Directors think fit to give effect to any resolution altering the Company's share capital.

#### 10.2 Reduction of Capital

Subject to the Corporations Act and the Listing Rules, the Company may reduce its share capital in any way including, but not limited to, distributing to shareholders securities of any other body corporate and, on behalf of the shareholders, consenting to each shareholder becoming a member of that body corporate and agreeing to be bound by the constitution of that body corporate.

#### 10.3 Buy-Backs

- (a) In this clause "Buy-Back Provisions" means the provisions of Part 2J.1 Division 2 of the Corporations Act.
- (b) The Company may, subject to the Corporations Act and the Listing Rules and in accordance with the Buy-Back Provisions, purchase its own Shares on such terms and at such times as may be determined by the Directors from time to time.
- (c) The Company may give financial assistance to any person or entity for the purchase of its own Shares in accordance with the Buy-Back Provisions on such terms and at such times as may be determined by the Directors from time to time.

#### 10.4 Fractions

If as a result of any issue of shares or any alteration to the Company's share capital any Shareholders would become entitled to fractions of a share, the Directors may deal with those fractions as the Directors think fit including by:

- (a) ignoring fractional entitlements or making cash payments in lieu of fractional entitlements;
- (b) appointing a trustee to deal with any fractions on behalf of Shareholders; and
- (c) rounding up each fractional entitlement to the nearest whole share by capitalising any amount available for capitalisation under clause 23.1 even though only some of the Shareholders participate in the capitalisation.

#### 11. WRITTEN RESOLUTIONS

Where the Company has only one Shareholder, to the extent permitted by law, a resolution in writing signed by that Shareholder, shall be as valid and effectual as if it had been passed at a meeting of Shareholders duly convened and held. A facsimile transmission, an email bearing the signature of the Shareholder or an email of the Shareholder addressed to an officer of the Company confirming agreement with the resolution and undertaking to sign the resolution as soon as practicable shall be deemed to be a document in writing signed by the Shareholder.

#### 12. GENERAL MEETINGS

#### 12.1 Convening of General Meetings of Shareholders by Directors' Resolution

The Directors may, by a resolution passed by a majority of Directors, convene a general meeting of Shareholders in accordance with this clause 11 and the requirements of the Corporations Act.

#### 12.2 Change of place or postponement of a General Meeting of Shareholders

The Directors may, subject to the Corporations Act and the Listing Rules, postpone a meeting of Shareholders or change the place for a general meeting of Shareholders by giving written notice to ASX. If a meeting of Shareholders is postponed for one month or more, the Company must give new notice of the postponed meeting. The only business that may be transacted at a general meeting the holding of which is postponed is the business specified at the original meeting.

#### 12.3 Convening of General Meetings of Shareholders by a Director or requisition

Any Director may, whenever he or she thinks fit, convene a general meeting of Shareholders, and a general meeting shall also be convened on requisition as is provided for by the Corporations Act, or in default, may be convened by such requisitions as empowered to do so by the Corporations Act. If there are no Directors for the time being, a Secretary may convene a general meeting of Shareholders for the purpose of enabling the election of Directors but for no other purpose. A general meeting may be held at two or more venues simultaneously using any technology that gives the Shareholders as a whole a reasonable opportunity to participate.

# 12.4 Cancellation of a General Meeting of Shareholders

- (a) A general meeting of Shareholders convened by the Directors in accordance with clause 12.1 may be cancelled by a resolution passed by a majority of Directors.
- (b) A general meeting of shareholders convened on a requisition as provided for by the Corporations Act, may, if the application for requisition is withdrawn in writing, be cancelled by a resolution passed by a majority of Directors.
- (c) Notice of the cancellation of a general meeting of Shareholders must be given to the Shareholders in accordance with clause 26, but notice of such cancellation must be given to each Shareholder not less than two (2) days prior to the date on which the meeting was proposed to be held.

# 12.5 Notice

A notice of a general meeting shall be given in accordance with the requirements of the Corporations Act, clause 26 and the Listing Rules, and:

- (a) must specify the place, the day and the time of the meeting;
- (b) must state the general nature of the business to be transacted at the meeting;
- (c) must, if a special resolution is proposed at the meeting, set out an intention to propose the special resolution and state the resolution;
- (d) must include such statements about the appointment of proxies as are required by the Corporations Act;
- (e) must specify a place and fax number for the purposes of receipt of proxy appointments; and
- (f) may specify an electronic address for the purposes of receipt of proxy appointments,

and shall include any other information required to be included in the notice by the Listing Rules. The non-receipt of a notice of a general meeting by a Shareholder or the accidental omission to give this notice to a Shareholder shall not invalidate any resolution passed at the meeting.

### 12.6 Irregularities in giving notice

A person who attends any general meeting waives any objection that the person may have to any failure to give notice or any other irregularity in the notice of that meeting unless that person objects to the holding of the meeting at the start of the meeting. The accidental failure to give notice of a general meeting to, or the nonreceipt of the notice by, any person entitled to receive notice of that meeting does not invalidate the proceedings at the meeting or any resolution passed at that meeting.

# 12.7 Business at General Meeting

Subject to the Corporations Act, only matters that appear in a notice of meeting shall be dealt with at a general meeting or an annual general meeting, as the case may be.

# 12.8 Notice to Home Branch

- (a) The Company shall notify the Home Branch of any meeting at which Directors are to be elected at least 5 Business Days before the closing day for receipt of nominations for Directors, and in any other case (other than a meeting to pass a special resolution) at least 10 Business Days before the meeting is held, and in the case of a meeting convened to pass a special resolution, at least 15 Business Days before the meeting is held. All notices convening meetings shall specify the place, date and hour of the meeting, and shall set out all resolutions to be put to the meeting.
- (b) The Company shall notify the Home Branch as soon as is practicable after any general meeting in the case of special business as to whether or not the resolutions were carried and in the case of ordinary business as to which of those resolutions were not carried or were amended or were withdrawn.

# 12.9 Annual General Meeting

An annual general meeting shall be held in accordance with the requirements of the Corporations Act.

# 13. PROCEEDINGS AT GENERAL MEETINGS

# 13.1 Quorum

No business, the election of a chairman and the adjournment of the meeting, shall be transacted at any general meeting unless a quorum is present comprising two Shareholders present in person, by proxy, attorney or Representative. For the purpose of determining whether a quorum is present, a person attending as a proxy, attorney or Representative, shall be deemed to be the Shareholder present in person. If a quorum is not present within 30 minutes after the time appointed for a general meeting, the meeting, if convened upon a requisition shall be dissolved, but in any other case, it shall stand adjourned to a date and at the time and place to be fixed by the Directors. If at such adjourned meeting a quorum is not present within 30 minutes after the time appointed for the adjourned meeting, the meeting is dissolved.

# 13.2 Persons Entitled to Attend a General Meeting

The persons entitled to attend a general meeting shall be:

- (a) Shareholders, in person, by proxy, attorney or Representative;
- (b) Directors and public officers of the Company;
- (c) the Company's auditor; and
- (d) any other person or persons as the chairman may approve.

# 13.3 Refusal of Admission to Meetings

The chairman of a general meeting may refuse admission to a person, or require a person to leave and not return to, a meeting if the person:

- (a) refuses to permit examination of any article in the person's possession;
- (b) is in possession of any:
  - (i) electronic or broadcasting or recording device;
  - (ii) placard or banner; or
  - (iii) other article,

which the chairman considers to be dangerous, offensive or liable to cause disruption;

- (c) causes any disruption to the meeting; or
- (d) is not entitled to attend the meeting under the Corporations Act or this Constitution.

The Chairman may delegate the powers conferred by this clause 13.3 to any person. Nothing in this clause limits the powers conferred on the chairman by law.

## 13.4 Insufficient room

The chairman may arrange for any persons attending the meeting who the chairman considers cannot reasonably be accommodated in the place where the meeting is to take place to attend or observe the meeting from a separate place using any technology that gives members present at the meeting as a whole a reasonable opportunity to participate in the meeting.

# 13.5 Chairman

The person elected as the chairman of the Directors' meeting under clause 16.9 shall, if willing, preside as chairman at every general meeting. Where a general meeting is held and a chairman has not been elected under clause 16.9 or the chairman or, in his absence, the vice-chairman is not present within 15 minutes after the time appointed for holding of the meeting or is unwilling to act:

- (a) the Directors present may elect a chairman of the meeting; or
- (b) if no chairman is elected in accordance with subsection (a), the Shareholders present shall elect one of their number to be the acting chairman of the meeting.

# 13.6 Vacating Chair

At any time during a meeting and in respect of any specific item or items of business, the chairman may elect to vacate the chair in favour of another person nominated by the chairman (which person must be a Director unless no Director is present or willing to act). That person is to be taken to be the chairman and will have all the power of the chairman (other than the power to adjourn the meeting), during the consideration of that item of business or those items of business.

# 13.7 Disputes Concerning Procedure

If there is a dispute at a general meeting about a question of procedure, the chairman may determine the question.

## 13.8 General Conduct

The general conduct of each general meeting of the Company and the procedures to be adopted at the meeting will be determined by the chairman, including the procedure for the conduct of the election of Directors.

## 13.9 Adjournment

The chairman may adjourn the meeting from time to time and from place to place, but no business shall be transacted on the resumption of any adjourned meeting other than the business left unfinished at the meeting from which the adjournment took place. A poll cannot be demanded on any resolution concerning the adjournment of a general meeting except by the chairman.

## 13.10 Notice of Resumption of Adjourned Meeting

When a meeting is adjourned for 30 days or more, notice of the resumption of the adjourned meeting shall be given in the same manner as for the original meeting, but otherwise, it is not necessary to give any notice of any adjournment or of the business to be transacted on the resumption of the adjourned meeting.

## 13.11 How resolutions are decided

Subject to the requirements of the Corporations Act, a resolution is taken to be carried if a majority of the votes cast on the resolution are in favour of it.

# 13.12 Casting Vote

In the case of an equality of votes, the chairman of the meeting shall have a second or casting vote.

# 13.13 Voting Rights

Subject to any rights or restrictions for the time being attached to any class or classes of Shares, at meetings of Shareholders or classes of Shareholders:

- (a) each Shareholder entitled to vote may vote in person or by proxy, attorney or Representative or, if a determination has been made by the Board in accordance with clause 13.35, by Direct Vote);
- (b) on a show of hands, every person present who is a Shareholder or a proxy, attorney or Representative of a Shareholder has one vote (even though he or she may represent more than one member); and

(c) on a poll, every person present who is a Shareholder or a proxy, attorney or Representative of a Shareholder (or where a Direct Vote has been lodged) shall, in respect of each fully paid Share held by him, or in respect of which he is appointed a proxy, attorney or Representative, have one vote for the Share, but in respect of partly paid Shares, shall have such number of votes being equivalent to the proportion which the amount paid (not credited) is of the total amounts paid and payable in respect of those Shares (excluding amounts credited).

# 13.14 Voting - Show of Hands

At any general meeting a resolution put to the vote of the meeting shall be decided on a show of hands unless a poll is demanded in accordance with clause 13.16.

# 13.15 Results of Voting

Unless a poll is so demanded, a declaration by the chairman that a resolution has on a show of hands been carried or carried unanimously or by a particular majority, or lost, and an entry to that effect in the book containing the minutes of the proceedings of general meetings of the Company, is conclusive evidence of the fact without proof of the number or proportion of the votes recorded in favour of or against the resolution.

# 13.16 Poll

A poll may be demanded before or immediately upon the declaration of the result of the show of hands by:

- (a) the chairman of the general meeting;
- (b) at least 5 Shareholders present in person or by proxy, attorney or Representative having the right to vote on the resolution; or
- (c) any one or more Shareholders holding not less than 5% of the total voting rights of all Shareholders having the right to vote on the resolution.

The Chairman must demand a poll if, having regard to the number of votes cast by proxy and Direct Vote, the outcome of the poll will or may be different from the outcome of a show of hands.

# 13.17 Manner of Taking Poll

If a poll is duly demanded, it shall be taken in such manner and either at once or after an interval or adjournment or otherwise as the chairman directs, and the result of the poll shall be the resolution of the meeting at which the poll was demanded. A poll demanded on the election of a chairman or on a question of adjournment shall be taken forthwith.

### 13.18 Meeting May Continue

A demand for a poll shall not prevent the continuation of the meeting for the transaction of other business.

# 13.19 Voting by Joint Holders

In the case of joint holders of Shares, the vote of the senior who tenders a vote, whether in person or by proxy, attorney or Representative or by Direct Vote, shall be accepted to the exclusion of the votes of the other joint holders and, for this purpose, seniority shall be determined by the order in which the names stand in the Register of Shareholders.

# 13.20 Shareholder under Disability

If a Shareholder is of unsound mind or is a person whose person or estate is liable to be dealt with in any way under the law relating to mental health, his committee or trustee or any other person that properly has the management of his estate may exercise any rights of the Shareholder in relation to a general meeting as if the committee, trustee or other person were the Shareholder.

# 13.21 Payment of Calls

A Shareholder is not entitled to any vote at a general meeting unless all calls presently payable by him in respect of Shares have been paid. Nothing in this clause prevents such a Shareholder from voting at a general meeting in relation to any other Shares held by that Shareholder provided all calls and other sums payable by him have been paid on those other Shares.

## 13.22 Objection to Voting

An objection may be raised to the qualification of a voter only at the meeting or adjourned meeting at which the vote objected to is given or tendered. This objection shall be referred to the chairman of the meeting, whose decision shall be final. A vote not disallowed pursuant to such an objection is valid for all purposes.

# 13.23 Restrictions on voting

A Shareholder is not entitled to vote on a resolution at a general meeting if they are prevented from doing so by the Corporations Act, the Listing Rules or this Constitution. The Company must disregard any vote (including a Direct Vote) purported to be cast on a resolution by a member or a Representative, proxy or attorney in breach of this clause 13.23.

### 13.24 Proxies

A Shareholder who is entitled to attend and cast a vote at a general meeting may appoint a person as the Shareholder's proxy to attend and vote for the Shareholder at the general meeting. The appointment may specify the proportion or number of votes that the proxy may exercise. Each Shareholder may appoint a proxy. A Shareholder who is entitled to cast 2 or more votes at the meeting may appoint 2 proxies. If the Shareholder appoints 2 proxies and the appointment does not specify the proportion of votes that the proxy may exercise, each proxy may exercise half the votes. Any fraction of votes resulting from the application of this clause 13.23 shall be disregarded. An instrument appointing a proxy:

(a) shall be in writing under the hand of the appointor or of his attorney, or, if the appointor is a corporation, executed in accordance with the Corporations Act;

- (b) may specify the manner in which the proxy is to vote in respect of a particular resolution and, where an instrument of proxy so provides, the proxy is not entitled to vote on the resolution except as specified in the instrument;
- (c) shall be deemed to confer authority to demand or join in demanding a poll;
- (d) shall be in such form as the Directors determine and which complies with Division 6 of Part 2G.2 of the Corporations Act;
- (e) shall not be valid unless the original instrument and the power of attorney or other authority (if any) under which the instrument is signed, or a copy or facsimile which appears on its face to be an authentic copy of that proxy, power or authority, is or are deposited or sent by facsimile or electronic transmission to the Registered Office, or at such other place (being the place or being in the reasonable proximity of the place at which the meeting is to be held) as is specified for that purpose in the notice convening the meeting (with any Duty paid where necessary), by the time (being not less than 48 hours) prior to the commencement of the meeting (or the resumption of the meeting if the meeting is adjourned and notice is given in accordance with clause 13.10) as shall be specified in the notice convening the meeting (or the notice under clause 13.10, as the case may be); and
- (f) shall comply with the Listing Rules.

# 13.25 Electronic Appointment of Proxy

For the purposes of clause 13.24, a proxy appointment received at an electronic address will be taken to be signed by the appointor if:

- (a) a personal identification code allocated by the Company to the appointor has been input into the appointment;
- (b) the appointment has been verified in another manner approved by the Directors; or
- (c) is otherwise authenticated in accordance with the Corporations Act.

# 13.26 Name of proxy

A proxy form issued by the Company must allow for the insertion of the name of the person to be primarily appointed as proxy and may provide that, in circumstances and on conditions specified in the form that are not inconsistent with this Constitution, the chairman of the relevant meeting (or another person specified in the form) is appointed as proxy.

# 13.27 Incomplete proxy appointment

Where an instrument appointing a proxy has been received by the Company within the period specified in clause 13.24(e) and the Company considers that the instrument has not been duly executed or authenticated or is otherwise incomplete (other than by reason only that the name or office of the proxy has not been completed), the board, in its discretion, may:

(a) return the instrument appointing the proxy to the appointing Shareholder; and

(b) request that the appointing Shareholder take such steps to complete, sign, execute or authenticate the proxy instrument within the time period notified to the appointing Shareholder.

# 13.28 No right to speak or vote if appointing Shareholder present

The appointment of a proxy is not revoked by the appointing Shareholder attending and taking part in the meeting, unless the appointing Shareholder actually votes at the meeting on the resolution for which the proxy is proposed to be used, in which case the proxy's appointment is deemed to be revoked with respect to voting on that resolution.

# 13.29 Rights where 2 proxies or attorneys are appointed

Where a Shareholder appoints 2 proxies or attorneys to vote at the same general meeting:

- (a) on a show of hands, if more than one proxy or attorney attends, neither may vote; and
- (b) on a poll, each proxy or attorney may only exercise votes in respect of those shares or voting rights the proxy or attorney represents.

## 13.30 More than 2 proxies or attorneys appointed

If the Company receives notice of the appointment of a proxy or attorney in accordance with this Constitution that results in more than 2 proxies or attorneys being entitled to act at a general meeting then in determining which proxies or attorneys may act at that meeting:

- (a) a proxy or attorney appointed for that particular meeting may act ahead of any proxy or attorney whose appointment is a standing appointment; and
- (b) subject to clause 13.30(a) the proxies or attorneys whose appointments are received by the Company most recently in time may act.

### 13.31 Proxy Votes

A vote given in accordance with the terms of an instrument of proxy or attorney is valid notwithstanding the previous death or unsoundness of mind of the principal, the revocation of the instrument (or the authority under which the instrument was executed) or the transfer of the Share in respect of which the instrument or power is given, if no intimation in writing of the death, unsoundness of mind, revocation or transfer has been received by the Company at the Registered Office before the commencement of the meeting or adjourned meeting at which the instrument is used or the power is exercised.

# 13.32 Representatives of Corporate Shareholders

A body corporate (the **appointor**) that is a Shareholder may authorise, in accordance with section 250D of the Corporations Act, by resolution of its Directors or other governing body, such person or persons as it may determine to act as its Representative at any general meeting of the Company or of any class of Shareholders. A person so authorised shall be entitled to exercise all the rights and privileges of the appointor as a Shareholder. When a Representative is present at a general meeting of the Company, the appointor shall be deemed to be personally present at the meeting unless the Representative is otherwise entitled to be present at the meeting. The original form of appointment of a Representative, a certified copy of the appointment, or a certificate of the body corporate evidencing the appointment of a Representative is evidence of a Representative having been appointed.

# 13.33 More than one Representative present

If more than one Representative appointed by a Shareholder (and in respect of whose appointment the Company has not received notice of revocation) is present at a general meeting then:

- (a) a Representative appointed for that particular meeting may act to the exclusion of a Representative whose appointment is a standing appointment; and
- (b) subject to clause 13.33(a), the Representative appointed most recently in time may act to the exclusion of a Representative appointed earlier.

# 13.34 Rights of Representatives, proxies and attorneys

Subject to clauses 13.23 to 13.33, unless the terms of appointment of a Representative, proxy or attorney provide otherwise, the Representative, proxy or attorney:

- (a) has the same rights to speak, demand a poll, join in the demanding of a poll or act generally at the meeting as the appointing Shareholder would have if the Shareholder had been present but may not cast a vote by Direct Vote;
- (b) is taken to have authority to vote on any amendment moved to the proposed resolutions, any motion that the proposed resolutions not be put or any similar motion and any procedural resolution, including any resolution for the election of a chairman or the adjournment of a general meeting; and
- (c) may attend and vote at any postponed or adjourned meeting unless the appointing Shareholder gives the Company notice in writing to the contrary not less than 48 hours before the time to which the holding of the meeting has been postponed or adjourned.

This clause 13.34 applies even if the terms of appointment of a Representative, proxy or attorney refers to specific resolutions or to a specific meeting to be held at a specific time.

# 13.35 Board may determine Direct Voting to apply

- (a) The Board may determine that Shareholders may cast votes to which they are entitled on any or all of the resolutions (including any special resolution) proposed to be considered at, and specified in the notice convening, a meeting of Shareholders, by Direct vote.
- (b) If the Board determines that votes may be cast by Direct Vote, the Board may make such regulations as it considers appropriate for the casting of Direct Votes, including regulations for:
  - (i) the form, method and manner of voting by Direct Vote; and
  - (ii) the time by which the votes of Shareholders to be cast by Direct Vote must be received by the Company in order to be effective.
- (c) If the Board determines to allow voting by Direct Vote on a resolution at a meeting, the notice of meeting must inform shareholders of their right to vote by direct vote in respect of that resolution.

# 13.36 Direct Voting instrument – form, signature and deposit

- (a) If sent by post or fax, a Direct Vote must be signed by the Shareholder or properly authorised attorney or, if the Shareholder is a company, either under seal or by a duly authorised officer, attorney or representative.
- (b) If sent by electronic transmission, a Direct Vote is taken to have been signed if it has been signed or authorised by the Shareholder in the manner approved by the Board or specified in the notice of meeting.
- (c) At least 48 hours before the time for holding the relevant meeting, an adjourned meeting or a poll at which a person proposes to vote, the Company must receive at its registered office or such other place as specified for that purpose in the notice of meeting, or be transmitted to a facsimile number or electronic address specified for that purpose in the notice of meeting:
  - (i) the Direct Vote; and
  - (ii) if relevant, any authority or power under which the Direct Vote was signed or a certified copy of that power or authority if not already lodged with the Company.
- (d) A notice of intention of voting is valid if it contains the following information:
  - (i) the Shareholder's name and address and any applicable identifying notations such as the holder identification number or similar approved by the Board or specified in the notice of meeting; and
  - (ii) the Shareholder's voting intention on any or all of the resolutions to be put before the meeting.

# 13.37 Voting Forms

- (a) If a single voting form contains instructions for both a Direct Vote and appointment of a proxy, the Shareholder will be understood not to have appointed a proxy by exercising their right to Direct Vote pursuant to that voting form. The authority of any proxy will be revoked and only the Direct Votes will be counted.
- (b) If a single voting form is received and neither the direct voting box nor the appointment of proxy box is selected, the Shareholder will be taken to have appointed the person named in the form as proxy and if no person is named, the chair of the meeting as proxy.
- (c) The Shareholder may include in their voting form the number of shares to be voted on any resolution by inserting the percentage or number of shares. Otherwise the instructions apply to all Shares held by the Shareholder.
- (d) If more than one joint holder votes on a resolution, only the vote of the joint holder whose name appears first in the register of members is counted.

# 13.38 Direct Votes count on a poll

- (a) Direct Votes are not counted if a resolution is decided on a show of hands.
- (b) Subject to clauses 13.39 and 13.40, if a poll is held on a resolution a vote cast by Direct Vote by a Shareholder entitled to vote on the resolution is taken to have been cast on the poll as if the Shareholder had cast the vote in the poll at the meeting.
- (c) Direct Votes abstained will not be counted in computing the required majority on a poll.
- (d) If the Direct Votes lodged (together with the proxies received) could result in a different outcome from a vote on a show of hands, the Chair of the meeting should call for a poll.
- (e) A Direct Vote received by the Company on a resolution which is amended is taken to be a Direct Vote on that resolution as amended, unless the Chair of the meeting determines that this is not appropriate.
- (f) Receipt of a Direct Vote from a Shareholder has the effect of revoking (or, in the case of a standing appointment, suspending) the appointment of a proxy, attorney or representative made by the shareholder under an instrument received by the Company before the Direct Vote was received.

### 13.39 Withdrawal of a Direct Vote

A Direct Vote:

(a) may be withdrawn by the Shareholder by notice in writing received by the Company before the commencement of the meeting (or in the case of an adjournment, the resumption of the meeting;

- (b) is automatically withdrawn if:
  - (i) the Shareholder attends the meeting in person and registers to vote at the meeting (including in the case of a body corporate, by representative);
  - (ii) the Company receives from the Shareholder a further Direct Vote or Direct Votes (in which case the most recent Direct Vote is, subject to the rules in clause 13.35 to 13.40, counted in lieu of the prior Direct Vote; and
  - (iii) the Company receives, after the Direct Vote, an instrument under which a representative, proxy or attorney is appointed to act for the Shareholder at the meeting in accordance with clause 13.24 and 13.32.

A Direct Vote withdrawn under this clause 13.39 is not counted.

# 13.40 Validity of Direct Vote

- (a) A Direct Vote received by the Company is valid even if, before the meeting, the Shareholder:
  - (i) dies or becomes mentally incapacitated;
  - (ii) becomes bankrupt or an insolvent under administration or is wound up;
  - (iii) transfers the Shares in respect of which the Direct vote was given; and
  - (iv) where the Direct Vote is given on behalf of the Shareholder by an attorney, revokes the appointment of the attorney or the authority under which the appointment was made by a third party,

unless the Company has received written notice of the matter before the commencement or resumption of the meeting.

(b) A decision by the Chair of the meeting as to whether a Direct Vote is valid is conclusive.

# 13A. USE OF TECHNOLOGY AT GENERAL MEETINGS (HYBRID MEETINGS)

# 13A.1 Use of technology

- (a) To the extent permitted under the Corporations Act, Listing Rules and any other applicable law, a general meeting may be convened as a hybrid meeting, which is a physical venue that also uses virtual technology, or at two or more venues, provided that the form of technology used provides shareholders entitled to attend the meeting, as a whole, a reasonable opportunity to participate in the meeting without being physically present in the same place.
- (b) The provisions of this Constitution relating to general meetings apply, so far as they can and with any necessary changes to ensure compliance with the

Corporations Act, Listing Rules and any other applicable law, to general meetings held using that technology.

- (c) Where a general meeting is held which includes virtual technology or at two or more venues using any form of technology:
  - (i) a Shareholder participating in the meeting is taken to be present in person at the meeting;
  - (ii) any documents required or permitted to be tabled at the meeting will be taken to have been tabled at the meeting if the document is given, or made available, to the persons entitled to attend the meeting (whether physically or using technology) before or during the meeting; and
  - (iii) the meeting is taken to be held at the physical venue set out in the notice of meeting.

# 13A.2 Communication of meeting documents

To the extent permitted under the Corporations Act, Listing Rules and any other applicable law, any document that is required or permitted to be given to a Shareholder that relates to a Shareholders' meeting (including, but not limited to, the notice of meeting) may be distributed:

- (a) by means of electronic communication; or
- (b) by giving the Shareholder (by means of an electronic communication or otherwise) sufficient information to allow the person to access the document electronically,

in accordance with the Corporations Act.

# 14. THE DIRECTORS

# 14.1 Number of Directors

The Company shall at all times have at least 3 Directors. The number of Directors shall not exceed 9. Subject to the Corporations Act, the Company may, by ordinary resolution, increase or reduce the number of Directors and may also determine in what rotation the increased or reduced number is to go out of office. Subject to any resolution of the Company determining the maximum and minimum numbers of Directors, the Directors may from time to time determine the respective number of Executive and non-executive Directors.

# 14.2 Rotation of Directors

Subject to clause 18.4, at the Company's annual general meeting in every year, onethird of the Directors for the time being, or, if their number is not a multiple of 3, then the number nearest one-third (rounded upwards in case of doubt), shall retire from office, provided always that no Director except a Managing Director shall hold office for a period in excess of 3 years, or until the third annual general meeting following his or her appointment, whichever is the longer, without submitting himself for reelection. The Directors to retire at an annual general meeting are those who have been longest in office since their last election, but, as between persons who became Directors on the same day, those to retire shall (unless they otherwise agree among themselves) be determined by drawing lots. A retiring Director is eligible for reelection. An election of Directors shall take place each year.

In determining the number of Directors to retire, no account is to be taken of:

- (a) a Director who only holds office until the next annual general meeting pursuant to clause 14.4; and/ or
- (b) a Managing Director,

each of whom are exempt from retirement by rotation. However, if more than one Managing Director has been appointed by the Directors, only one of them (nominated by the Directors) is entitled to be excluded from any determination of the number of Directors to retire and/or retirement by rotation.

# 14.3 Election of Directors

Subject to the provisions of this Constitution, the Company may elect a person as a Director by resolution passed in general meeting. A Director elected at a general meeting is taken to have been elected with effect immediately after the end of that general meeting unless the resolution by which the Director was appointed or elected specifies a different time. No person other than a Director seeking re-election shall be eligible for election to the office of Director at any general meeting unless the person or some Shareholder intending to propose his or her nomination has, at least 30 Business Days before the meeting, left at the Registered Office a notice in writing duly signed by the nominee giving his or her consent to the nomination and signifying his or her candidature for the office or the intention of the Shareholder to propose the person. Notice of every candidature for election as a Director shall be given to each Shareholder with or as part of the notice of the meeting at which the election is to take place. The Company shall observe the requirements of the Corporations Act with respect to the election of Directors. If the number of nominations exceeds the vacancies available having regard to clause 14.1, the order in which the candidates shall be put up for election shall be determined by the drawing of lots supervised by the Directors and once sufficient candidates have been elected to fill up the vacancies available, the remaining candidates shall be deemed defeated without the need for votes to be taken on their election.

# 14.4 Additional Directors

The Directors may at any time appoint a person to be a Director, either to fill a casual vacancy or as an addition to the existing Directors, but so that the total number of Directors does not at any time exceed the maximum number specified by this Constitution. Any Director so appointed (except a Managing Director, but if there is more than one Managing Director, only one of them is entitled to be excluded from determination of re-election requirements) holds office only until the next following annual general meeting and is then eligible for re-election but shall not be taken into account in determining the Directors who are to retire by rotation (if any) at that meeting.

# 14.5 Removal of Director

The Company may by resolution remove any Director before the expiration of his period of office, and may by resolution appoint another person in his place. The person so appointed is subject to retirement at the same time as if he had become

a Director on the day on which the Director in whose place he is appointed was last elected a Director.

# 14.6 Vacation of Office

The office of Director shall automatically become vacant if the Director:

- (a) ceases to be a Director by virtue of section 203D or any other provision of the Corporations Act;
- (b) becomes bankrupt or insolvent or makes any arrangement or composition with his creditors generally;
- (c) becomes prohibited from being a Director by reason of any order made under the Corporations Act;
- (d) becomes of unsound mind or a person whose person or estate is liable to be dealt with in any way under the law relating to mental health;
- (e) resigns his or her office by notice in writing to the Company;
- (f) is removed from office under clause 14.5; or
- (g) is absent for more than 6 months, without permission of the Directors, from meetings of the Directors held during that period.

## 14.7 Remuneration

The Directors shall be paid out of the funds of the Company, by way of remuneration for their services as Directors. Subject to clause 14.8 below, the total aggregate fixed sum per annum to be paid to the Directors (excluding salaries of executive Directors) from time to time will not exceed the sum determined by the Shareholders in general meeting and the total aggregate fixed sum will be divided between the Directors as the Directors shall determine and, in default of agreement between them, then in equal shares. No non-executive Director shall be paid as part or whole of his remuneration a commission on or a percentage of profits or a commission or a percentage of operating revenue, and no executive Director shall be paid as whole or part of his remuneration a commission on or percentage of operating revenue. The remuneration of a Director shall be deemed to accrue from day to day. Remuneration under this clause 14.7 may be provided in such manner that the Directors decide (including by way of contribution to a superannuation fund on behalf of the Director) and if any part of the fees of any Director is to be provided other than in cash the Directors may determine the manner in which the non-cash component of the fees is to be valued.

# 14.8 Initial Fees to Directors

The total aggregate fixed sum per annum to be paid to Directors (excluding salaries of executive Directors) in accordance with clause 14.7 shall initially be no more than \$500,000 and may be varied by ordinary resolution of the Shareholders in general meeting.

# 14.9 Expenses

The Directors shall be entitled to be paid reasonable travelling, accommodation and other expenses incurred by them respectively in or about the performance of their duties as Directors. If any of the Directors being willing are called upon to perform extra services or make any special exertions on behalf of the Company or its business, the Directors may remunerate this Director in accordance with such services or exertions, and this remuneration may be either in addition to or in substitution for his or her share in the remuneration provided for by clause 14.7.

# 14.10 No Share Qualification

A Director is not required to hold any Shares.

# 15. POWERS AND DUTIES OF DIRECTORS

# 15.1 Management of the Company

Subject to the Corporations Act and the Listing Rules and to any other provision of this Constitution, the business of the Company shall be managed by the Directors, who may pay all expenses incurred in promoting and forming the Company, and may exercise all such powers of the Company as are not, by the Corporations Act or the Listing Rules or by this Constitution, required to be exercised by the Company in general meeting.

# 15.2 Borrowings

Without limiting the generality of clause 15.1, the Directors may at any time:

- (a) exercise all powers of the Company to borrow money, to charge any property or business of the Company or all or any of its uncalled capital and to issue debentures or give any other security for a debt, liability or obligation of the Company or of any other person;
- (b) sell or otherwise dispose of the whole or any part of the assets, undertakings and other properties of the Company or any that may be acquired on such terms and conditions as they may deem advisable, but:
  - (i) if the Company is listed on ASX, the Company shall comply with the Listing Rules which relate to the sale or disposal of a company's assets, undertakings or other properties; and
  - (ii) on the sale or disposition of the Company's main undertaking or on the liquidation of the Company, no commission or fee shall be paid to any Director or Directors or to any liquidator of the Company unless it shall have been ratified by the Company in general meeting, with prior notification of the amount of such proposed payments having been given to all Shareholders at least 7 days prior to the meeting at which any such payment is to be considered; and
- (c) take any action necessary or desirable to enable the Company to comply with the Listing Rules.

# 15.3 Attorneys

The Directors may, by power of attorney, appoint any person or persons to be the attorney or attorneys of the Company for the purposes, with the powers, authorities and discretions (being powers, authorities and discretions vested in or exercisable by the Directors), for the period and subject to the conditions as they think fit. This power of attorney may contain provisions for the protection and convenience of persons dealing with the attorney as the Directors may determine and may also authorise the attorney to delegate all or any of the powers, authorities and discretions vested in the person.

# 15.4 Cheques, etc.

All cheques, promissory notes, bankers drafts, bills of exchange, electronic transfers and other negotiable instruments, and all receipts for money paid to the Company, shall be signed, drawn, accepted, endorsed or otherwise executed, as the case may be, by any two Directors or in any other manner as the Directors determine.

# 15.5 Retirement Benefits for Directors

The Directors may at any time, subject to the Listing Rules, adopt any scheme or plan which they consider to be in the interests of the Company and which is designed to provide retiring or superannuation benefits for both present and future non-executive Directors, and they may from time to time vary this scheme or plan. Any scheme or plan may be effected by agreements entered into by the Company with individual Directors, or by the establishment of a separate trust or fund, or in any other manner the Directors consider proper. The Directors may attach any terms and conditions to any entitlement under any such scheme or plan that they think fit, including, without limitation, a minimum period of service by a Director before the accrual of any entitlement and the acceptance by the Directors of a prescribed retiring age. No scheme or plan shall operate to confer upon any Director or on any of the dependants of any Director any benefits exceeding those contemplated in section 200F of the Corporations Act or the Listing Rules, except with the approval of the Company in general meeting.

# 15.6 Securities to Directors or Shareholders

If a Director acting solely in the capacity of Director of the Company shall become personally liable for the payment of any sum primarily due by the Company, the Directors may create any mortgage, charge or security over or affecting the whole or any part of the assets of the Company by way of indemnity to secure the persons or person so becoming liable from any loss in respect of such liability.

# 16. PROCEEDINGS OF DIRECTORS

# 16.1 Convening a Meeting

A Director may at any time, and a Secretary shall, whenever requested to do so by one or more Directors, convene a meeting of the Directors, but not less than 24 hours' notice of every such meeting shall be given to each Director, and to each Alternate Director, either by personal telephone contact or in writing by the convenor of the meeting. The Directors may by unanimous resolution agree to shorter notice. An accidental omission to send a notice of a meeting of Directors to any Director or the non-receipt of such a notice by any Director does not invalidate the proceedings, or any resolution passed, at the meeting.

# 16.2 Procedure at Meetings

The Directors may meet together for the despatch of business and adjourn and, subject to this clause 16, otherwise regulate the meetings as they think fit.

# 16.3 Quorum

No business shall be transacted at any meeting of Directors unless a quorum is present, comprising a majority of the Board of Directors present in person, or by instantaneous communication device, notwithstanding that less than two Directors may be permitted to vote on any particular resolution or resolutions at that meeting for any reason whatsoever. Where a quorum cannot be established for the consideration of a particular matter at a meeting of Directors, one or more of the Directors may call a general meeting of the Company to deal with the matter. In determining whether a quorum is present, each individual participating as a Director or as an Alternate Director for another Director is to be counted except that an individual participating in more than one capacity is to be counted only once.

## 16.4 Secretary May Attend and Be Heard

The Secretary is entitled to attend any meeting of Directors and is entitled to be heard on any matter dealt with at any meeting of Directors.

# 16.5 Majority Decisions

Questions arising at any meeting of Directors shall be decided by a majority of votes. A resolution passed by a majority of Directors shall for all purposes be deemed a determination of "the Directors". An Alternate Director has one vote for each Director for whom he or she is an alternate. If an Alternate Director is also a Director, he or she also has a vote as a Director.

# 16.6 Casting Votes

In the case of an equality of votes, the chairman of the meeting shall have a second or casting vote, but the chairman shall have no casting vote where only 2 Directors are competent to vote on the question.

# 16.7 Alternate Directors

A Director may, with the approval of a majority of the other Directors, appoint any person to be an alternate Director in his or her place during any period as he or she thinks fit, and the following provisions shall apply with respect to any Alternate Director:

- (a) he or she is entitled to notice of meetings of the Directors and, if his or her appointor Director is not present at such a meeting, he or she is entitled to attend and vote in the place of the absent Director;
- (b) he or she may exercise any powers that his or her appointor Director may exercise (except the power to appoint an Alternate Director), and the exercise of any such power by the alternate Director shall be deemed to be the exercise of the power by his or her appointor Director;
- (c) he or she is subject to the provisions of this Constitution which apply to Directors, except that Alternate Directors are not entitled in that capacity to

any remuneration from the Company (but Alternate Directors are entitled to reasonable travelling, accommodation and other expenses as the Alternate Director may properly incur in travelling to, attending and returning from meetings of Directors or meetings of a committees by the Directors of which the appointor is not present);

- (d) he or she is not required to hold any Shares;
- (e) his or her appointment may be terminated at any time by his or her appointor Director notwithstanding that the period of the appointment of the alternate Director has not expired, and the appointment shall terminate in any event if his or her appointor Director vacates office as a Director;
- (f) the appointment, or the termination of an appointment, of an alternate Director shall be effected by a written notice signed by the Director who made the appointment given to the Company; and
- (g) is, whilst acting as an Alternate Director, an officer of the Company and not the agent of the appointor and is responsible to the exclusion of the appointor for the Alternate Director's own acts and defaults.

# 16.8 Continuing Directors May Act

In the event of a vacancy or vacancies in the office of a Director, the remaining Directors may act but, if the number of remaining Directors is not sufficient to constitute a quorum at a meeting of Directors, they may act only for the purposes of appointing a Director or Directors, or in order to convene a general meeting of the Company.

### 16.9 Chairman

The Directors shall elect from their number a chairman of their meetings and may determine the period for which he or she is to hold office. Where a Directors' meeting is held and a chairman has not been elected or is not present at the meeting within 10 minutes after the time appointed for the meeting to begin, the Directors present shall elect one of their number to be the acting chairman of the meeting. The Directors may elect a Director as deputy chairman to act as chairman in the chairman's absence.

### 16.10 Committees

The Directors may delegate any of their powers to a committee or committees consisting of such of their number as they think fit. The Directors may at any time revoke any such delegation of power. A committee to which any powers have been so delegated shall exercise the powers delegated in accordance with any directions of the Directors, and a power so exercised shall be deemed to have been exercised by the Directors. The members of such a committee may elect one of their number as chairman of their meetings. Questions arising at a meeting of a committee shall be determined by a majority of votes of the members present and voting. In the case of an equality of votes, the chairman shall have a casting vote. The provisions of this Constitution applying to meetings and resolutions of Directors apply, so far as they can and with any necessary changes, to meetings and resolutions of a committee of Directors, except to the extent they are contrary to any direction given under this clause 16.10.

# 16.11 Written Resolutions

A resolution in writing signed by all the Directors for the time being (or their respective alternate Directors), except those Directors (or their alternates) who expressly indicate their abstention in writing to the Company and those who would not be permitted, by virtue of section 195 of the Corporations Act to vote, shall be as valid and effectual as if it had been passed at a meeting of the Directors duly convened and held. This resolution may consist of several documents in like form, each signed by one or more Directors. Copies of the documents to be signed under this clause must be sent to every Director who is entitled to vote on the resolution. The resolution is taken to have been passed when the last Director signs the relevant documents. A facsimile transmission, an email bearing the signature of the Director or an email of the Director addressed to another officer of the Company confirming agreement with the resolution and undertaking to sign the resolution as soon as practicable shall be deemed to be a document in writing signed by the Directors.

# 16.12 Defective Appointment

All acts done by any meeting of the Directors or of a committee of Directors or by any person acting as a Director are, notwithstanding that it is afterwards discovered that there was some defect in the appointment of a person to be, or to act as, a Director, or that a person so appointed was disqualified, as valid as if the person had been duly appointed and was qualified to be a Director or to be a member of the committee.

# 16.13 Directors May Hold Other Offices

A Director may hold any other office or place of profit in or in relation to the Company or a related body corporate of the Company (except that of auditor) in conjunction with his or her office of Director and on any terms as to remuneration or otherwise that the Directors shall approve.

# 16.14 Directors May Hold Shares, etc.

A Director may be or become a shareholder in or director of or hold any other office or place of profit in or in relation to any other company promoted by the Company or a related body corporate of the Company or in which the Company may be interested, whether as a vendor, shareholder or otherwise.

# 16.15 Directors Not Accountable for Benefits

No Director shall be accountable for any benefits received as the holder of any other office or place of profit in or in relation to the Company or any other company referred to in clause 16.14 or as a shareholder in or director of any such company.

# 16.16 Disclosure of Interests in Related Matters

As required by the Corporations Act, a Director must give the Directors notice of any material personal interest in a matter that relates to the affairs of the Company. No Director shall be disqualified by his office from contracting with the Company whether as vendor, purchaser or otherwise, nor shall any such contract or any contract or arrangement entered into by or on behalf of the Company in which any Director shall be in any way interested be avoided or prejudiced on that account, nor shall any Director be liable to account to the Company for any profit arising from any such contract or agreement by reason only of such Director holding that office

or of the fiduciary relationship thereby established. A Director who has a material personal interest in a matter that is being considered at a meeting of Directors must not be present while the matter is being considered at the meeting or vote on that matter except where permitted by the Corporations Act. Nothing in this Constitution shall be read or construed so as to place on a Director any restrictions other than those required by the Corporations Act or the Listing Rules.

# 16.17 Disclosure of Shareholding

A Director must give to the Company such information about the Shares or other securities in the Company in which the Director has a relevant interest and at the times that the Secretary requires, to enable the Company to comply with any disclosure obligations it has under the Corporations Act or the Listing Rules.

# 16.18 Related Body Corporate Contracts

A Director shall not be deemed to be interested or to have been at any time interested in any contract or arrangement by reason only that in a case where the contract or arrangement has been or will be made with, for the benefit of, or on behalf of a Related Body Corporate, he or she is a shareholder in that Related Body Corporate.

# 16.19 Voting, Affixation of Seal

A Director may in all respects act as a Director in relation to any contract or arrangement in which he or she is interested, including, without limiting the generality of the above, in relation to the use of the Company's common seal, but a Director may not vote in relation to any contract or proposed contract or arrangement in which the Director has directly or indirectly a material interest except as permitted by the Corporations Act.

# 16.20 Home Branch to be Advised

The Directors shall advise the Home Branch without delay of any material contract involving Director's or Directors' interests. The advice shall include at least the following information:

- (a) the names of the parties to the contract;
- (b) the name or names of the Director or Directors who has or have any material interest in the contract;
- (c) particulars of the contract; and
- (d) particulars of the relevant Director's or Directors' interest or interests in that contract.

# 17. MEETING BY INSTANTANEOUS COMMUNICATION DEVICE

# 17.1 Meetings to be Effectual

A Director shall be entitled to attend a Directors' meeting by means of an instantaneous communication device rather than in person. In those circumstances, a Director shall still receive all materials and information to be made available for the purposes of the Directors' meeting.

For the purposes of this Constitution, the contemporaneous linking together by instantaneous communication device of a number of consenting Directors not less than the quorum, whether or not any one or more of the Directors is out of Australia, shall be deemed to constitute a Directors' meeting and all the provisions of this Constitution as to the Directors' meetings shall apply to such meetings held by instantaneous communication device so long as the following conditions are met:

- (a) all the directors for the time being entitled to receive notice of the Directors' meeting (including any alternate for any Director) shall be entitled to notice of a meeting by instantaneous communication device for the purposes of such meeting. Notice of any such Directors' meeting shall be given on the instantaneous communication device or in any other manner permitted by this Constitution;
- (b) each of the Directors taking part in the Directors' meeting by instantaneous communication device must be able to hear each of the other Directors taking part at the commencement of the Directors' meeting; and
- (c) at the commencement of the Directors' meeting each Director must acknowledge his or her presence for the purpose of a Directors' meeting of the Company to all the other Directors taking part.

A Directors' meeting held by instantaneous communication device shall be deemed to have been held at the Registered Office.

# 17.2 Procedure at Meetings

A Director may leave a Directors' meeting held under clause 17.1 by informing the Chairman of the Directors' meeting and then disconnecting his instantaneous communication device. Unless this procedure has been followed a Director shall be conclusively presumed to have been present and to have formed part of the quorum at all times during the Directors' meeting by instantaneous communication device.

### 17.3 Minutes

A minute of the proceedings at a meeting held under clause 17.1 shall be sufficient evidence of such proceedings and of the observance of all necessary formalities if certified as a correct minute by the chairman or the person taking the chair at the meeting under clause 17.1.

### 17.4 Definition

For the purposes of this Constitution, "**instantaneous communication device**" shall include telephone, television or any other audio or visual device which permits instantaneous communication.

### 18. MANAGING AND EXECUTIVE DIRECTORS AND SECRETARIES

### 18.1 Appointment

The Directors may from time to time appoint one or more of their number to the office of managing director (**Managing Director**) of the Company or to any other office, (except that of auditor), or employment under the Company, either for a fixed term or at will, but not for life and, subject to the terms of any agreement entered into in a particular case, may revoke any such appointment. A Director other than a Managing Director so appointed is in this Constitution referred to as an executive director (**Executive Director**). The appointment of a Managing Director or Executive Director so appointed automatically terminates if he ceases for any reason to be a Director.

## 18.2 Remuneration

Subject to clause 14.7, a Managing Director or Executive Director shall, subject to the terms of any agreement entered into in a particular case, receive remuneration (whether by way of salary, commission or participation in profits, or partly in one way and partly in another) as the Directors may determine.

## 18.3 Powers

The Directors may, upon such terms and conditions and with such restrictions as they think fit, confer upon a Managing Director or Executive Director any of the powers exercisable by them. Any powers so conferred may be concurrent with, or be to the exclusion of, the powers of the Directors. The Directors may at any time withdraw or vary any of the powers so conferred on a Managing Director or Executive Director.

## 18.4 Rotation

Subject to clause 14.2, a Managing Director shall not retire by rotation, but Executive Directors shall.

## 18.5 Secretary

A Secretary of the Company shall hold office on such terms and conditions, as to remuneration and otherwise, as the Directors determine. There must be at least one Secretary of the Company at all times.

# 19. SEALS

## 19.1 Common Seal

Subject to the Corporations Act, the Company may have a Seal. The Directors shall provide for the safe custody of the Seal. The Seal shall only be used by the authority of the Directors, or of a committee of the Directors authorised by the Directors to authorise the use of the Seal. Every document to which the Seal is affixed shall be signed by a Director and countersigned by another Director, (who may be an alternate Director) a Secretary or another person appointed by the Directors to countersign that document or a class of documents in which that document is included.

## 19.2 Execution of Documents Without a Seal

Without limiting the ways a document can be signed under the Corporations Act, the Company may execute a document without using the Seal if the document is signed by:

- (a) two Directors;
- (b) a Director and a Secretary; or

(c) any person or persons authorised by the Directors for the purposes of executing that document or the class of document to which that document belongs.

# 19.3 Share Seal

Subject to the Corporations Act, the Company may have a duplicate Seal, known as the Share Seal, which shall be a facsimile of the Seal with the addition on its face of the words "**Share Seal**", and the following provisions shall apply to its use:

- (a) any certificate for Shares may be issued under the Share Seal and if so issued shall be deemed to be sealed with the Seal;
- (b) subject to the following provisions of this clause 19.3, the signatures required by clause 19.1 on a document to which the Seal is affixed may be imposed by some mechanical means;
- (c) subject to the following provisions of this clause 19.3, the Directors may determine the manner in which the Share Seal shall be affixed to any document and by whom a document to which the Share Seal is affixed shall be signed, and whether any signature so required on such a document must be actually written on the document or whether it may be imposed by some mechanical means;
- (d) the only documents on which the Share Seal may be used shall be Share or stock unit certificates, debentures or certificates of debenture stock, secured or unsecured notes, option certificates and any certificates or other documents evidencing any Share Options or rights to take up any Shares in or debenture stock or debentures or notes of the Company; and
- (e) signatures shall not be imposed by mechanical means nor (except when the requirements of clause 19.1 as to signatures are complied with) shall the Share Seal be used on any certificate or other document mentioned in clause 19.3(d) unless the certificate or other document has first been approved for sealing or signature (as the case may be) by the Board or other authorised person or persons.

# 20. ACCOUNTS, AUDIT AND RECORDS

### 20.1 Accounting records to be kept

The Directors shall cause proper accounting and other records to be kept by the Company and shall distribute copies of the Company's accounts and reports as required by the Corporations Act and the Listing Rules.

### 20.2 Audit

The Company shall comply with the requirements of the Corporations Act and the Listing Rules as to the audit of accounts, registers and records.

### 20.3 Inspection

The Directors shall determine whether and to what extent, and at what time and places and under what conditions, the accounting records and other documents of the Company or any of them will be open to the inspection of Shareholders other than Directors. A Shareholder other than a Director shall not be entitled to inspect any document of the Company except as provided by law or authorised by the Directors or by the Company in general meeting.

# 21. MINUTES

### 21.1 Minutes to be Kept

The Directors shall cause to be kept, in accordance with section 1306 of the Corporations Act, minutes of:

- (a) all proceedings of general meetings and Directors meetings; and
- (b) all appointments of Officers and persons ceasing to be Officers.

### 21.2 Signature of Minutes

All minutes shall be signed by the chairman of the meeting at which the proceedings took place or by the chairman of the next succeeding meeting.

### 21.3 Requirements of the Corporations Act

The Company and the Officers shall comply with the requirements of Part 2G.3 of Chapter 2G of the Corporations Act.

### 22. DIVIDENDS AND RESERVES

### 22.1 Dividends

Subject to and in accordance with the Corporations Act, the Listing Rules, the rights of any preference Shareholders and to the rights of the holders of any shares created or raised under any special arrangement as to dividend, the Directors may from time to time decide to pay a dividend to the Shareholders entitled to the dividend which shall be payable on all Shares according to the proportion that the amount paid (not credited) is of the total amounts paid and payable (excluding amounts credited) in respect of such Shares. The Directors may rescind a decision to pay a dividend if they decide, before the payment date, that the Company's financial position no longer justifies the payment.

### 22.2 Interim Dividend

The Directors may from time to time pay to the Shareholders any interim dividends that they may determine.

# 22.3 No Interest

No dividend shall carry interest as against the Company.

## 22.4 Reserves

The Directors may set aside out of the profits of the Company any amounts that they may determine as reserves, to be applied at the discretion of the Directors, for any purpose for which the profits of the Company may be properly applied. Pending any application of the reserves, the Directors may invest or use the reserves in the business of the Company or in other investments as they think fit. Any amount set aside as a

reserve is not required to be held separately from the Company's other assets and may be used by the Company or invested as the Directors think fit.

# 22.5 Carrying forward profits

The Directors may carry forward any part of the profits of the Company that it decides not to distribute as dividends without transferring those profits to a reserve.

# 22.6 Alternative Method of Payment of Dividend

When declaring any dividend and subject at all times to the Corporations Act and the Listing Rules, the Directors may:

- (a) direct payment of the dividend to be made wholly or in part by the distribution of specific assets or documents of title (including, without limitation, paid-up Shares, debentures or debenture stock of this or any other company, gold, gold or mint certificates or receipts and like documents) or in any one or more of these ways, and where any difficulty arises with regard to the distribution the Directors may settle it as they think expedient and in particular may issue fractional certificates and may fix the value for distribution of specific assets or any part of them and may determine that cash payments shall be made to any Shareholders upon the basis of the value so fixed in order to adjust the rights of all parties and may vest any of these specific assets in trustees upon trusts for the persons entitled to the dividend as may seem expedient to the Directors; or
- (b) direct that a dividend be payable to particular Shareholders wholly or partly out of any particular fund or reserve or out of profits derived from any particular source and to the remaining Shareholders wholly or partly or of any other particular fund or reserve or out of profits derived from any other particular source and may so direct notwithstanding that by so doing the dividend will form part of the assessable income for taxation purposes of some Shareholders and will not form part of the assessable income of others.

For the purposes of this clause, the Company is authorised to distribute securities of another body corporate by way of dividend and, on behalf of the Shareholders, provide the consent of each Shareholder to becoming a member of that body corporate and the agreement of each Shareholder to being bound by the constitution of that body corporate.

# 22.7 Shareholders entitled to dividend

Subject to this Constitution, a dividend in respect of a Share is payable to the person registered as the holder of that share:

- (a) if the Directors have fixed a time for determining entitlements to the dividend, at that time; and
- (b) in any other case, on the date on which the dividend is paid.

# 22.8 Payment of Dividends

Any dividend payable may be paid by:

(a) cheque sent through the mail directed to:

- (i) the address of the Shareholder shown in the Register or to the address of the joint holders of Shares shown first in the Register; or
- (ii) an address which the Shareholder has, or joint holders have, in writing notified the Company as the address to which dividends should be sent;
- (b) electronic funds transfer to an account with a bank or other financial institution nominated by the Shareholder and acceptable to the Company; or
- (c) any other means determined by the Directors.

# 22.9 Unclaimed Dividends

Except as otherwise provided by statute, all dividends unclaimed for one year after having been declared may be invested or otherwise made use of by the Directors for the benefit of the Company until claimed.

## 22.10 Breach of Restriction Agreement

In the event of a breach of the Listing Rules relating to Restricted Securities or of any escrow arrangement entered into by the Company under the Listing Rules in relation to any Shares which are classified under the Listing Rules or by ASX as Restricted Securities, the Shareholder holding the Shares in question shall cease to be entitled to be paid any dividends in respect of those Shares for so long as the breach subsists.

# 23. CAPITALISATION OF PROFITS

### 23.1 Capitalisation

The Directors, subject to the Listing Rules and any rights or restrictions for the time being attached to any class of Shares, may from time to time resolve to capitalise any amount, being the whole or a part of the amount for the time being standing to the credit of any reserve account or the profit and loss account or otherwise available for distribution to Shareholders, and that that amount be applied, in any of the ways mentioned in clause 23.2 for the benefit of Shareholders in the proportions to which those Shareholders would have been entitled in a distribution of that amount by way of dividend.

# 23.2 Application of Capitalised Amounts

The ways in which an amount may be applied for the benefit of Shareholders under clause 23.1 are:

- (a) in paying up any amounts unpaid on Shares held by Shareholders;
- (b) in paying up in full, at an issue price decided by Director's resolution, unissued Shares or debentures to be issued to Shareholders as fully paid; or
- (c) partly as mentioned in paragraph (a) and partly as mentioned in paragraph (b).

# 23.3 Procedures

The Directors shall do all things necessary to give effect to the resolution referred to in clause 23.1 and, in particular, to the extent necessary to adjust the rights of the Shareholders among themselves, may:

- (a) issue fractional certificates or make cash payments in cases where Shares or debentures could only be issued in fractions; and
- (b) authorise any person to make, on behalf of all the Shareholders entitled to any further Shares or debentures upon the capitalisation, an agreement with the Company providing for the issue to them, credited as fully paid up, of any further Shares or debentures or for the payment up by the Company on their behalf of the amounts or any part of the amounts remaining unpaid on their existing Shares by the application of their respective proportions of the sum resolved to be capitalised,

and any agreement made under an authority referred to in paragraph (b) is effective and binding on all the Shareholders concerned.

# 24. BONUS SHARE PLAN

# 24.1 Authorisation of Bonus Share Plan

Subject to the Listing Rules and the Corporations Act, the Company may, by ordinary resolution in general meeting, authorise the Directors to implement a Bonus Share Plan on such terms and conditions as are referred to in the resolution and which plan provides for any dividend which the Directors may declare from time to time under clause 22, less any amount which the Company shall either pursuant to this Constitution or any law be entitled or obliged to retain, not to be payable on Shares which are participating Shares in the Bonus Share Plan but for those Shares to carry instead an entitlement to receive an allotment of additional fully paid ordinary Shares to be issued as bonus Shares.

# 24.2 Amendment and Revocation

Any resolution passed by the Company in general meeting pursuant to clause 24.1 may, at any time, be amended or revoked by the Company by ordinary resolution in general meeting.

# 25. DIVIDEND REINVESTMENT PLAN

# 25.1 Authorisation of Dividend Reinvestment Plan

Subject to the Listing Rules and the Corporations Act, the Company may, by resolution of the Directors, implement a Dividend Reinvestment Plan on such terms and conditions as are referred to in the resolution and which plan provides for any dividend which the Directors may declare from time to time under clause 22 and payable on Shares which are participating Shares in the Dividend Reinvestment Plan, less any amount which the Company shall either pursuant to this Constitution or any law be entitled or obliged to retain, to be applied by the Company to the payment of the subscription price of ordinary fully paid Shares.

# 25.2 Amendment and Revocation

Any resolution passed by the Directors pursuant to clause 25.1 may, at any time, be amended or revoked by the Company by ordinary resolution in general meeting.

# 26. NOTICES

# 26.1 Service by the Company to Shareholders

A notice may be given by the Company to any Shareholder either by:

- (a) serving it on him or her personally; or
- (b) sending it by post to the Shareholder at his or her address as shown in the Register of Shareholders or the address supplied by the Shareholder to the Company for the giving of notices to this person. Notices to Shareholders whose registered address is outside Australia shall be sent by airmail or, where applicable, by the means provided for by clause 26.9; or
- (c) sending it by fax or other electronic means (including providing a URL link to any document or attachment) to the fax number or electronic address nominated by the Shareholder for giving notices.

## 26.2 Service of notices by the Company to Directors

A notice may be given by the Company to a Director or Alternate Director by:

- (a) serving it on him or her personally;
- (b) sending it by ordinary post to his or her usual residential or business address, or any other address he or she has supplied to the Company for giving notices;
- (c) sending it by fax or other electronic means (including providing a URL link to any document or attachment) to the fax number or electronic address he or she has supplied to the Company for giving notices.

# 26.3 Service of notices by Directors, Alternate Directors and Shareholders to the Company

Without limiting any other way that a communication may be given under the Corporations Act, a notice may be given by a Director or Alternate Director or a Shareholder to the Company by:

- (a) delivering it to the Company's registered office;
- (b) sending it by ordinary post to the Company's registered office;
- (c) sending it by fax or other electronic means to the principal fax number or electronic address at the Company's registered office.

### 26.4 Deemed receipt of Notice

A notice will be deemed to be received by a Shareholder when:

(a) where a notice is served personally, service of the notice shall be deemed

to be effected when hand delivered to the member in person;

- (b) where a notice is sent by post, service of the notice shall be deemed to be effected by properly addressing, pre-paying and posting a letter containing the notice, and to have been effected, in the case of a notice of a meeting, on the date after the date of its posting and, in any other case, at the time at which the letter would be delivered in the ordinary course of post;
- (c) where a notice is sent by facsimile, service of the notice shall be deemed to be effected upon confirmation being received by the Company that all pages of the notice have been successfully transmitted to the Shareholder's facsimile machine at the facsimile number nominated by the Shareholder; and
- (d) where a notice is sent to an electronic address by electronic means, service of the notice shall be deemed to be effected once sent by the Company to the electronic address nominated by the Shareholder (regardless of whether or not the notice is actually received by the Shareholder).

# 26.5 Notice to Joint Holders

A notice may be given by the Company to the joint holders of a Share by giving the notice to the joint holder first named in the Register of Shareholders in respect of the Share.

# 26.6 Notices to Personal Representatives and Others

A notice may be given by the Company to a person entitled to a Share in consequence of the death or bankruptcy of a Shareholder by serving it on him or her or by sending it to him or her by post addressed to the person by name or by the title or representative of the deceased or assignee of the bankrupt, or by any like description, at the address (if any) supplied for the purpose by the person or, if such an address has not been supplied, at the address to which the notice might have been sent if the death or bankruptcy had not occurred.

# 26.7 Persons Entitled to Notice

Notice of every general meeting shall be given to each person who at the time of giving the notice is:

- (a) a Shareholder;
- (b) a person entitled to a Share in consequence of the death or bankruptcy of a Shareholder who, but for his death or bankruptcy, would be entitled to receive notice of the meeting;
- (c) a Director or Alternate Director;
- (d) the auditor for the time being of the Company; and
- (e) if the Company has issued and there are currently any Listed Securities, the Home Branch,

unless that person waives the right to receive notice by written notice to the Company. No other person is entitled to receive notices of general meetings.

# 26.8 Change of Address

The Company shall acknowledge receipt of all notifications of change of address by Shareholders.

## 26.9 Incorrect Address

Where the Company has bona fide reason to believe that a Shareholder is not known at his or her registered address, and the Company has subsequently made an enquiry in writing at that address as to the whereabouts of the Shareholder and this enquiry either elicits no response or a response indicating that the Shareholder or his present whereabouts are unknown, all future notices will be deemed to be given to the Shareholder if the notice is exhibited in the Registered Office (or, in the case of a Shareholder registered on a Branch Register, in a conspicuous place in the place where the Branch Register is kept) for a period of 48 hours (and shall be deemed to be duly served at the commencement of that period) unless and until the Shareholder informs the Company of a new address to which the Company may send him notices (which new address shall be deemed his registered address).

# 27. WINDING UP

# 27.1 Distribution in Kind

If the Company is wound up, the liquidator may, with the authority of a special resolution, divide among the Shareholders in kind the whole or any part of the property of the Company, and may for that purpose set a value as the liquidator considers fair upon any property to be so decided, and may determine how the division is to be carried out as between the Shareholders or different classes of Shareholders. No member is obliged to accept any Shares, securities or other assets in respect of which there is any liability.

# 27.2 Trust for Shareholders

The liquidator may, with the authority of a special resolution, vest the whole or any part of any property in trustees upon such trusts for the benefit of the contributories as the liquidator thinks fit, but so that no Shareholder is compelled to accept any Shares or other securities in respect of which there is any liability.

# 27.3 Distribution in Proportion to Shares Held

Subject to the rights of Shareholders (if any) entitled to Shares with special rights in a winding-up and the Corporations Act all monies and property that are to be distributed among Shareholders on a winding-up, shall be distributed in proportion to the Shares held by them respectively, irrespective of the amount paid-up or credited as paid-up on the Shares.

# 28. INDEMNITIES AND INSURANCE

# 28.1 Liability to Third Parties

To the extent permitted by law, the Company:

(a) indemnifies and agrees to keep indemnified every Director, executive officer or Secretary of the Company; and

(b) may, by deed, indemnify or agree to indemnify an officer (other than a Director, executive officer or Secretary) of the Company,

against a liability to another person, other than the Company or a related body corporate of the Company, PROVIDED THAT:

- (c) the provisions of the Corporations Act (including, but not limited to, Chapter 2E) are complied with in relation to the giving of the indemnity; and
- (d) the liability does not arise in respect of conduct involving a lack of good faith on the part of the officer.

# 28.2 Defending Proceedings

To the extent permitted by law, the Company:

- (a) hereby indemnifies and agrees to keep indemnified every Director, executive officer and Secretary of the Company; and
- (b) may, by deed, indemnify or agree to indemnify an officer of the Company (other than a director, executive officer or secretary);

out of the property of the Company in relation to the period during which that officer held his or her office against a liability for costs and expenses incurred by that officer in that capacity:

- (c) in defending proceedings, whether civil or criminal, in which:
  - (i) judgment is given in favour of that officer; or
  - (ii) that officer is acquitted; or
- (d) in connection with an application in relation to any proceedings referred to in clause 28.2(c) in which relief is granted to that officer by the Court under the Corporations Act.

### 28.3 Insurance

To the extent permitted by law, the Company or a related body corporate of the Company may pay, or agree to pay, a premium under a contract insuring an officer in relation to the period during which that officer held that office, including in respect of a liability for costs and expenses incurred by a person in defending civil or criminal proceedings whether or not the officer has successfully defended himself or herself in these proceedings, provided that:

- (a) the provisions of the Corporations Act (including, but not limited to, Chapter 2E) are complied with in relation to the payment of the premium; and
- (b) the liability does not arise out of conduct involving a wilful breach of duty to the Company or a contravention of sections 182 or 183 of the Corporations Act.

# 28.4 Disclosure

Subject to any exception provided for in the Corporations Act, full particulars of the Company's indemnities and insurance premiums in relation to the officers must be included each year in the Directors' Report.

# 28.5 Definition

For the purposes of this clause 28, "officer" means:

- (a) a Director, Secretary or executive officer of the Company, whether past, present or future by whatever name called and whether or not validly appointed to occupy or duly authorised to act in such a position; and
- (b) any person who by virtue of any applicable legislation or law is deemed to be a Director or officer of the Company, including without limitation, the persons defined as an officer of a company by section 9 of the Corporations Act.

Nothing in this clause 28 precludes the Company from indemnifying employees (other than officers) and consultants or sub-contractors where the Directors consider it is necessary or appropriate in the exercise of their powers to manage the Company.

# 29. DIRECTORS' ACCESS TO INFORMATION

Where the Directors consider it appropriate, the Company may:

- (a) give a former Director access to certain papers, including documents provided or available to the Directors and other papers referred to in those documents; and
- (b) bind itself in any contract with a Director or former Director to give the access.

# **30. OVERSEAS SHAREHOLDERS**

Each Shareholder with a registered address outside Australia acknowledges that, with the approval of the Home Branch, the Company may, as contemplated by the Listing Rules, arrange for a nominee to dispose of any of its entitlement to participate in any issue of Shares or Share Options by the Company to Shareholders.

# 31. LOCAL MANAGEMENT

# 31.1 Local Management

The Directors may from time to time provide for the management and transaction of the affairs of the Company in any specified locality whether in or outside the State in such manner as it thinks fit and the provisions contained in clauses 31.2, 31.3 and 31.4 shall be without prejudice to the general powers conferred by this clause 31.1.

# 31.2 Local Boards or Agencies

The Directors may at any time and from time to time establish any local boards or agencies for managing any of the affairs of the Company in any specified locality and appoint any persons to be Shareholders of a local board or any managers or agents and may fix their remuneration. The Directors may from time to time and at any time delegate to any person so appointed any of the powers, authorities and discretions for the time being vested in the Directors other than the power of making calls and may authorise the Shareholders for the time being of any local board or any of them to fill up any vacancies on a local board and to act notwithstanding vacancies. This appointment or delegation may be made on the terms and subject to the conditions that the Directors think fit and the Directors may at any time remove any person so appointed and may annul or vary any or all of this delegation.

# 31.3 Appointment of Attorneys

The Company may at any time and from time to time by power of attorney appoint any person or persons to be the attorney or attorneys of the Company for purposes and with powers, authorities and discretions (not exceeding those vested in or exercisable by the Company) and for the period and subject to the conditions that the Company may from time to time think fit. This appointment may (if the Company thinks fit) be made in favour of the Shareholders or any of the Shareholders of any local board established under clause 31.2 or in favour of any company or of the Shareholders, directors, nominees or managers of any company or firm or in favour of any fluctuating body of persons whether or not nominated directly by the Company. The power of attorney may contain any provisions for the protection or convenience of persons dealing with such attorney or attorneys that the Company thinks fit.

# 31.4 Authority of Attorneys

Any such delegates or attorneys as appointed under this Constitution may be authorised by the Company to sub-delegate all or any of the powers, authorities and discretions for the time being vested in them.

# 32. DISCOVERY

Save as provided by the Corporations Act or the Listing Rules no Shareholder shall be entitled to require discovery of any information in respect of any details of the Company's trading or any matter which is or may be in the nature of a trade secret, mystery of trade or technical process which may relate to the business of the Company and which in the opinion of the Directors it would be expedient in the interests of the Shareholders of the Company to communicate.

# 33. COMPLIANCE (OR INCONSISTENCY) WITH THE LISTING RULES

- (a) In this Constitution, a reference to the Listing Rules is to have effect if, and only if, at the relevant time, the Company has been admitted to and remains on the Official List and is otherwise to be disregarded.
- (b) If the Company is admitted to the Official List, the following clauses apply:

- (i) notwithstanding anything contained in this Constitution, if the Listing Rules prohibit an act being done, the act shall not be done;
- (ii) nothing contained in this Constitution prevents an act being done that the Listing Rules require to be done;
- (iii) if the Listing Rules require an act to be done or not to be done, authority is given for that act to be done or not to be done (as the case may be);
- (iv) if the Listing Rules require this Constitution to contain a provision and it does not contain such a provision, this Constitution is deemed to contain that provision;
- (v) if the Listing Rules require this Constitution not to contain a provision and it contains such a provision, this Constitution is deemed not to contain that provision; and
- (vi) if any provision of this Constitution is or becomes inconsistent with the Listing Rules, this Constitution is deemed not to contain that provision to the extent of inconsistency.

# 34. CONSISTENCY WITH CHAPTER 2E OF THE CORPORATIONS ACT

## 34.1 Requirements of Chapter 2E

Notwithstanding any other provision to the contrary contained in this Constitution:

- (a) the Company shall not give a financial benefit to a related party except as permitted by Chapter 2E of the Corporations Act;
- (b) all notices convening general meetings for the purposes of section 208 of the Corporations Act shall comply with the requirements of sections 217 to 227 of the Corporations Act;
- (c) all meetings convened pursuant to section 221 shall be held in accordance with the requirements of section 225 of the Corporations Act; and
- (d) no holder of Shares or person on their behalf shall be entitled to vote or vote on a proposed resolution under Part 2E.1 of the Corporations Act if that holder of Shares is a related party of the public company to whom the resolution would permit a financial benefit to be given or an associate of such a related party.

# 34.2 Definitions

For the purposes of this clause 34 the terms:

- (a) "financial benefit" and "related party" shall have the meanings given or indicated by Part 2E.1 and Part 2E.2 of the Corporations Act; and
- (b) "**associate**" shall have the meaning given to it in Division 2 of Part 1.2 of the Corporations Act.

# 35. INADVERTENT OMISSIONS

If some formality required by this Constitution is inadvertently omitted or is not carried out the omission does not invalidate any resolution, act, matter or thing which but for the omission would have been valid unless it is proved to the satisfaction of the Directors that the omission has directly prejudiced any Shareholder financially. The decision of the Directors is final and binding on all Shareholders.

# 36. TRANSITIONAL

# 36.1 Provisions Relating to Official Quotation of Securities

Subject to clause 36.2 the provisions of this Constitution which relate to the official quotation of the Company's securities on ASX (**Official Quotation**), including but not limited to clauses which refer to ASX, the Listing Rules, the ASX Settlement Operating Rules, the Home Exchange, CHESS, Restricted Securities or Listed Securities shall not have effect except while the Company is admitted to the official list of entities that ASX has admitted and not removed.

## 36.2 Severance

To the extent that any of the provisions of this Constitution referred to in clause 36.1 above can continue to have effect following severance of the matters relating to Official Quotation, then such provisions shall be valid and effectual, notwithstanding clause 36.1, as from the date of adoption of this Constitution by special resolution of the Shareholders of the Company.

# SCHEDULE 1 - PREFERENCE SHARES (CLAUSE 2.6)

## 1. In this schedule, unless the context otherwise requires:

**Dividend Date** means, in relation to a Preference Share, a date specified in the Issue Resolution on which a dividend in respect of that Preference Share is payable.

**Dividend Rate** means, in relation to a Preference Share, the term specified in the Issue Resolution for the calculation of the amount of dividend to be paid in respect of that Preference Share on any Dividend Date, which calculation may be wholly or partly established by reference to an algebraic formula.

**Franked Dividend** has the same meaning ascribed to Franked Distribution in Part 3-6 of the Tax Act.

**Issue Resolution** means the resolution specified in clause 4 of this schedule.

Preference Share means a preference share issued under clause 2.6.

**Redeemable Preference Share** means a Preference Share which the Issue Resolution specified as being, or being at the option of the Company to be, liable to be redeemed.

**Redemption Amount** means, in relation to a Redeemable Preference Share, the amount specified to be paid on redemption of the Redeemable Preference Share.

**Redemption Date** means, in relation to a Redeemable Preference Share, the date specified in the Issue Resolution for the redemption of that Preference Share.

Tax Act means the Income Tax Assessment Act 1997.

- 2. Each Preference Share confers upon its holder:
  - (a) the right in a winding up to payment in cash of the capital (including any premium) then paid up on it, and any arrears of dividend in respect of that Preference Share, in priority to any other class of Shares;
  - (b) the right in priority to any payment of dividend to any other class of Shares to a cumulative preferential dividend payable on each Dividend Date in relation to that Preference Share calculated in accordance with the Dividend Rate in relation to that Preference Share; and
  - (c) no right to participate beyond the extent elsewhere specified in clause 2 of this schedule in surplus assets or profits of the Company, whether in a winding up or otherwise.
- 3. Each Preference Share also confers upon its holder the same rights as the holders of ordinary Shares to receive notices, reports, audited accounts and balance sheets of the Company and to attend general meetings and confers upon its holder the right to vote at any general meeting of the Company in each of the following circumstances and in no others:
  - (a) during a period during which a dividend (or part of a dividend) in respect of the Preference Share is in arrears;

- (b) on a proposal to reduce the Company's share capital;
- (c) on a resolution to approve the terms of a buy-back agreement;
- (d) on a proposal that affects rights attached to the Preference Share;
- (e) on a proposal to wind up the Company;
- (f) on a proposal for the disposal of the whole of the Company's property, business and undertaking;
- (g) during the winding up of the Company; and
- (h) in any other circumstances in which the Listing Rules require holders of preference shares to vote.
- 4. The Board may only allot a Preference Share where by resolution it specifies the Dividend Date, the Dividend Rate, and whether the Preference Share is or is not, or at the option of the Company is to be, liable to be redeemed, and, if the Preference Share is a Redeemable Preference Share, the Redemption Amount and Redemption Date for that Redeemable Preference Share and any other terms and conditions to apply to that Preference Share.
- 5. The Issue Resolution in establishing the Dividend Rate for a Preference Share may specify that the dividend is to be one of:
  - (a) fixed;
  - (b) variable depending upon any variation of the respective values of any factors in an algebraic formula specified in the Issue Resolution; or
  - (c) variable depending upon such other factors as the Board may specify in the Issue Resolution,

and may also specify that the dividend is to be a Franked Dividend or not a Franked Dividend.

- 6. Where the Issue Resolution specifies that the dividend to be paid in respect of the Preference Share is to be a Franked Dividend the Issue Resolution may also specify:
  - (a) the extent to which such dividend is to be franked (within the meaning of the Tax Act); and
  - (b) the consequences of any dividend paid not being so franked, which may include a provision for an increase in the amount of the dividend to such an extent or by reference to such factors as may be specified in the Issue Resolution.
- 7. Subject to the Corporations Act, the Company must redeem a Redeemable Preference Share on issue:
  - (a) on the specified date where the Company, at least 15 Business Days before that date, has given a notice to the holder of that Redeemable Preference Share stating that the Redeemable Preference Share will be so redeemed on the specified date; and

(b) in any event, on the Redemption Date,

but no Redeemable Preference Share may be redeemed and no notice of redemption may be given before the date set by the Directors (if any) upon which that Redeemable Preference Share is issued.

- 8. The certificate issued by the Company in relation to any Preference Share must specify in relation to that Preference Share:
  - (a) the date of issue of the Preference Share;
  - (b) the Dividend Rate and Dividend Dates;
  - (c) whether the Preference Share is a Redeemable Preference Share and if it is:
    - (i) the Redemption Amount and Redemption Date; and
    - (ii) the conditions of redemption (if any);
  - (d) the conditions of participation (if any) in respect of the Preference Share set out in clause 3 of this schedule; and
  - (e) any other matter the Board determines.
- 9. On redemption of a Redeemable Preference Share, the Company, after the holder has surrendered to the Company the certificate in respect of that Redeemable Preference Share, must pay to the holder the Redemption Amount in cash, by cheque or in any other form that the holder agrees to in writing.



#### Cyprium Metals Limited | ABN 48 002 678 640

Your proxy voting instruction must be received by **10.00am (AWST) on Wednesday, 29 January 2025**, being **not later than 48 hours** before the commencement of the Meeting. Any Proxy Voting instructions received after that time will not be valid for the scheduled Meeting.

#### SUBMIT YOUR PROXY

#### Complete the form overleaf in accordance with the instructions set out below.

#### YOUR NAME AND ADDRESS

The name and address shown above is as it appears on the Company's share register. If this information is incorrect, and you have an Issuer Sponsored holding, you can update your address through the investor portal: https://investor.automic.com.au/#/home Shareholders sponsored by a broker should advise their broker of any changes.

#### STEP 1 – APPOINT A PROXY

If you wish to appoint someone other than the Chair of the Meeting as your proxy, please write the name of that Individual or body corporate. A proxy need not be a Shareholder of the Company. Otherwise if you leave this box blank, the Chair of the Meeting will be appointed as your proxy by default.

#### DEFAULT TO THE CHAIR OF THE MEETING

Any directed proxies that are not voted on a poll at the Meeting will default to the Chair of the Meeting, who is required to vote these proxies as directed. Any undirected proxies that default to the Chair of the Meeting will be voted according to the instructions set out in this Proxy Voting Form, including where the Resolutions are connected directly or indirectly with the remuneration of Key Management Personnel.

#### **STEP 2 - VOTES ON ITEMS OF BUSINESS**

You may direct your proxy how to vote by marking one of the boxes opposite each item of business. All your shares will be voted in accordance with such a direction unless you indicate only a portion of voting rights are to be voted on any item by inserting the percentage or number of shares you wish to vote in the appropriate box or boxes. If you do not mark any of the boxes on the items of business, your proxy may vote as he or she chooses. If you mark more than one box on an item your vote on that item will be invalid.

#### APPOINTMENT OF SECOND PROXY

You may appoint up to two proxies. If you appoint two proxies, you should complete two separate Proxy Voting Forms and specify the percentage or number each proxy may exercise. If you do not specify a percentage or number, each proxy may exercise half the votes. You must return both Proxy Voting Forms together. If you require an additional Proxy Voting Form, contact Automic Registry Services.

#### SIGNING INSTRUCTIONS

**Individual:** Where the holding is in one name, the Shareholder must sign.

Joint holding: Where the holding is in more than one name, all Shareholders should sign.

**Power of attorney:** If you have not already lodged the power of attorney with the registry, please attach a certified photocopy of the power of attorney to this Proxy Voting Form when you return it.

**Companies:** To be signed in accordance with your Constitution. Please sign in the appropriate box which indicates the office held by you.

Email Address: Please provide your email address in the space provided.

By providing your email address, you elect to receive all communications despatched by the Company electronically (where legally permissible) such as a Notice of Meeting, Proxy Voting Form and Annual Report via email.

#### **CORPORATE REPRESENTATIVES**

If a representative of the corporation is to attend the Meeting the appropriate 'Appointment of Corporate Representative' should be produced prior to admission. A form may be obtained from the Company's share registry online at https://automicgroup.com.au.

#### Lodging your Proxy Voting Form:

If you are attending the Meeting in person, please bring this with you for Securityholder registration.

#### Online

**Proxy Votina F** 

Use your computer or smartphone to appoint a proxy at

https://investor.automic.com.au/#/loginsah or scan the QR code below using your smartphone

Login & Click on 'Meetings'. Use the Holder Number as shown at the top of this Proxy Voting Form.



#### BY MAIL:

Automic GPO Box 5193 Sydney NSW 2001

#### IN PERSON:

Automic Level 5, 126 Phillip Street Sydney NSW 2000

BY EMAIL:

meetings@automicgroup.com.au

**BY FACSIMILE:** +61 2 8583 3040

All enquiries to Automic: WEBSITE:

https://automicgroup.com.au

#### PHONE:

1300 288 664 (Within Australia) +61 2 9698 5414 (Overseas)

#### STEP 1 - How to vote

Sole Director and Sole Company Secretary

Contact Name:

Email Address:

Contact Daytime Telephone

#### **APPOINT A PROXY:**

I/We being a Shareholder entitled to attend and vote at the Extraordinary General Meeting of Cyprium Metals Limited, to be held at 10.00am (AWST) on Friday, 31 January 2025 at Vibe Hotel 9 Alvan St, Subiaco WA 6008 hereby:

Appoint the Chair of the Meeting (Chair) OR if you are not appointing the Chair of the Meeting as your proxy, please write in the box provided below the name of the person or body corporate you are appointing as your proxy or failing the person so named or, if no person is named, the Chair, or the Chair's nominee, to vote in accordance with the following directions, or, if no directions have been given, and subject to the relevant laws as the proxy sees fit and at any adjournment thereof.

|  |  |  |  | 1 1 |  |  |  |  | 1 |  |  |  |  |
|--|--|--|--|-----|--|--|--|--|---|--|--|--|--|
|  |  |  |  |     |  |  |  |  |   |  |  |  |  |

#### The Chair intends to vote undirected proxies in favour of all Resolutions in which the Chair is entitled to vote.

Unless indicated otherwise by ticking the "for", "against" or "abstain" box you will be authorising the Chair to vote in accordance with the Chair's voting intention.

#### AUTHORITY FOR CHAIR TO VOTE UNDIRECTED PROXIES ON REMUNERATION RELATED RESOLUTIONS

Where I/we have appointed the Chair as my/our proxy (or where the Chair becomes my/our proxy by default), I/we expressly authorise the Chair to exercise my/our proxy on Resolutions 7, 8, 9, 10, 11 and 12 (except where I/we have indicated a different voting intention below) even though Resolutions 7, 8, 9, 10, 11 and 12 are connected directly or indirectly with the remuneration of a member of the Key Management Personnel, which includes the Chair.

|   | STEP 2 - Your voting direction  |      |                 |                             |  |        |           |           |
|---|---|------|-----------------|-----------------------------|--|--------|-----------|-----------|
|   | Resolutions   | For  | Against Abstain | Resolutions                 |  | For    | Against   | t Abstair |
|   | RATIFICATION OF PRIOR ISSUE OF<br>TRANCHE 1 PLACEMENT SHARES  |      |                 | BENEFITS TO<br>(EXECUTIVE O | OF POTENTIAL TERMINATION<br>MR J MATTHEW FIFIELD<br>CHAIR) UNDER EMPLOYEE<br>NCENTIVE PLAN |        |           |           |
| • | 2 RATIFICATION OF PRIOR ISSUE OF<br>TRANCHE 1 PLACEMENT OPTIONS   |      |                 | SACRIFICE O                 | PF EXECUTIVE CHAIR SALARY<br>FFER (FY25) TO MR J<br>FIELD UNDER EMPLOYEE<br>NCENTIVE PLAN  |        |           |           |
|   | APPROVAL FOR THE ISSUE OF TRANCHE 2<br>CORNERSTONE SECURITIES UNDER<br>LISTING RULE 7.1   |      |                 | OFFER (FY25                 | OF NED SALARY SACRIFICE<br>) TO MR SCOTT PERRY UNDER<br>ECURITIES INCENTIVE PLAN           |        |           |           |
|   | APPROVAL FOR THE ISSUE OF TRANCHE 2<br>PLACEMENT SECURITIES TO MR J<br>MATTHEW FIFIELD (EXECUTIVE CHAIR)<br>UNDER LISTING RULE 10.11    |      |                 | OFFER (FY25                 | OF NED SALARY SACRIFICE<br>) TO MR ROSS BHAPPU UNDER<br>ECURITIES INCENTIVE PLAN           |        |           |           |
| ! | 5 APPROVAL FOR THE ISSUE OF TRANCHE 2<br>PLACEMENT SECURITIES TO MR SCOTT<br>PERRY (NON-EXECUTIVE DIRECTOR)<br>UNDER LISTING RULE 10.11 |      |                 | OFFER (FY25                 | OF NED SALARY SACRIFICE<br>) TO MR GARY COMB UNDER<br>ECURITIES INCENTIVE PLAN             |        |           |           |
| ( | 5 APPROVAL FOR THE ISSUE OF TRANCHE 2<br>PLACEMENT SECURITIES TO MR ROSS<br>BHAPPU (NON-EXECUTIVE DIRECTOR)<br>UNDER LISTING RULE 10.11 |      |                 | 13 AMENDMENT                | TO CONSTITUTION  |        |           |           |
|   | 7 APPROVAL FOR THE ISSUE OF<br>PERFORMANCE RIGHTS TO MR J MATTHEW<br>FIFIELD (EXECUTIVE CHAIR) UNDER LISTING<br>RULE 10.14              |      |                 |                             |  |        |           |           |
|   | Please note: If you mark the abstain box for a particul<br>a poll and your votes will not be counted in computin                        |      |                 |                             | not to vote on that Resolution on  | a shov | / of hand | ds or on  |
|   | STEP 3 – Signatures and contact   | dete | ails            |                             |  |        |           |           |
|   | Individual or Securityholder 1  |      | Security        | holder 2                    | Securityholde  | er 3   |           |           |

Director

By providing your email address, you elect to receive all communications despatched by the Company electronically (where legally permissible).

|   | O |
|---|---|
| , | _ |
| è | Σ |
| J | 0 |
|   |   |
|   |   |
|   | D |
| 5 | 4 |
|   |   |

C⊀M

Director / Company Secretary

Date (DD/MM/YY)